

Impact of digital transformation on public governance

New forms of policy-making and the provision of innovative, peoplecentric and inclusive public services

> Author: Millard, J. Editors: Manzoni, M. & Schade, S.

2023



This publication is an External Study report prepared for the Joint Research Centre (JRC), the European Commission's science and knowledge service. It aims to provide evidence-based scientific support to the European policymaking process. The contents of this publication do not necessarily reflect the position or opinion of the European Commission. Neither the European Commission nor any person acting on behalf of the Commission is responsible for the use that might be made of this publication. For information on the methodology and quality underlying the data used in this publication for which the source is neither European to other Commission services, users should contact the referenced source. The designations employed and the presentation of material on the maps do not imply the expression of any opinion whatsoever on the part of the European Union concerning the legal status of any country, territory, city or area or of its authorities, or concerning the delimitation of its frontiers or boundaries.

Contact information

Name: Marina MANZONI Address: Via E. Fermi 2749, 21027 Ispra (VA) - Italy Email: marina.manzoni@ec.europa.eu

EU Science Hub

https://joint-research-centre.ec.europa.eu

JRC133975

Print	ISBN 978-92-68-04552-7	doi:10.2760/355723	KJ-03-23-237-EN-C
PDF	ISBN 978-92-68-04551-0	doi:10.2760/204686	KJ-03-23-237-EN-N

Luxembourg: Publications Office of the European Union, 2023

© European Union 2023



The reuse policy of the European Commission documents is implemented by the Commission Decision 2011/833/EU of 12 December 2011 on the reuse of Commission documents (OJ L 330, 14.12.2011, p. 39). Unless otherwise noted, the reuse of this document is authorised under the Creative Commons Attribution 4.0 International (CC BY 4.0) licence (<u>https://creativecommons.org/licenses/by/4.0</u>). This means that reuse is allowed provided appropriate credit is given and any changes are indicated.

For any use or reproduction of photos or other material that is not owned by the European Union permission must be sought directly from the copyright holders. The European Union does not own the copyright in relation to the following elements:

- page 25, Figure 2, source: Ellen MacArthur Foundation, 2013

- page 38, Figure 3, source: UNESCO, 2016

- page 47, Figure 4, source Diez, et al, 2019

How to cite this report: Millard, J., Impact of digital transformation on public governance - New forms of policy-making and the provision of innovative, people-centric and inclusive public services, Manzoni, M. and Schade, S. (Editors), Publications Office of the European Union, Luxembourg, 2023, doi:10.2760/204686, JRC133975.

Contents

1	Introducti	DN	9
	1.1 Rese	arch context	9
	1.2 Purp	ose and objectives of the study	9
	1.3 Meth	odology	9
2	Definition	s and terminology	
	2.1 Publ	c governance definitions	
	2.2 Publ	c service definitions	
	2.3 Defi	nitions of public values and public value	11
	2.4 Digit	al technology definitions and developments	
3	Existing p	ublic governance paradigms and digital technology impacts	
	3.1 Over	view of existing public governance paradigms	
	3.2 Trac	ng the relationship between digital technology and public services	16
	3.3 Aligr	ing public values and public value with changing public governance paradigms	
4	Strategic	challenges facing public governance from 2020 onward	21
	4.1 Stra	egic challenge 1: Living with turbulence	22
	4.1.1	Open Strategic Autonomy	22
	4.1.2	Security and risk	23
	4.1.3	Societal resilience	24
	4.2 Stra	egic challenge 2: Changing geographies	27
	4.2.1	European and global geo-politics	27
	4.2.2	Re-ordering settlements and communities	
	4.2.3	Turbo-charged localities and communities	
	4.3 Stra	egic challenge 3: Hybrid human-centred relationships	
	4.3.1	Balancing 'episodic' with 'preventative' public services	
	4.3.2	'Community-based solutions'	
	4.3.3	Justice, faimess and inclusion by design	
	4.4 Stra	egic challenge 4: People-planet systems	
	4.4.1	Quintuple helix	
	4.4.2	From sustainability to resilience to regeneration	
	4.4.3	Dynamic systems	41
5	The oppo	tunities and challenges of digital technology for public governance	43
	5.1 A ne	w relationship between people and technology?	43
	5.1.1	Generation 5.0 technologies and services	43
	5.1.2	Industrial revolution 5.0	44
	5.2 The	numan-technology relationship in focus	44
	5.3 The	nisleading notion of 'technological neutrality'	45
	5.4 Tech	nology out of control?	46

6	Implicatior	is for three example sectors	47
	6.1 Food	system governance	47
	6.1.1	Governing the food system prior to 2020	47
	6.1.2	The shock of COVID-19	49
	6.1.3	Food system examples	51
	6.2 Energ	y system governance	
	6.2.1	Governing the energy system prior to 2020	52
	6.2.2	Europe's energy governance shaken by geo-political conflicts	54
	6.2.3	Energy system examples	
	6.3 Work	and employment system governance	
	6.3.1	Governing work and employment prior to 2020	
	6.3.2	The shock of COVID-19	
	6.3.3	Work and employment examples	61
7	Rethinking	European public governance to meet new strategic challenges	63
	7.1 Socie	tal governance of human society enabled by nature's life-support systems	63
	7.2 Re-or	ientating public value and public values from 2020 onward	65
	7.3 A new	/ European societal contract for delivering public value	66
	7.3.1	Good governance public values prior to 2020	66
	7.3.2	Good governance public values from 2020 onward	68
	7.4 Comn	non features and framework conditions	74
	7.4.1	It is the mix that matters	74
	7.4.2	Finding balance	75
	7.4.3	New multi-level governance roles	77
	7.4.4	Breaking down silos	80
8	Conclusion	s and recommendations	80
	8.1 What	are the determinants of changes in public governance	80
	8.2 Sumn	nary of insights from existing public governance paradigms	81
	8.2.1	Digital technology and public governance are intertwined	82
	8.2.2	Public governance evolving features and influencing factors	82
	8.3 Overa	Il conclusions	
	8.4 Recor	nmendations for action and research on present and future European public governan	ce85
	8.5 Recor	mmendations for action and research on the ongoing deployment and development of	[;] digital
	technology	and data	87
Re	ferences		
Lis	st of key abl	previations	
Lis	st of figures		
Lis	st of tables		

Abstract

Public governance concerns how the roles and relations of all actors are organised, structured, managed and administered, including power balance, and capacity and competence, relationships and the levers that each actor has, particularly when using digital technologies. Its overall purpose is to deliver public value benefits that promote the prosperity, wellbeing and flourishing of all people and all parts of society, as enabled by nature's life-support systems. The focus is on the EU but also with a broader international perspective. The setting is (liberal) democratic systems, with their country, regional and local variations and the EU as the overarching level of governance. Since its birth, the EU developed many policies to promote public values of good governance as the means of tackling its social, economic, cultural and environmental challenges and achieve the benefits brought about such values. Yet, at the end of 2022, there is a renewed need for two reappraisals. First, a thorough mapping and understanding of the public governance arrangements, defined as paradigms in this report that have and are being deployed across all the EU's multi-governance levels, the role played by digital technology and the impacts these have had. Second, an analysis of the new strategic challenges the EU faces, especially since early 2020 with the start of the COVID-19 pandemic, the war in Ukraine and the dramatically escalating social, economic and environmental crises, taking account of the role that digital technology and data can play.

Foreword

The digital revolution has transformed our way of working, our working environment, our social interactions, and almost all aspects of our lives on a global scale. Alongside, digital technologies are transforming the way public services are designed and delivered, generating new value ways of interacting between government, its structural organisation, businesses and citizens. These technological 'disruptions' in the role of government and the way to manage and deliver public services are very closely linked with the rise in importance of data as a fundamental basis for decision making, and the reshaping of relations among all actors involved in the creation of public value.

The ongoing digital transition, and related challenges to achieve the targets set for Europe's Digital Decade to 2030, were already well recognised in the European Union policies. Yet, the COVID-19 crisis showed - more than ever - the importance of digital technologies in supporting and enhancing governance processes to ensure a path of transformation towards sustainability and shared prosperity for a better future of our society. As a response, digital transformation became also a central pillar in the Recovery Plan for Europe (NextGenerationEU) while the Recovery and Resilience Facility (RRF) was designed to provide financial aid to Member States to make the European economy more digital and consequently more resistant to future shocks.

The Joint Research Centre (JRC) is the science and knowledge service of the European Commission, with the mission to support EU policies with independent scientific evidence throughout the whole policy cycle, plays a significant role in achieving those priorities. This includes research on monitoring and shape digital transitions¹, and on trustworthy Artificial Intelligence (AI)². The work of the JRC contributes to the scientific evidence base for European policy making and is the foundation of many rich partnerships with policy Directorate-Generals (DGs) of the European Commission, including DIGIT, DG Connect, DG Reform, DG Grow and DG Research and Innovation.

In providing the required scientific advice to EU policy making, we see a need for more research to:

- Investigate the impacts of digital transformation on new forms of governance, including the changing power-relationships between public authorities, the commercial sector, academia and civil society;
- Explore public sector innovation in a digitalised society, including new forms of policy-making;
- Analyse the opportunities of emerging technologies for the co-design, co-development and co-delivery of public services by public and private actors; and
- Outline and assess possible future scenarios for digital governance and framework conditions for the provision of innovative public services (to citizens and businesses).

The aim of this report is to provide scientific support to the JRC to investigate the impact of digital transformation on new forms of public governance, including the changing power-relationships amongst civil society, public authorities, businesses sector and academia. The work presented here follows up on an overview of the prevailing public governance models throughout Europe between approximately 1945 and 2019, and complements another dedicated investigation on emerging public governance models from the perspective of communities. It identifies and analyses the most relevant emerging governance models, including the key features of the evolutionary trends, in selected critical socio-economic sectors (such as trade, consumption, work and employment). In doing so, it helps answering questions, such as (1) How do disruptive technologies foster (positive or negative) change of public governance models? (2) Which innovative governance models increased public value in particular sectors? (3) What are the influencing factors and underlying framework conditions affecting the reconfiguration of public governance processes? (4) Which are the most critical areas for future research to support the digital transformation of multi-level governance processes in the European Union?

Answers all suggest that it is the socio-economic-cultural factors that determine the development of public governance models and their supporting digital technologies. However, they also clearly call for the need of a human-centric based paradigm - as an impelling fundament for a fairer and more sustainable society that is strongly supported by digital technologies, yet solely ruled for and by men, beyond any digital capabilities, and in full respect of human's and nature's rights.

¹ <u>https://joint-research-centre.ec.europa.eu/jrc-science-and-knowledge-activities/digital-transition_en</u>

² https://joint-research-centre.ec.europa.eu/jrc-science-and-knowledge-activities/trustworthy-artificial-intelligence-ai_en

Many more related materials, and latest news about our work are available from the innovative public governance section of the JRC's Science Hub³.

Marina Manzoni

Scientific Project Officer - Digital Governance and Public Service Innovation Directorate for Digital Transformation and Data, Digital Economy Unit European Commission, Joint Research Centre (JRC)

³ <u>https://joint-research-centre.ec.europa.eu/scientific-activities-z/innovations-public-governance_en</u>

Acknowledgements

The author would like to express his gratitude to Dr. Sven Schade and Marina Manzoni of the JRC in Ispra (Italy) for their constant support and trust. To Lucia Errandonea for her generosity in sharing her experience, knowledge, and teamwork, and to Robin Smith for his very useful feedback through the research process.

Also, to the independent experts that participated in the validation workshop for their contributions:

- Maria Alonso Roldan, Agriculture and Digitalization Expert, Universidad de Cordoba, Spain.
- Jaume Martin Bosch, External consultant to the European Commission, JRC.
- Ana Colom, Democracy Expert, Democratic Society, The Open University, England, UK
- Maria del Mar Delgado-Serrano, Agricultural Economics Department, University of Córdoba, Spain
- Noella Edelmann, Department for E-Governance & Administration, Danube University Krems, Austria.
- Silvina Frucella, Air Pollution Expert, AIRENETSpain
- Sabine Hielscher, Institut für ökologische Wirtschaftsforschung (IÖW) GmbH, gemeinnützig, Germany.
- Tina Jukić, Faculty of Public Administration, University of Ljubljana, Slovenia.
- Katerina Makrogamvraki, Civic AI Lab, Universiteit van Amsterdam, the Netherlands.
- Greta Nasi, Department of Social and Political Sciences, Università Bocconi, Italy.
- Morten Meyerhoff Nielsen, eGovernance advisor, United Nations University, Portugal.
- Agamemnon Otero, MBE, Founder & CEO, Energy Garden Cooperative, London, UK.
- Goran Pastrovic, Senior Expert in Quality Management EQUILIBRIUM.INT, and expert for SIGMA OECD, UNDO
- Valeria Righi, Associate Professor. Department for Life Quality Studies, University of Bologna
- Rosalinda Stojova, Head of Sector. State Advisor, Ministry of Public Administration innovation, North Macedonia.
- Louise Thomasen, eGovernance Expert, coThomasen, Denmark.
- Slim Turki, Senior researcher, Open Data & Data Ecosystems expert, Luxembourg Institute of Science and Technology, Luxembourg.
- Zoi Lachana Uaegean, Information and Communication Systems Engineering, University of the Aegean, g Greece.
- Paul Waller, eGovernance Expert, Thorney Island, UK.
- Vishant Weerakkody, Professor of Digital Governance at the School of Management at University of Bradford, UK.

And to those who were not able to participate in the workshop but nevertheless provided feedback:

- Frank Bannister, Associate Professor in information systems in Trinity College. Dublin Ireland.
- James Kahan, Public Policy Expert, (retired from RAND Europe, the Netherlands), USA.
- Daniela Melandri, Founder and Executive Director of UniverCities, Italy and UK
- Jacob Torfing, Professor in Politics and Institutions and research Director of the Roskilde School of Governance, Roskilde University, Denmark.
- Soenke Zehle, media theorist in Experimental Media Lab and the non-profit research company K8, Germany.

Many thanks are due to all the above experts for their contributions, but the author alone takes full responsibility for the final contents of this report.

Author

Jeremy Millard, BSc, MSc

Editors

Marina Manzoni, MSc

Sven Schade, PhD

Executive summary

Policy context

The overall purpose of public governance as addressed in this report is to develop and implement policies that promote the prosperity, wellbeing and flourishing of all people and all parts of society, as enabled by and in full respect of nature's life-support systems. The focus is on the EU but also with a broader international perspective. The setting is (liberal) democratic systems, with their country, regional and local variations and the EU as the overarching level of governance. The EU developed many policies and initiatives that promote its public values and principles of good governance as the means of tackling the challenges it faces to achieve a large range of public value benefits, both with and without the assistance of digital technology. However, at this moment at the end of 2022, in the midst of European and global crises, there is a renewed need for two reappraisals. First, a thorough mapping and understanding of the public governance settings, defined as paradigms in this report that have and are being deployed across all the EU's multi-governance levels, the role played by digital technology and the impacts these have had on public governance. Second, an analysis of unprecedented strategic challenges the EU faces, especially since early 2020, with the start of the COVID-19 pandemic, the war in Ukraine (both revealing fragilities in critical public sectors, such as health, social benefits, employment, immigration, and speculative financial schemes) and the dramatically escalating social, economic and environmental crises, taking account of the role that digital technology and data can play.

Key conclusions

This report shows that successive public governance paradigms appear to align, to a greater or lesser extent, with the major political and societal developments and shocks since 1945, as well as with digital technology developments, especially around a number of pivotal cleavage dates.

The early 1990s: The early beginnings of Generation 1.0, largely one-way, digital technology use in public governance, supporting the traditional Weberian public governance paradigm, boosting New Public Management and then helping to underpin the Neo-Weberian paradigm (emphasizing the need for efficient and timely service to citizens).

About 2000: It began to be accepted that government could use digital technology and data to be the prime mover in proactively delivering public value benefits to wider society and not just to improve the administrative machinery in the back-office. At this time, digital technology first began to change the shape and workings of public governance, as opposed simply to being used by it. It is probably no coincidence that the 2000 cleavage date coincided with the significant jump from Generation 1.0 to Generation 2.0 technology enabling two-way participation and interactions between providers and users. This facilitated both, the networked, and public value governance paradigms as two different political philosophies of society, although arising from the same societal and technological conditions.

From 2008: The financial crisis, itself triggered by the use of Generations 1.0 and 2.0 technology in the banking and finance sectors spurring globalisation, gave simultaneous rise to two quite different public governance paradigms. First, lean and austerity governance and, second, a new plethora of governance paradigms and models characterised by open governance. Both were significantly enabled by new Generation 3.0 semantic-based technology but based on quite different political philosophies of society.

About 2015: Serious moves began to break down governance silos, enabled by the increasing digitisation of both government front- and back-offices. This was in the context of further increasing globalisation with its ensuing economic growth, but scarred by rising inequalities, dissatisfaction and loss of trust in government, coinciding with significant populism and 'post-truth' movements. Both were pushed and pulled by Generation 4.0 distributed and mobile technology. These developments are paired with a significant increase in the availability of data (including personal data) and challenges related to data gathering, capture, access, sharing, and discernment.

From 2020: A new age of more or less continuous crisis, disruption and turbulence seems highly likely. Although the groundwork was arguably laid in the 2008 financial crisis, this was turbo-charged first by the COVID-19 pandemic commencing in 2020, followed by a widely denounced war in Ukraine, as well as the even greater threats posed by the ever increasing environmental, social and economic crises. New forms of public governance are undoubtedly required to meet these existential challenges, so it is also imperative that we design, implement and use digital technologies in a more appropriate and responsible way. In these conditions, whether the putative Generation 5.0 technologies and the Fifth Industrial Revolution, implying the need of dramatically new relationships between humans and machines with people in full control, will suffice remains to be seen.

Main findings

While keeping "public value creation" as its key mandate, it is noteworthy that the types of public govemance paradigms both, change in character, and increase in variety over time. Chronologically, they change from a small number of basic models mainly concerned with internal process, administration and organisation, drawing their justification from an increasing array of 'good governance' public values, towards a much larger number of models bearing a clear public value focus that is directly conceived to address important and concrete societal challenges. This switch taking place around the year 2000 coincides with the big uptake of digital technologies by the public sector and its administrations, starting with Generation 2.0 interactive technology, acting as a highly significant enabler for users, as well as to changing politics and policies. From 2020, a step change now seems to be taking place, mainly influenced by a series of economic, political and environmental turbulences that challenge previously established stakeholders' relations, and have major impacts on the geopolitical landscape. These changes seem to be due to: (i) the huge variety of ways different countries and authorities react both politically and organisationally to their churning socio-economic, environmental, cultural and historical situations; (ii) the influence of external factors and crises; and (iii) the increasingly diverse possibilities enabled by innovative digital technology.

The main findings up to 2019 regarding digital technology and public governance include:

- Everything is inevitably influenced by digital technology design, development and use, and it is better to
 explicitly account for that, than ignoring digital potentialities.
- Yet, digital technology and digital data are best seen as necessary but never as sufficient tools for public governance. Their deployment is always mediated by organisational, institutional, legal, ethical and social conditions, as well as challenges such as digital exclusion and increasing digital gap.
- Considering the many and diverse interrelationships between all actors involved in public governance, digital technologies repurposing and a responsible data harvesting, aggregation, management and use, might be considered as the only possible way to reach fairness, transparency, accountability and inclusiveness.
- The complex roles of digital technology and human-technology relations need to take account of how peoples' quality of life, values and ethics are impacted by increasingly omnipresent digital machines, escalating surveillance by both, public, and private sectors, as well as by the 'post-truth' society manipulative communication strategies and tools, and the 'black-box', impenetrable and inevitably biased AI-based solutions.
- The mix of public governance paradigms that are simultaneously present in any one place and time, including across multi-governance levels, is an important contextual response to prevailing political, socio-economic and cultural differences in given times. However, these responses and mixes may or may not be successful, so governing this mix becomes a crucial issue where the deployment of digital technology is necessary.
- To cope with the challenges of our times, we thus need public services that are, first of all, relevant to the different contexts, digital-ready and interoperable by design – across borders, across sectors, and across different levels of public administrations.
- We are still in an era of understanding the best (context dependent) ways of using digital data and technologies for the public good, and a rich set of diverse experiments still yet to be completed or be entirely conducted to fulfil an entire wave of digital transitions of public governance.

In terms of existing public governance models, the main findings up to 2019 include:

- Political, policy, socio-economic, environmental, historical and cultural factors, especially societal-wide and
 often international shocks and crises, are the strongest influences on public governance developments.
- The public governance paradigms accumulate layer upon layer resulting in a form of sedimentation over time. Each of the nine paradigms identified between 1945 and 2019 (each with one or more models) continue to have relevance today, although the more recent novel and emerging paradigms tend now to be more dominant but perhaps only because collectively they are more numerous.
- These public governance paradigms do not form a linear progression where one replaces the other but represent a process of co-evolution. Differences can also be seen between central and local governments in the same country, underlining the importance of re-visiting multi-level governance, and to elaborate on

possible evolutions of the original approach – especially as applied within the EU (within and across countries).

- It is the actual 'mix that matters'. The mix of paradigms and models at a specific place and time depends on the context of global, national and local politics, as well as history, culture, socio-economics, environmental factors and the political choices made. This mix may or may not be a successful arrangement, so governing this mix is a crucial issue where the deployment of digital technology is necessary.
- Co-creation, and thus generative and emergent governance paradigms, seem imminent. Both tacit and codified knowledge are a powerful combination for learning, identifying good practices of successful transformations, and thereby for widespread co-creation, replication, scaling and knowledge sharing. At least up to 2020 only promising examples are seen without a truly systemic approach at EU or national levels. It is not yet clear what a balanced approach, which applies representative methodologies to some societal challenges while investing in more participatory approaches in others, might look like. These deliberations on co-creation might be further divided into co-initiation, co-design, co-implementation and co-evaluation, as a cyclic approach that departs from the traditional ladder analogy for citizen participation.
- The most desirable re-balancing of power and responsibilities for future-proof public governance still needs innovation and experimentation so is yet to be found. This will also require a cultural change within public institutions that tend to be risk averse. These risks are often seen in the short term (for example, but not only, connected to election cycles) without considering the longer-term risks of not rebalancing power relations, and without experimenting with new approaches to public governance.

Related and suggested future work

The recent crises (from 2020 onward) have already set back decades of public governance progress that had earlier, and despite much unevenness, led overall to greater prosperity, wellbeing and flourishing of most people in most parts of society, and that had begun to make some progress in addressing the vulnerability of nature's life-support systems. At the same time, however, the recent crises also provide opportunities to rethink how public governance and the role of digital technology can assist Europe in getting back on track. This research has identified four strategic challenges that now require urgent public governance action and research, although this is not to imply that there are no other challenges likely to arise:

Living with turbulence: Public governance needs to accommodate shocks and crises by moving to more resilient paradigms that move away from traditionally-determined efficiency criteria by also prioritising diversity and interconnectivity. The goal is to be equipped to react better, more nimbly and more timely to unexpected events as well as longer term risks, including environmental breakdown, whilst still providing as much continuity, stability and predictably to society as possible. This should be the case even when it decreases some short-term efficiencies given the clear evidence of the societal damage prevented, including the improved long-term efficiencies that can instead be realised. There should be increased focus on the EU's open strategic autonomy in essential goods and services, also through multi-governance levels where beneficial. Cyber threats and other known risks also need continuing focus, as does seeing resilience in social and economic as well as environmental terms given these are intimately interrelated.

Changing geographies: Changes to Europe's geo-political position have been turbo-charged by the recent turbulence creating new global realignments that are reshaping economic and security arrangements and their political manifestations. New EU public governance at all levels needs to draw upon and strengthen the EU's values and principles if these changes are to be successfully addressed. Sub-national level trends include the growing importance of small cities and larger towns within the hinterlands of very large cities which, often excepting capital cities, are themselves decreasing in importance driven by digital work, learning, shopping, etc, There is also a very strong resurgence of local economies and identities linked to efforts to retain as much as possible of locally-created value within the locality. This is seen, for example, by residents becoming more aware of their locality and community as well as the economic benefits of local production and consumption, and local governments focusing much more on local democracy, partnerships and procurement. Urban, rural and community-level geographies are also changing in directions that both accelerate existing trends and carve out new pathways, especially ones that prioritise decentralisation within strong national and EU structures in order to ensure just transitions for all wherever they live.

Hybrid human-centred relationships: Traditional welfare state public services, first designed in the early post-1945 period, are no longer fit for purpose. The new age of turbulence, coupled with rapidly changing demographics towards greater cultural and ethnic diversity and an increasingly aged population, require public

services that prioritise the prevention or slowing of social and economic misfortunes affecting individuals and communities, especially manifest at local level and through community-based solutions. Such a shift can be both economically efficient as well as socially beneficial and, especially when meeting the needs of the most vulnerable, requires strong human relationships alongside hybrid service delivery mechanisms to be successful Resilient structures across all multi-governance levels are necessary to delivery such solutions with a focus on justice, fairness and inclusion by design.

People-planet systems: Arguably the greatest challenge that public governance has arises from the scientific facts that climate change, biodiversity loss and other stresses on nature are having profound deleterious impacts on the functions of societies and on the lives of their inhabitants. The needs of a flourishing society are bound together with a flourishing nature, both organic and inorganic, so nature must be centrally incorporated into all public governance paradigms. This should involve moving on from a concept of sustainability that aims only to keep what we currently have through conservation and preservation, and instead aims to regenerate what we really need. This will also involve public governance that better understands the complex dynamic systems that intricately link human societies with natural systems.

The above strategic challenges, if successfully addressed by public governance, will provide highly significant public value benefits across all European societies. This also requires further action and research into the European public values and principles as the means necessary to deliver these public value benefits. It is proposed these include: democracy and power relations; trust and trustworthiness; risk governance and crisis management; regulation; public sector innovation, experimentation and learning; behaviour, values, principles and culture; new innovation frameworks and mission strategies; public governance management and integrity; future proofing; and policy coherence, coordination, alignment and evaluation.

Further research and action into the best roles that digital technology can play in support of the above is needed, especially in relation to the human-centred approaches to digital technology that the putative government Generation 5.0 technologies can play. This aims to focus on real people and real peoples' lives where the human, human-touch and human control are more important than the digital, but where public value can be enhanced by the digital. Digital 'sanitisation', in the sense of audits that reduce or remove the digital where this does not add value, but instead increases digital divides, should take place. In parallel with this there is a need for research that the anticipated Industrial Revolution 5.0 can play in envisaging humans and machines performing work hand-in-hand. Combining the unique cognitive abilities of workers and the accurate technical expertise of, for example, robots can better promote an innovative and productive culture in the workforce.

Quick guide

This report is organised into eight sections, as follows:

- 1. An introduction summarising the research context, purpose, objectives and methodology of the study.
- 2. An overview of definitions and terminology used.
- 3. A summary of nine existing public governance paradigms and digital technology impacts from 1945 to 2019, which also refers to an accompanying report that examines this state-of-art in detail.
- 4. An evidence-based description of the four proposed strategic challenges facing public governance in Europe, from 2020 onward.
- 5. An analysis of the opportunities and challenges of digital technology for public governance going forward.
- 6. The implications of these strategic challenges and the role of digital technology in three example sectors: food, energy and work and employment.
- 7. Proposals for rethinking public governance to meet the new strategic challenges.
- 8. Conclusions that summarise the determinants of changes in public governance, the insights from experience to date, and recommendations for action and research in relation to both public governance and the role digital technology can play going forward.

1 Introduction

1.1 Research context

The digital revolution has transformed our working environment, our social interactions, and almost all aspects of our lives from the local to the global scale. In parallel, digital technologies are transforming the way public services are delivered and generate new ways of interacting between governments, businesses and citizens. These technological 'disruptions' in the role of government and the way to manage and deliver public services are very closely linked with the rise in importance of data as a fundamental basis for decision making, and the reshaping of relationships among all actors involved throughout the public sector value chain.

The outbreak caused by the COVID-19 crisis showed - more than ever - the importance of digital technologies to enhance governance processes to ensure a path of transformation towards sustainability and shared prosperity for the future of our society - in line with new Digital Strategy for Europe (European Commission, 2021k). In this context, the role of the public sector both, as enabler and regulator is crucial. This calls for investigating deeply the emerging paradigms of governance "with and of" the digital technologies, which are transforming the way governments operate. It can enhance how public services are designed and delivered, and how policies are shaped, implemented and evaluated. However, more research is needed to:

- Investigate the impacts of digital transformation on new forms of governance, including the changing
 power-relationships between public authorities, the commercial sector, academia and civil society.
- Explore public sector innovation in a digital society including new forms of policy-making.
- Analyse the opportunities of emerging technologies for the co-design, co-development and co-delivery of public services by public and private actors
- Outline and assess possible future strategic challenges for digital governance and framework conditions for the provision of innovative public services (to citizens and businesses).

1.2 Purpose and objectives of the study

The aim of this study is to investigate the impact of digital transformation on new forms of public governance, including the changing power relationships amongst public authorities the business sector, academia and civil society. The **two specific objectives of the study** are the following:

- Provide a thorough mapping and understanding of the public governance arrangements, defined as
 paradigms in this report, that have and are being deployed across all the EU's multi-governance levels
 and the role played by digital technology. An accompanying report examines this state-of-art in detail.
- Undertake an analysis of the new strategic challenges the EU faces, especially since early 2020 with the start of the COVID-19 pandemic, the war in Ukraine and the dramatically escalating social, economic and environmental crises, taking account of the role that digital technology and data can play.

This report provides the findings from the work, including a summary of the public governance models that prevailed in democratic systems (especially in the EU) from approximately 1945 to 2019. A more detailed elaboration on these models has been published separately (Millard, 2023).

1.3 Methodology

The methodological approach adopted in this study consists mainly of a desk research survey of the main public governance activities and rationales adopted by public government and other relevant actors since 1945. This involves unpicking the reported purposes, structures, actor roles and relationships, digital technologies used, strengths, weaknesses, framework conditions and influencing factors that these changes have involved. Thus, the approach is largely empirical in the sense that it draws directly on what the involved actors, including academics and other commentators, report during different periods since 1945. It then attempts to analyse the impacts this has had on public services, public values and public value, illustrated by numerous examples. There is no attempt to impose a top-down theoretical structure, however useful this might be in another context, but rather to trace the explanations and justifications relevant actors give for the different types of public governance adopted. In addition to this desk study based upon a wide-ranging literature review of many actual examples and their impacts using a chronological approach, i.e. how both ideas and practice change over time, an online workshop with practitioners and experts was held on 7 October 2022 as well as widespread consultations with other experts. This provided both a validation of work done up to this point as well as useful critiques and insights, most of which informed the final stages of the study.

2 Definitions and terminology

2.1 Public governance definitions

For the purposes of this report, public governance concerns how the **roles and relations of all actors are** organised, structured, managed and administered, including power and competence relationships and the levers (i.e. instruments for getting things done) that each actor has, for the overall purpose of delivering public services and public value. A focus of the report is on the role of digital technology⁴ in the evidence gathered, although examples not currently using digital technology to a significant extent are also included given that digital technology could or is likely to have an impact in the future.

Five actor types are in focus: public institutions including their direct and indirect representatives; civil society and citizens; the private sector; educational and research institutions; and nature, the needs of which need to be considered given that meeting these underpins the viability of the other four actors to function. Different public governance purposes are served by appropriate roles, arrangements and relationships between all five actors expressed through different **public governance paradigms**, defined by Torfing et al (2020) as a relatively coherent and comprehensive sets of norms and ideas about how to govern, organise and lead the public sector. Each paradigm is composed of one or more specific **public governance models** that demonstrate some of the main characteristics of the paradigm in question.

As this report focusses on paradigms and models of public governance, the public sector is the pivotal actor. Yet, this does not always mean that it will take the leading or most important role in terms of resources invested or decision-making powers, It does mean its role is essential by leading or contributing to the provision of public services and public value. The focus is on **democratic public governance systems**, especially the different forms found in Europe where democracy can be thought of as the 'power of the people' as a way of governing which depends on the will of the people and a strong grounding in human rights (Council of Europe, 2022).

All paradigms are normally also relevant across **multi-governance levels** which, in Europe, typically operate at three country-specific levels, i.e. national, regional and local, although other and/or additional or fewer levels may be appropriate in different administrative regimes. On top of these three levels is the overarching public governance level of the European Union. Example public governance models are often taken from one or more specific governance levels.

Section 3 of this report summarises existing public governance paradigms and models in Europe from 1945 to 2019, whilst an accompanying report on the state-of-art examines these in detail and provides over 50 short case examples to illustrate its findings. Section 4 describes the four proposed strategic challenges facing public governance in Europe, from 2020 onward, Section 5 analyses the opportunities and challenges of digital technology for public governance going forward, and Section 6 looks at the implications for three example sectors: food, energy and work and employment. Section 7 proposes how to rethink public governance to meet the new strategic challenges, whilst Section 8 concludes with a summary of the determinants of changes in public governance, the insights from experience to date, and recommendations for action and research in relation to both public governance and the role digital technology can play going forward.

2.2 Public service definitions

A working definition of public services is **services that are designed and delivered for the sole purpose of creating various types of public value**, as defined in Section 2.3 below⁵. Public services can be designed and delivered by the public sector on its own, with or via the other actors mentioned above operating within different government paradigms and models, particularly but not only in this report when using a variety of digital technology. This report proposes examining public services as either or both '**individual' and 'collective' public services**, for example:

⁴ The term 'digital technology' is used in this report as a collective noun encompassing all possible 'digital technologies' without implying any specific type. The term 'digital technologies' is used when the focus is on more than one specific type which are normally further specified. In addition, 'digital technology' is not to be confused with 'digital transformation' where the focus is on the changes in wider society or in nature that digital technology causes, enables or assists.

⁵ This is similar in intent to earlier definitions of public services as services of 'general public interest', i.e. as services of interest/relevance to everyone in society, where 'everyone' refers to both the individual and the collective levels and is denoted by the word 'general'.

- 'Individual'/'direct' public services: include traditional public services aimed primarily at individual citizens in areas like education and training, health, care, social services, libraries, employment, tax, etc. These are individual services as they normally operate at the individual level by establishing a direct interaction between individual service providers and individual service users, thereby providing individual public value as only the individual gains direct benefit.
- "Collective'/'indirect' public services: created particularly by the collective impact of laws, regulations and the overall actions of the public sector in areas like planning, policy- and decision-making, institutional frameworks, legal and regulatory frameworks, administration, standard-setting and monitoring, participation and democracy, public procurement, services benefiting the environment, trade, defence, security (both cyber and non-cyber), investments in infrastructures, R&D, etc. In addition, public utilities like energy and water, police and prisons, although these also clearly have impact at the individual/direct level. These are all examples of collective services as their benefits are used or enjoyed by everybody at the same time, mainly continuously and often 'out-of-sight'. They normally operate at societal level as part of a public value framework/structure within which all actors operate, thereby providing collective public value, as everybody gains benefit⁶.

2.3 Definitions of public values and public value

European public values are ultimately derived from the European Union's overarching values as defined in Article 2 of the EU Treaty: "The Union is founded on the values of respect for human dignity, freedom, democracy, equality, the rule of law and respect for human rights, including the rights of persons belonging to minorities."⁷ The overall basis of the EU and its public values is its governance system which designates the body of rules, procedures and practices that relate to the way powers are exercised in the EU. The objective is to strengthen democracy at the EU level and to bring citizens closer to the EU institutions. In concrete terms, the EU's approach to its public values used in this report is reflected in its good governance framework based on an amalgam of both principles and values defined as (European Commission, 2017):

- **"Good governance** is an agreed set of principles and values widely shared. There is no 'right' or 'wrong' formulation: each administration has its own typology and terminology, but there are recurring themes."
- "Principles should be fundamental and enduring. An example is honesty, which should apply to all public officials, irrespective of time or place. In some cases, principles are adopted in laws or regulations, as rights or obligations on the administration, including in the form of civil service acts."
- "Values may also be constant, but equally can emerge and evolve over time as conditions change. They might appear to be timeless, but can arise as a product of circumstance, such as transparency which is a relatively recently phenomenon."

A consensus view of current European good governance can be summarised in the following nine public values of good governance (European Commission, 2017), which also directly align with the UN approach to good governance (United Nations, 2009) underlying the Sustainable Development Goals (United Nations, 2015b).

- 1. Accountable
- 2. Transparent
- 3. Responsive
- 4. Equitable and inclusive
- 5. Efficient
- 6. Effective
- 7. Follow the rule of law
- 8. Participatory
- 9. Consensus oriented

⁶ Collective public value can also be defined as 'societal value'.

⁷ <u>https://eur-lex.europa.eu/eli/treaty/teu_2012/art_2/oj</u>

In distinction to public values, seen as the means of good governance, public value is seen as the actual benefits to society created. The working definition used in this report distinguishes **three levels of public value** (Kelly et al, 2002):

- 1. "Services that provide the vehicle for delivering public value through actual service encounters for users or clients and the distribution of fairness, equity and associated values for citizens."
- "Outcomes that commonly overlap with services but should be considered separately as they
 encompass much higher order aspirations such as national security, poverty reduction, public health,
 etc." to which different combinations of services collectively contribute depending on the political and
 policy context.
- 3. "Trust, legitimacy and confidence in government at an even higher level which are critical to public value creation: even if formal service and outcome targets are met, a failure of trust will effectively destroy public value". As trust is today also seen as one of the essential values of good governance, it thereby provides an overlap with the European good governance public values listed above.

It can be seen from these three levels that the first approximates to the benefits created for individuals by individual/direct services, and the second approximates to the benefits created by collective/indirect services, as defined in Section 2.2 above. Not necessarily created by public services, however, is Kelly et al (2002)'s third level, although the first and second levels together can, of course, help create or destroy trust, legitimacy and confidence in government.

Leading on from the above, it is useful to clarify the following distinction used in this report between the public values of good governance and public value benefits:

- **The public values of good governance** (based on the European Commission, 2017) refer to how public value is created thus public values can be seen as the 'means' of creating the 'ends' of public value
- Public value (based on Kelly et al, 2002) refers to the actual benefits derived from public governance that
 accrue to all actors thus public value can be seen as the 'ends' delivered by the 'means' of public values.

It is important to note that this report, its analysis and conclusions, is prepared from the perspective of European public values of good governance and an understanding of European public value derived from democratic public governance systems, especially the different forms found in Europe.

2.4 Digital technology definitions and developments

In the context of this report, examining the evolution and future development of public governance paradigms and models, the impacts of both 'specific' and 'general' digital technology lenses are examined. The **'specific digital technology'** lens looks at individual technologies or groups of technologies and how they evolve over time. This is outlined in Table 1, and briefly summarised at the bottom of Figure 1, showing how each of the five generations of both government and of the web found in the literature tend to be aligned, despite some contrasting definitions and timelines depending on the particular purpose and perspective of each source. Specific generations of government and the web tend to develop together, although web advances often precede government advances by several years as the technology is typically a facilitating factor. However, it is also clear that the government-digital technology relationship is not uni-directional from the latter to the former, given that governance independently also shapes digital technology through economic and technology policy, regulation, investment, procurement, etc.

The '**general digital technology**' lens sees today's digital technology as the latest stage in the evolution of 'general purpose' technologies through successive industrial revolutions each of which builds on previous stages like steam and electricity that were already in place prior to 1945. Thus, digital technology underlies and supports virtually all other technologies and everything in a modern society. For example, today digital technology enables robotics, additive manufacturing and 3D printing, data processing, social media, environmentally-sustainable technologies, food production, health care, etc. According to the World Economic Forum (WEF, 2016): "The First Industrial Revolution used water and steam power to mechanize production. The Second used electric power to create mass production. The Third used electronics and information technology to automate production. Now a Fourth Industrial Revolution is building on the Third, the digital revolution that has been occurring since the middle of the last century. It is characterized by a fusion of technologies that is blurring the lines between the physical, digital, and biological spheres".

Generation	Government generation ⁸	Web generation ⁹	Example digital technology ¹⁰
1.0 c. 1993→	Online access to service information; standalone services and automation of some traditional back-office procedures but without re-engineering.	'Read-only' using html pages to display information and deliver content.	Static web-pages and web- directories, search engines (e.g. Netscape, early Google)
2.0 c. 2000 - ≯	Interactive 'socialised' user-centred services; user generated content; collaborative tools for more open, accountable, responsive and efficient government; recognition of digital divide: start of re-engineering both front-office services and back-office procedures to better fit societal demands.	'Read-write' functionality; social and interactive media (e.g. fora); early machine- machine interaction & networking.	Early APIs; web apps like podcasts, blogs, images, videos; social networks.
3.0 c. 2008→	Semantic web-based government that can start to personalise & proactively deliver services; balancing efficiency and effectiveness; open & platform government, open government data; service co-creation; inclusion and participation; CAPs (community- awareness-platforms); crowd-sourcing; e- Signature and e-Identity; gamification; once-only; digital-by-default; policy modelling; smart city.	Semantic 'intelligent' web; interoperability; open-source; open data; intelligent machine- machine interaction; early data decentralisation sharing by users; smart mobile services; traceable & transparent interactions.	Data integration, automation and discovery; natural language processing; web ontology language (OWL); 3D graphics and spatial 3D; connectively and ubiquity; gamification; cloud computing; IoT; open data; early AR, VR; robotics, AI & blockchain.
4.0 c. 2015→	From e-gov to digital gov; optimising work processes and systems; operational efficiencies and increased productivity through agile approaches using big data; life-events triggering advanced personal- ised & proactive government; breaking down silos, joined-up and whole-of- government; one-stop-shop; learning, 'cognitive' & predictive government, e.g. no-stop-shop (through AI and machine learning).	Distributed & mobile data regardless of location; self-learning systems; full user- control;	Voice and symbiotic interaction between physical and virtual worlds; big, linked & interactive data; more advanced AI, blockchain, robotics, AR & VR; facial recognition; crypto currencies; decentralised apps; RFID tags; digital twinning (replicas).
5.0 c. 2020 - ≯	From user- to human-centric; focus on real people & real peoples' lives; services fitting peoples' lives, not the reverse, using a 'whole-of-life' mindset; the human is as important as the digital; governing not for but with people; people expect more 'human touch' through digital sanitisation; balancing efficiency with both effectiveness & 'affectiveness' (peoples' emotions); full virtual assistants.	Decentralised web apps platform & APIs to empower people to regain and retain control of own data and identity; but opposite dangers of public and/or private surveillance.	Emotional intelligence; full digital wallets; full social personas; full digital twinning; full VR (metaverse?), decentralised web nodes; decentralised identifiers; digital sovereignty; cryptographic security.

Table 1. Indicative cumulative developments in government and web generations

Source: The author (Millard, 2023).

3 Existing public governance paradigms and digital technology impacts

3.1 Overview of existing public governance paradigms

Based on the literature survey and evidence undertaken for this report (see Millard (2023) for more details), Figure 1 provides a synopsis of the main developments since 1945 of European public governance paradigms,

⁸ Governments 1.0, 2.0 and 3.0 are largely derived from GOV 3.0 (2021); Government 4.0 is largely derived from Telecom Review (2022), Long et al (2021) and European Commission (2021b); Government 5.0 is largely derived from PwC Consulting (2019) and European Commission (2021b).

⁹ Web 1.0, 2.0 and 3.0 are largely derived from Mosinegutu (2022) and Hackeroon (2001); Web 4.0 is largely derived from Hackeroon (2021); Web 5.0 is largely derived from Weston (2022) and Uptech Solution (2022).

¹⁰ Sources include those shown for government and web generations.

each of which gives rise to one or more public governance models in three main groups: A) prevailing, B) novel and experimental, and C) new and emerging. Only A) and B) are summarised in this Section 3 of the report with a summary of their characteristics provided in Table 3, while a detailed consideration of C) is presented in Section 4. The intention is not to provide exhaustive portraits of each group, but to attempt to highlight archetypes of representative public governance paradigms and their constituent models rather than attempting to cover everything.

A. Prevailing: illustrated by four representative governance paradigms (numbered 1), 2), 3) and 6) in Figure 1. To a large extent, these still provide the basic foundations of public governance today in all European countries, depending on their different historical, political and cultural conditions. The heyday of 1) and 2) long predates any significant relevance of digital technology for public governance, which first started to have an impact in the early-1990s. On the other hand, 3) coincides with the early impact of digital technology and 6) arose as a heightened version of 2) in the wake of the 2008 financial crisis. All four are, however, less prominent today, if only because there is now a large number of more recent governance paradigms that crowd the field (as covered in B below):

1) Traditional Pubic (Weberian) Administration: from about 1945 with digital technology only relevant from the early-1990s.

2) New Public Management (NPM); market-based: from about 1980 with digital technology only relevant from the early-1990s.

3) The Neo-Weberian State: from the late 1990s: a reaction against NPM and some reversion to Weberian Administration but with a more external orientation.

6) Lean and austerity: a particular version of NPM in response to the financial crisis commencing in 2007-8.

These four paradigms arise from a long-established combination of mainly top-down and market-driven policies, and their influence is still prevailing today in the form of underlying structures and principles. However, their dominance is now weakening as more recent governance paradigms, often enabled by digital technology, are taken up.

B. **Novel and experimental**: illustrated by at least five governance paradigms, forming two chronologically defined clusters in Figure 1:

4) Networked and 5) Public value management: these generally coincide with the significant impact of Generation 2.0 digital technology on public governance from about 2000.

7) Open: begins to have a significant impact on public governance in the wake of the 2007-8 financial crisis coinciding with Generation 3.0 digital technology

8) Sustainability and 9) Locality and community: these begin to have a significant impact on public governance from about 2015 coinciding with Generation 4.0 digital technology.

These group B) five novel and experimental governance paradigms give rise to a large number of distinct models that are today more important collectively in most countries than the prevailing models in group A). However, their variety and relatively recent provenance mean their individual adoption is, to date, not as ubiquitously widespread as Group A). Their developmental speed and sophistication is necessarily based on digital technology to varying degrees, although the technology is never itself a sufficient condition. Indeed, all of the paradigms illustrated in Figure 1 are primarily driven by contemporaneous influencing factors and framework conditions in different European countries, as outlined below in Section 3.3.

C. New and emerging strategic challenges: the number and type of governance paradigms from about 2020 is as yet unclear as they are largely in the future and thus highly speculative. Thus, at this stage, four strategic challenges are identified which new public governance paradigms will need to address through a rich set of diverse experiments. These strategic challenges are proposed by tracing very recent developments and commentary and are strongly interrelated as they arise from, or are at least significantly accelerated by, turbulence and uncertainty. Although these challenges initially kicked off in the aftermath of the 2007-8 economic and financial crisis, they become acute and much more visible in early 2020 with the onset of the COVID-19 pandemic, now reinforced by the invasion of Ukraine as well as the dramatically worsening climate crisis and other ongoing shocks (for an overview of shocks, see WEF, 2022a):

10) Living with turbulence; 11) Changing geographies; 12) Hybrid human-centred relationships; and 13) People-planet systems.



Figure 1. Evolution of European public governance paradigms¹¹

Source: The author (Millard, 2023).

During the whole time span examined from 1945 to the present, each public governance paradigm, once established, retains real influence to the present day, although this often declines in relative terms as new paradigms are created. Thus, commencing in about 2000, new paradigms will also run in parallel, building upon each other in distinctive ways related to each countries', as well as to Europe's, key influencing factors and framework conditions. In these different contexts, different paradigms are likely to both complement as well as contradict each other. Full details of the existing public governance paradigms and many of their constituent models are available in an accompanying report that examines their state-of-art from 1945 to 2019.

¹¹ Note that the thirteen paradigms are numbered chronologically, but positioned in Figure 1 according to which of the three groups (A, B or C) the paradigm is allocated.

3.2 Tracing the relationship between digital technology and public services

Section 2.4 described the specific roles of digital technology for designing, developing and especially delivering specific types of public services, thereby facilitating specific types of public governance. How this has developed since the early 1990s is summarised in Table 2. The use of data and information is crucial for digital technology, especially in terms of how they are acquired, stored, re-accessed and analysed to create an 'information regime' for the rational or efficient conduct of governance as well as of business. In all pre-digital public governance models, data compression had been used to reduce complex realities to data and information in formats and quantities that could be easily classified, indexed, filed and re-found when needed, mainly by physical means. With the advent of New Public Management from about 1980 and then the Gov 1.0 and Web 1.0 technologies from about 1993 in Table 2, hybrid forms of machine/professional bureaucracy developed, focusing on metrics-based compression using pre-fixed statistics, key performance indicators and similar data in a central governance role.

From about 2008 onwards with Gov 3.0 and Web 3.0, the importance of data and data governance become increasingly important as one of the prime assets of the public sector. Technologies facilitating big data, artificial intelligence and data science approaches have made feasible a new information regime of 'lossless' uncompressed data and expanded data science, opening a potential for bureaucratic operations to alter in fundamental ways fostering new forms of post hoc knowledge development, e.g., via machine learning and algorithmic governance (Annoni et al, 2018; Dunleavy 2022).

In this context, the 2022 European Data Governance Act, fully in line with EU values and principles, is designed to bring significant benefits to EU citizens and companies by increasing trust in data sharing, strengthening mechanisms to increase data availability and overcoming technical obstacles to the reuse of data. Common European data spaces will be established in strategic domains, involving both private and public players, in sectors like health, environment, energy, agriculture, mobility, finance, manufacturing, public administration and skills. This will be a powerful engine for innovation and new jobs allowing the EU to ensure that it is at the forefront of the 'second wave of innovation based on data'. Society as a whole will benefit from more evidence-based policies and better solutions to societal challenges, such as climate change and the COVID-19 pandemic (European Commission, 2022a; De Nigris et al, 2020). Understanding the influence of online technologies on political behaviour and decision-making is also important, for example in terms of behavioural psychology where it is clear that "social media changes people's political behaviour", in particular through the attention economy, choice architectures, algorithmic content curation; and mis/disinformation (Lewandowsky et al, 2020).

Drawing on Table 1, and examining the evidence presented in this report and especially the accompanying state-of-art report, Table 2 provides a summary overview of how the specific types of digital technology aligned with each public governance paradigm have impacted public services since the early 1990s. Table 2 has been adapted, clarified and updated from the European Digital Forum (2015) and aligns surprisingly well with the four government and web generations used in Table 1, themselves derived directly from the literature. The fifth generation has been added in this report to bring the schema fully up-to-date.

Table 2 thus shows how each of the digital technology generations impacts public services, changing cumulatively from providing support in Generation 1.0, through enabling, then driving followed by digital-only services in Generation 4.0. Each generation reveals a specific relationship between humans and digital technology in how the services are delivered and operate and, in turn, each generation offers its own specific array of public services linked to the cumulative development of public governance paradigms. It must be stressed that, like Table 1, and the public governance paradigms of Figure 1, Table 2 is cumulative. Thus, it is not a question of one generation of human-technology relationships and types of service replacing those of a previous generation, but rather complementing and building on what already exists.

The human-digital technology relationship in Table 2 is of particular interest as through Generations 1.0 to 40 it changes sequentially from technology being the main supporter of the public services provided to doing most or all of the work to the extent that it sometimes becomes invisible. The next step, to the future Generation 50, is crucial as this involves profound human, societal and philosophical issues about what we want technology to do. This is examined in more detail in Section 5.1.

	Digitally- supported public services (from early 1990s)	Digitally- enabled public services (from 2000)	Digitally-driven public services (from 2008)	Digital-only public services (from 2015)	New hybrid public services (from 2020, but mainly in future)
Description	Humans provide service but back- office processes are digitised. Typically, these services rely on physical work and/or tangible assets	The potential for digitising public services is huge, but these services are only partly digitised, so digital technology but can only play an enabling role	Public services can be largely digitised, hidden & operate semantically & intelligently in the background, providing personalised & proactive services. They do not normally produce physical deliverables	Artificial intelligence (AI) will analyse available relevant data to deliver insights & automatically intervene to achieve best outcomes via fully joined-up & predictive services	Decentralised web app platforms & APIs, advanced AI & blockchain, etc., to empower people to regain and retain control of digital life & how it mixes with physical life
Human role	Provide the service, supported by the digital technology	Provide all evidence-, value- and judgement- based decisions	Only intervene where necessary	Define & design framework conditions and determine what are 'best' and/or acceptable outcomes	'Whole-of-life' physical & digital control; hybrid of emotional, human & digital interactions.
Digital technology role & generation ¹² (cumulative)	Supports humans in providing the service. Gov 1.0 & web 1.0	Enables the service. Gov 2.0 & web 2.0	Both generates & delivers the service. Gov 3.0 & web 3.0	Find & analyse data, make & implement decisions within framework. Gov 4.0 & web 4.0	All tasks, but under full conscious human control and/or predictive & benign intervention. Gov 5.0 & web 5.0
Public service examples (cumulative)	Law making & enforcement, defence, utilities, emergency services, transport, social housing, healthcare, social care, education, etc.	e-Healthcare, e-Medicine, e-Learning, e-Town planning, e-Mapping, e-Property data etc.	Payment of social security, pensions & other benefits & taxes; purchase of services; administrative services; obtaining licenses, permits, permissions, etc.	Full life / busines events / other event services (e.g. for education, health prevention, running a company); early warning systems to identify, diagnose & prevent known & unknown risks, etc.	Fully hybrid services of all kinds; optimise interaction between physical & virtual life with human in full control also of own data & identity; full virtual assistance, digital twinning, personas, etc.
Public1) TraditionalgovernanceWeberianparadigms2) New Public(cumulative)3) Neo-Weberian		4) Networked 5) Public value	 6) Lean & austerity 7) Open governance: open, platform, participatory, inclusive, co-creative, behavioural & cultural 8) Sustainability governance: sustainability, circularity, nature-based 9) Local & community governance: locality, civil cociety involvement, community, 		10) Turbulence 11) Changing geographies 12) Hybrid human- centred systems 13) People-planet systems

 Table 2: Tracing the cumulative relationship between digital technology and public services since the early 1990s

Source: Adapted and updated from European Digital Forum (2015), see also Millard (2023).

 $^{^{\}rm 12}$ For definitions of government and web generations, see Table 1.

3.3 Aligning public values and public value with changing public governance paradigms

Table 3 provides a cumulative chronological overview and summary of public governance paradigm characteristics constructed directly from the detailed evidence and sources provided in the accompanying stateof-art report. It shows the nine public governance paradigms in rows arranged chronologically, plus summaries in columns A to I of important characteristics of each. Of particular interest are column C which summarises which of the nine European public values are addressed by each paradigm, column D shows the changes in public value, column E the digital technology used and Column I the **framework conditions and influencing factors**.

It is very clear from column C in Table 3 that the number of European public values incorporated increases over time but with two important exceptions against this trend that show significant reductions in the numbers. First the introduction of NPM from about 1980 reduces the focus to just two dominant public values, i.e. being efficient and the rule of law. Although the rule of law is clearly present in NPM in a European context one of NPM's weaknesses is that it can sometimes be difficult for other actors to see this given the paradigm's opaqueness, for example when outsourcing and partnership contracts are typically hidden because of 'commercial confidentiality'. The primacy of efficiency in NPM is related to its raison d'être that market principles are the best way to operate more or less any organisation or institution and that public service users should be viewed as customers with a purely transactional relationship with the public sector. NPM's highly focused entrepreneurial culture views outsourcing, a focus on cost-cutting efficiencies and 'value for money' principles, as the most important for good governance. Second, the lean and austerity public governance paradigm was rapidly introduced in the aftermath of the global economic and financial crisis of 2007-8. However, rather than just two dominant public values of being efficient and the rule of law as in NPM, leanness also includes being accountable and transparent. This more nuanced public values portfolio is likely due to the fact that the Neo-Weberian governance paradigm appears after NPM but before learness and adds additional public values. In addition, it is almost certainly also related to the introduction of Generation 2.0 and 3.0 modes of government and web technologies which have opened up governance to a considerable extent immediately before and after 2008.

In terms of public value, the Weberian, NPM and Neo-Weberian public governance paradigms establish the basic building blocks of welfare state services at level 1 public value. They do so in different ways, however, related to the changing framework conditions, first of post-war austerity, followed by the 1970s financial crisis and then Neo-Weberian attempts to find a balanced 'third way' between 'small' and 'big' government. The immediate post-2008 lean governance paradigm undoes this balancing attempt and reverts to 'small' government policies.

After about 2000 the networked and public value governance paradigms both come to the fore, again as reactions to NPM as well as to the Neo-Weberian lack of innovation, but also enabled by the arrival of Generation 2.0 government and web technologies. This digital technology revolution was clearly a major turning point as it enabled interactive rather than one-way governance, social media and completely new types of public services as shown also in Table **2**. It is noteworthy that the networked and public value governance paradigms appear at about the same time with the same technological opportunities and societal challenges, but represent somewhat different political, policy and cultural responses to these conditions, showing **the paramount importance of the type of political and policy responses enacted**. After 2008's financial crisis, but also again driven by further digital technology revolutions (Generation 3.0), both lean and open governance emerge as two very distinctively different, and in many ways contradictory, responses to the same societal challenges and technological opportunities. Similarly, from about 2015, in the context of Generation 4.0 technology, the sustainable and locality-community paradigms emerge as distinctively different responses but, this time, as largely complementary.

These changing public values of good governance are also reflected in the types of services delivering public value. From about 2000, the network and public value governance paradigms in the context of Generation 20 interactive technologies start attempts to join-up and coordinate services. However, it is only with the introduction of the open governance paradigm and Generation 3.0 semantic technology that such joining-up attempts, through a life-event and one-stop-shop focus, become more mainstream, although still have only very limited success in breaking down silos. From about 2015, however, further political and policy changes, alongside technology Generation 4.0 with possibilities to better link distributed data that the breaking down of silos starts to have some real success in the context of the sustainability and locality-community paradigms that can deliver public value level 2.

Table 3. Cumulative chronological overview and summary of public governance paradigm characteristics

Public governance paradigm (dates)	A) Purpose	B) Governance structures, norms & cultures	C) Good governance public values for delivering public value	D) Public services contributing to public value	E) Digital technologie s (ref Table 1 & Figure 2)	F) Role of public sector in public governance	G) Role of other actors in public governance	H) Weaknesses	I) Framework conditions & influencing factors
1) Traditional public (Weberian) administration (from before 1945)	Authority & control; rule of law; professional; politically neutral; objective: central role of the state	Top-down hierarchy; centralised; internal orientation; clear rules; functional specialisation; standardised procedures; culture of civil service ethics	1) Accountable 5) Efficient 7) Rule of law	Increasing provision of all basic European welfare state services, incl. health, education, transport & housing	Only from mid- 1990s, G1.0 Industrial Rs: 2IR & 3IR	Monopoly on policy design & implementation, on service design & delivery & on public value creation	Formally minimal, but in early period services, later seen as 'public' still delivered by private & civil sectors	High implementation & monitoring costs; lack of useful know-ledge; rigid one-size-fits-all solutions so many not served leading to dissatisfaction	Post-1945 initial austerity, & later very strong growth; Keynesian interventionist 'big' government; welfare state starts but baby boom increases service pressure.
2) New Public Management (NPM) & market-based (from about 1980)	Operate as a business; market focus; treat users as 'customers' with private sector management who get 'choice'; cost reduction; targets & measurement	Decentralised & competitive; 'small government'; moves to privatise, reduction of government control; outsourcing & competition; culture of entrepreneurialism.	5) Efficient 7) Rule of law	Increasing role of non-public actors for all basic European welfare state services. From mid 1990s online service access.	Mainly G1.0, later G2.0 Industrial Rs: 2IR & 3IR	Significant outsourcing, 'arms-length' oversight & financing of governance functions & public services for implementation by private sector	Huge role for private sector in competing for and implementing many governance functions & public services.	Short-termism; budget rigidity put before ser- vice quality: opaque ethics; possible corruption; less democratic account- ability due to 'com- mercial confidence'; over reliance on KPIs	In wake of supply-side energy shocks of 1970s, Milton Friedman Chicago School supply-side economics; hands-off 'small' government; privatisation & market competitiveness; boosted by the fall of communism from 1990 when neo-liberal worldwide assumed the 'end of history'
3) Neo-Weberian State (from late 1990s)	As Weberian but more professional, responsive, open & results focused; some networking within public sector & with other actors	Some reversion to Weberian plus both more internal centralisation & external orientation with focus on meeting user needs; culture focused on increasing service quality & responsiveness.	1) Accountable 3) Responsive 5) Efficient 7) Rule of law	Greater state service provision; start of digital interactive user-centred services & user generated content.	Mainly G1.0 & later G2.0. Industrial Rs: 2IR & 3IR	Dominant actor for policy design & implementation, for service design & delivery & for public value creation	Less dominant as service providers cf. to NPM; but start of feedback from cit- izens & businesses as service users but not initially digital	Danger of some reversion to Weberian weaknesses, and emergence of digital divide problems.	Reaction against NPM in many countries with some retention / reintroduction of public interventionism; focus on a 'third-way' in politics between 'big' & 'small' governance
4) Networked governance (from about 2000)	Focus on formal, systemic network relationships & coordination across public sector & with other actors	Networking technology improves efficient & effective interconnection of public entities & multi- levels, & with non-public actors; culture of collaborative decision- making	1) Accountable 3) Responsive 5) Efficient 6) Effective 7) Rule of law	Increased service coordination & targeting between public entities & with other actors	Mainly G2.0. Industrial Rs: 2IR & 3IR	Facilitating network actor for coordinating different public entities, & with other actors, in service design & delivery & for public value creation	All other actors better able to find niche roles for different public services at different governance levels	Huge digital & mana- gerial competence challenges amongst all actors; not addressing 'wicked problems' sufficiently; start of digital divide; some technology hype	'Network society'; 'wicked' & cross-boundary problems; societal individualisation & horizontalisation; economic growth & globalisation; 2000 dotcom crash; 2001 9/11 terror attacks in New York.
5) Public value governance (from about 2000)	Government recog- nises itself as the prime mover in creating public value through its unique role as supporting interests of all actors	Systems of dialogue & exchange of multiple actors led by government through its organisational & managerial capacity, combining culture of all actors creating value together.	1) Accountable 2) Transparent 3) Responsive 5) Efficient 6) Effective 7) Rule of law 9) Consensus	Start of service focus on maximising value for users via consultation and joint value creation between all actors.	Mainly G2.0. Industrial Rs: 2IR & 3IR	Managing & orchestrating all actors through consultation & collaboration to create value for each based on political legitimacy	Motivated by their involvement in networks and partnerships, especially with pub- lic sector to join in creating their own & public value	Big strain on public leaders & managers; how to measure; actor consultation not direct involvement; silos try to cooperate but still follow own interests	Relatively 'big' government creating public value for all actors seen as necessary to address 'wicked' societal problems; accelerating economic growth & globalisation

Source: The author (Millard, 2023).

Table 3 (continued). Cumulative chronological overview and summary of public governance paradigm characteristics

Public governance model (dates)	A) Purpose	B) Governance structures, norms & cultures	C) Good governance public values for delivering public value	D) Public services contributing to public value	E) Digital technologies (ref Table 1 & Figure 2)	F) Role of public sector in public governance	G) Role of other actors in public governance	H) Weaknesses	I) Framework conditions & influencing factors
6) Lean and austerity (from 2008)	Continue basic NPM tradition but even greater focus on cost savings & service cuts; doing 'more with less'	As NPM but even stronger emphasis on outsourcing & 'arms- length' oversight of civil & private sector actors taking public tasks via networking; some collaborative culture	1) Accountable 2) Transparent 5) Efficient 7) Rule of law	Austerity scales down provision of basic welfare state services with some slack taken by other actors, especially civil society	Mainly G1.0, G2.0 & G3.0 Industrial Rs: 2IR & 3IR	Reversion to NPM with, even greater efforts to outsource but also collaborate with, & orchestrate roles of, other actors to create public value	Esp. civil society pushed to takeover more responsibility for social public services that private sector unwilling to do, whilst still outsourcing more lucrative services to private sector	Even less ability to directly address 'wicked' societal challenges than NPM as attempt to out- source much of this responsibility despite private sector often unwilling & civil sector often not able	2007-8 global financial crisis creating complex political, economic, managerial, cohesion & democratic challenges leading to a period of austerity, small gov & growing inequality & poverty
7) Open governance (from 2008)	Gov opens its processes, services & data, incl. decision & policy making inviting all other actors to do the same to create societal level public value for all; doing more by leveraging more.	Networking & platform structures become dominant fostering multiple governance structures & cultures, through transparent, participatory, inclusive, co-creative & behaviour-al/cultural gov.	 Accountable Transparent Responsive Inclusive Efficient Efficient Effective Rule of law Participatory Consensus Add 'open' 	Semantic platform based personalised & co- creative services via participation & max inclusion; e-Signature & e-Identity; gamification; once-only; digital-by- default; policy modelling; life events & one-stop- shop attempting to break down silos; joined-up, proactive & predictive.	Mainly G3.0 & then G4.0 with G2.0 still present Industrial Rs: 2IR, 3IR & then 4IR	As public value (& some lean) governance, with public sector becoming responsible for orchestrating public value but recognising it does not have monopoly on assets, knowhow & wisdom to do so.	Critical role of private, civil & education actors to contribute their assets & knowhow to create their own value within the overall public value context.	Lack of interoperability; reluctance to share data; lack skills & competences; openness risks data security; blurring of accountability & of how quality is ensured; new digital & socio- economic divides	Obama's 2007-8 crisis different response to lean with a relatively big gov approach; leading to strong economic growth but also contributing to the increase in inequality & poverty as globalisation, outsourcing & long supply-chains accelerate in importance.
8) Sustainability governance (from about 2015)	Society needs to make all its activities sustainable, socially, economically & environmentally in an integrated & nexus manner; meet needs of society both today & tomorrow.	Public sector is pivotal whilst working with all actors at all levels through networks & platforms in sustaina- bility culture including for circularity & nature- based approaches	As open governance. Add 'sustainability'	As open governance, plus focus on nature monitoring, resilience & regeneration services with attempts to break- down silos only partially successful.	As open governance	As open governance & public sector taking lead, including for policy & regulation, on all sustainability issues at all levels of government & across all sectors & parts of life; identify, reduce & assign respon-sibility for externalities	Variable roles; at minimum support public sector's lead, but also take own lead in contexts where actor is active or has specific interest.	Nature not given big enough voice; still seen just as source of eco- services & natural resources, with focus on these functions paramount rather than nature's own interests; problem of shifting baseline syndrome	As open governance, plus especially after 2015 UN SDGs & climate change agreement & corresponding EU initiatives.
9) Local & community governance (from about 2015)	'New localism' aims to retain as much value generated locally within the locality; not in isolation but based on networking, interconnectivity, local participation & democracy.	Interacting multi-level structures where local & community author- ities exercise subsidia- rity power & access resources relevant for their function; culture of local responsibility in national, historical & cultural context.	As open & sustainability. Add locality & community	As open & sustainability governance, but mainly focused on local and city levels where joining up silos easier than at national level	As open governance	As open governance, & national public sector providing open struc- tures & frameworks for localities to thrive, & local public sector often taking lead role locally & a strong enabling role to maximise & retain local public value.	Variable roles; often community, some- times firms, taking lead roles as they have extra resources, know- how & energy to innovate; e.g. sweet- spot of smart cities & transition towns.	Poor locality gov can lead to national postcode lottery & complex inoperable solutions, loss of scale economies & other benefits; local fiefdoms & isolationism; bad externality impacts on near/other localities.	As open governance & new devolved powers in many countries; new urban & city-region agendas; new focus on rural areas & rural-urban relationships; especially cities & neighbourhoods gaining more power & recognition; many city cooperation networks.

Source: The author (Millard, 2023).

A noteworthy observation of changing public values as new public governance paradigms appear is that being efficient and the rule of law are the only public values for good governance that run through the whole period from 1945, clearly reflecting the main underlying principle of Weberian administrative governance which has remained the bedrock of European governance since. As mentioned, apart from the significant reduction in the variance of public values in NPM and the smaller regression during the lean and austerity period, the number and diversity of public values addressed by successive public governance paradigms increases chronologically. It is also noteworthy that from about 2000 and Generation 2.0 technology, a very large number of new public governance paradigms and models was enabled that are clearly interdependent, often mutually reinforcing but sometimes also contradictory. Apart from the technology, these were also enabled by significant changes in the framework conditions, as summarised in column I of Table 3. Based on the evidence in this and the accompanying state-of-art report, these post-2000 paradigms were able, at least in principle but often in practice, depending on the specific focus of each, to start developing many more public values.

It is also relevant to observe that another **important change around the year 2000 sees paradigms and models switch from being mainly process and organisationally focused**, and labelled as such by the actors involved as well as by researchers, **to having a specific public value and societal focus**. Post-2000, labelling began to be directly conceived to address important societal challenges, enabled and driven by the fact that digital technologies have become powerful enough for public governance to exploit them in a myriad of significant ways. Before about the year 2000, there were only a few standard paradigms and models, but in the years since there has been a **mushrooming of novel and experimental trends and possibilities enabled by the technology but also driven by changes in society, politics and policy**.

As mentioned, apart from the lean and austerity retrenchment, all of the post-2008 public governance paradigms, covering openness, sustainability and locality/community, ostensibly address all ten public values, clearly also related to the increasing effectiveness of digital technology based on Generation 3.0 and then 4.0. It can thus be deduced that this period saw the fulfilment of all European good governance objectives. However, these were consolidated in the period up to 2017 and **it is clear from the evidence of the open, sustainability and locality/community paradigms that there is a need for agreeing additional European public values that reflect these paradigms.** Indeed, the 2017 consolidation of public values (European Commission, 2017) does hint that sustainability, both in terms of making government itself more sustainable but also related to problems in the use of finite resources, impacts on the natural environment and climate change, could be a relevant public value. However, neither locality nor community are promoted as possible public values in this document.

4 Strategic challenges facing public governance from 2020 onward

This part of the report takes the story of existing European public governance paradigms narrated above into the post-2019 present and future by attempting to paint a credible set of pictures about Europe's current and near future public governance challenges up to at least 2030. This and subsequent sections will also need to be seen in the context of Europe's Digital Decade to 2030 (European Commission, 2021k), and draws upon recent work by the JRC on the Future of Government 2030+ (Vesnic and Scapolo, 2019; Vesnic et al, 2019). Four overarching strategic challenges are identified for public governance in this report, each of which is exemplified through a number of potential paradigms and models. As was clear with the nine existing public governance paradigms identified above, and although there is chronological change driven largely by political, socio-economic, environmental and cultural factors and events where digital technology and data are best seen as necessary but never sufficient tools, a new paradigm does not replace earlier ones but rather cumulatively adds to the mix of those being used. Clearly, the most recent paradigms tend to attract most attention and, if seen as beneficial, become widely rolled out, but most if not all earlier paradigms typically continue to play important roles and exert significant influence. This is also due to the fact that, although some paradigms are in conflict with each other (e.g. as seen in the switches between 'big' and 'small' government as summarised in Table 3, most are complementary, so that it is the actual 'mix that matters' (Bevir, 2013) in any time and place

In contrast to **the 2007-08 financial crisis** characterised by a massive demand slump due to dramatically reduced consumer spending power, **COVID-19 hitting Europe in early 2020** caused a severe supply-side recession. This is being turbo-charged by **the invasion of Ukraine during 2022**, depositing a thick layer of geo-political tectonic change on top. Underlying all this is **the 'mother-of-all' crises that sees our natural environment stretched to near collapse**, thereby putting the very survival of our species in peril. All this has disrupted both global and local supply chains as much of the workforce becomes used to virtual, remote and hybrid working, restrictions on the movement of people and goods with transport and logistics put under severe

pressure. At the same time, demand for many goods and services mushrooms in Europe as elsewhere as pandemic restrictions are lifted.

The economic contraction caused by the COVID-19 pandemic in 2020-21 left a heavy health and human toll, shrank the Eurozone economy by a record 12.1% and wiped out more than a decade of expansion during the largest economic shock the world has experienced since 2008 (Elliot, 2020). A further sharp tightening of the economic screw began in early 2022 with the Russian invasion of Ukraine. The International Monetary Fund (IMF) drastically downgraded its growth forecasts, predicting further global economic fragmentation, rising debt and social unrest (IMF, 2022). The World Bank stated a "human catastrophe loomed" with an estimated unprecedented 37% rise in food prices, caused by war-related disruption to supplies, pushing millions into poverty, increasing malnutrition, and reducing funding for education and healthcare for the least well-off (BBC, 2022). By April 2022, more than five million people had fled Ukraine in two months, with more likely to follow, exacerbating an international migration emergency that extends from Afghanistan to the Sahel (Aljazeera, 2022). In drought-hit east Africa, the World Food Programme says 20 million people face starvation during 2022. The war in Ukraine did not create the drought, but the UN warns it could hurt efforts to reduce global heating, thereby triggering further displacement and forced migration (Harvey, 2022).

It is clear that **the 21st Century has ushered in a new age of more or less continuous crisis and disruption** and that these are not times for business-as-usual. It is time for rethinking many of our shibboleths, both sustainable development and resilience, how we re-structure our economies and politics, as well as how we work, play and live on the earth's surface. These are huge issues, intimately inter-related, and in which public governance and digital data and technology clearly play essential roles.

Referring to Figure 1, four strategic challenges are identified which new public governance paradigms will need to address through a rich set of diverse experiments which, although mainly in the future and somewhat tentative, are based on tracing very recent developments and commentary. They are likely to be strongly interrelated given they arise from, or at least are significantly accelerated by, turbulence and uncertainty. Although some of these challenges initially kicked-off in the aftermath of the 2007-8 economic and financial crisis, they are now much more acute. It must also be stressed that the four identified strategic challenges are not proposed as an exhaustive list as others are likely to arise during on-going turbulence. Each challenge also suggests several likely public governance paradigms and models designed to increase public value and that build on, complement and are perhaps sometimes in conflict with, the public value delivered by existing public governance, as summarised in Section 3.

Strategic challenge 1— Living with turbulence: open strategic autonomy; security and risk; societal resilience.

Strategic challenge 2— Changing geographies: European and global geo-politics; re-ordering settlements and communities; turbo-charged localities and communities.

Strategic challenge 3—Hybrid human-centred relationships: preventative public services; 'community-based solutions'; justice, fairness and inclusion by design.

Strategic challenge 4— People-planet systems: quintuple helix; from sustainability to resilience to regeneration; dynamic systems

Each of these four strategic challenges, together with their suggested public governance paradigms and models, are proposed for further experimentation testing, validation and refinement. In relation to digital technology, the question for experimentation would be what role does current and future technology have in supporting, improving and indeed extending the public value of these paradigms and models?

4.1 Strategic challenge 1: Living with turbulence

4.1.1 Open Strategic Autonomy

Open Strategic Autonomy is a relatively new EU concern designed to ensure that it is able to internally source most strategic goods and services, with the rest via 'friend-shoring' from allied, stable and reliable democratic countries. This has been developed in order to support Europe's post-coronavirus recovery programme (European Commission, 2020c), part of which is a review of the EU's trade and investment policy (European Commission, 2020d). "Two key objectives are driving this process. First, to assess how trade policy can contribute to a swift and sustainable socio-economic recovery, reinforcing competitiveness in the post-Covid 19 environment, addressing the challenges the EU will face, and helping to promote our values and standards. Second, to see how trade policy can help build a stronger EU based on a model of 'Open Strategic Autonomy'

reaping the benefits of openness for our businesses, workers and consumers, while protecting them from unfair practices and building up our resilience to be better equipped for future challenges." This policy review is designed to set the political direction for EU trade and investment policy in the years to come.

In 2021, the JRC carried out a foresight process to look at Open Strategic Autonomy in a systematic and systemic way, encompassing different dimensions in a holistic manner by presenting an overview of Europe's existing capacities, dependencies and vulnerabilities through synthesising existing knowledge on the current state and future possibilities in 2040 and beyond. This also encompassed examining trends and emerging issues, looking forward at how they could evolve over time, and looking at the opportunities and risks they entail. The report highlights ways the EU can start to seize the benefits from positive developments and ways to transform risks into potential for positive transformation (Cagnin et al, 2021).

According to Van den Abeele (2021), the President of the European Council, Charles Michel, dubbed 'open strategic economy' goal No. 1 for our generation: "For now, much of the debate on strategic autonomy has been on semantics, marking clear political divides between countries that see this as a threat to Europe's openness, and its proponents who see a chance to advance European industry and help foster 'champions' able to compete on a global scale." The concept was endorsed by the European Council of 1 and 2 October 2020, was referenced at the European Council of 26 March 2021, and at the same time needs to be seen in the context of the EU's green and digital transition and the resilience of the EU's socio-economic model in the aftermath of the pandemic, and now even more critical in light of the Russian invasion of Ukraine. Other countries, including close allies, are also pursuing their own forms of strategic autonomy. For example, France and Germany as the EU's two leading economies, are pushing for tougher industrial policies such as more state subsidies for European businesses, to counter the threat from U.S. reforms that risk triggering a transatlantic trade war. They have issued a joint statement vowing to "explore industrial policy possibilities" to safeguard European industries from discriminatory trade measures from Washington and also Beijing, signalling an escalation of European efforts to protect home-grown manufacturing from the threat of unfair competition as the era of free unfettered trade comes to an end. (Politico, 2022)

The European Parliament (2022a) now regularly reviews the EU's progress towards open strategic autonomy, with the latest in July 2022 examining post 2019 developments: in 2020 the Covid-19 shock in relation to supply chain vulnerabilities; in 2021 in terms of 'one concept under many names' showing that this is not a zero sum game but rather a sliding scale between full autonomy and full dependency with different results for different policy areas; and the latest in 2022 in reaction to the Ukraine shock, translating words into action, that brought the debate back to hard realities with the need to react with concrete, practical action. Thus, **open strategic autonomy** is **now for Europe more than just a goal but rather a strategic necessity**. In addition, according to the European Parliament (2021a), an important pillar is that "the EU should now strive for '**open digital autonomy**' by defining well integrated policies that combine more funds, regulation when needed but without excess, global strategic partnership and alliances, and a new and innovative approach to digital diplomacy that leverages European strengths also in areas such as the setting of global standards". In particular, although Europe has many current digital strengths, such as in software services, there is a need to address the EU's continuing dependence on non-European technology providers.

4.1.2 Security and risk

The WEF's latest Global Risks Report (WEF, 2022a), was published shortly before the Russian invasion of Ukraine in February 2022 but this has only further accentuated the risks highlighted. For example, COVID-19 and its economic and societal consequences continue to pose critical threats, further increasing inequality whilst uneven **economic recovery risks** compounding social fractures and geopolitical tensions. At a global level, this complicates the coordination needed to tackle the most common challenges identified as **societal and environmental concerns**, including strengthening climate action, enhancing digital safety and tackling cyber vulnerabilities, restoring livelihoods and societal cohesion, managing migration and addressing competition in space. Since the start of the pandemic in early 2020, societal risks in the form of social cohesion erosion, livelihood crises and mental health deterioration, have worsened the most. For the next three years, expectations are seen as either consistent volatility with multiple surprises or as fractured trajectories that will separate relative winners from losers. For the next five years, societal and environmental risks are seen as the most pressing, but over a 10-year time horizon, the health of the planet dominates all other concerns, especially climate action failure, extreme weather and biodiversity loss. Technology risks are also seen as increasing, especially digital inequality and cybersecurity failure (WEF, 2022a).

In addition to these global risks that also strongly threaten Europe, important European risks over the last 2 years have included threats to the continued provision of financial services, fiscal stability and the

resilience of the financial system. However, to date, this risk has been successfully addressed primarily due to the continued supply of credit (European Systemic Risk Board, 2022). However, the Russian invasion of Ukraine now threatens to undermine this success, especially given the uncertainty as to how long it will continue, as well as the certainty that European politics and security have already been dramatically re-set regardless of the war's duration. According to FERMA (2022) the top 5 risks in Europe within the next 12 months are cyber threats, supply chain or distribution failure, geopolitical uncertainties, uncertain economic growth and overregulation. Over the next 3 years, the top 3 risks are seen as changing customer behaviour, cyber threats and uncertain economic growth, whilst the top 3 over the next 10 years are perceived as climate change and environmental damage, plus changing customer behaviour and natural disasters. It is clear from this that in the short to medium term, cyber threats and uncertain economic growth stand out, to be replaced in the top three by climate and environmental threats and resulting natural disasters over the long-term. This long-term risk perception is similar to that seen globally.

In terms of **cyber security**, EU-wide legislation first started in 2016 when it launched the Network and Information Systems (NIS) Directive with some success, and this was followed in 2019 when the European Council supported a concerted approach to the security of 5G network European Parliament (2021a). In 2020, the JRC reported on cyber security as Europe's 'digital anchor' by providing multidimensional insights into the growth of cybersecurity over the last 40 years, identifying weaknesses in the current digital evolution and their impacts on European citizens and industry. The report also set out the elements that potentially could be used to shape a brighter and more secure future for Europe's digital society, taking into account the new cybersecurity challenges triggered by the COVID-19 crisis. According to some estimates, cybercrime cost the world EUR 55 trillion during 2020, up from EUR 2.7 trillion in 2015, due in part to the exploitation of the COVID-19 pandemic by cyber criminals. This figure represents the largest transfer of economic wealth in history, more profitable than the global trade in all major illegal drugs combined, putting at risk incentives for innovation and investment (Baldini et al, 2020).

In November 2022, the European Commission put forward a Joint Communication on an EU Cyber Defence policy and an Action Plan on Military Mobility 2.0 to address the deteriorating security environment following Russia's aggression and to boost the EU's capacity to protect its citizens and infrastructure (European Commission, 2022c). Given that cyberspace has no borders, there is a need for the EU to enhance cooperation and investments in cyber defence to better protect, detect, deter, and defend against a growing number of cyber-attacks on energy networks, transport infrastructure and space assets that pose risks to both civilian and military actors.

This builds on the EU's 2020 cyber security strategy that aims to address the fact that the EU's economy, democracy and society depend more than ever on secure and reliable digital tools and connectivity, so that cybersecurity is essential for building a resilient, green and digital Europe. This threat landscape is compounded by geopolitical tensions over the global and open Internet and over control of technologies across the whole supply chain, including for electronic components, data analytics, cloud, faster and smarter networks with 5G and beyond, encryption, AI, and new computing and trusted data processing paradigms such as blockchain, cloud-to-edge and quantum computing. Of particularly high importance are the malicious targeting of critical infrastructure and security threats as a major disincentive to using online services. Improving cybersecurity is therefore essential for people to trust, use, and benefit from innovation, connectivity and automation, and for safeguarding fundamental rights and freedoms, including the rights to privacy and to the protection of personal data, and the freedom of expression and information (European Commission, 2020e).

4.1.3 Societal resilience

In the aftermath of COVID-19, there is a need for a re-evaluation of global value chains given that the market is already contemplating **the move from a globalised to a more de-globalised world**, although not back to the same place before the last 20-30 years of massive globalisation. This will involve some new on-shoring and friend-shoring economic activity and employment and a re-focus back to more domestic markets in new forms of re-localisation, also down to regional and especially city levels, although it is important to retain openness, connectivity and cooperation at all levels including globally. This is likely to be seen especially in relation to 'strategic' manufactured goods, such as health and other vital goods, as well as in critical infrastructures, and increased focus on more domestic and regional sources in order to diversity supply towards greater resilience at times of future shock. This is already taking place to some extent, for example through more circular economies and a strong move to additive manufacturing using tools like 3D printing which is able to use waste from one industrial process as the feedstock for another, rather than importing new raw materials from the other side of the globe. Supply chains have considerably shortened during the pandemic and, although

there will likely be new rounds of re-globalisation, the advantages of better control through a stronger focus on resilience, reductions in environmental footprints and some re-shoring of jobs in the context of shifting geopolitical trends towards more inward-looking global blocs, is likely to continue to be priortised.

A fundamental issue is the **balance between economic efficiency** on the one hand, which has been the prevailing just-in-time mantra but increasingly seen as a short-term strategy, and economic resilience on the other, that also promotes societal and environmental wellbeing, but typically requires a medium- to longerterm perspective and a just-in-case approach. When efficiency is prioritised over all other requirements of an economic system, this tends to become 'hard-wired' so pressure is on to eliminate all so-called slack in the system, for example by eliminating stock in hand for productive industry, financial reserves for banks and additional human resources in all types of economic activity, given that each of these adds to organisational costs thereby reducing efficiency. However, this type of efficiency leads to so-called 'economic brittleness' there is no problem when the economic environment is highly predictable and changes very slowly, but when subject to shock there is no slack to respond in a robust manner. According to the Ellen MacArthur Foundation (2013), successful companies and successful economies require a "tradeoff between efficiency and resilience" and "that economic resiliency is based on economic diversification and complexity". According to a paper prepared for the UN in the summer of 2020 in the wake of COVID-19, these observations became even clearer (Millard, 2020), while an OECD paper in the same year (OECD, 2020c) observed that when "aspiring for maximum efficiency and optimisation, systems have neglected resilience against disruptions whose shocks may leave governments, the public, and the environment in a weakened state".

As shown in Figure 2, in order to optimise sustainability (defined as the ability for successful survival of any system over the long-term) there is a trade-off between efficiency (and brittleness), on the one hand, and resilience through diversity and interconnectivity (adaptive to shock but slow to change radically), on the other. To optimise the system requires finding the 'sweet spot' between too much efficiency, risking large scale breakdown from time to time as seen with the shock of COVID-19, and too much resilience, which risks stagnation and struggles to accept radical change. Most economies have so far failed to achieve let alone recognise the need for an 'optimal balance' where both are in play. This is seen in the prevailing focus on short-term shareholder, as opposed to shared or stakeholder, value (Porter and Kramer, 2011), 'lean' companies and value chains and indeed 'lean' governments (Millard, 2015a) and arguably an over-reliance on just-in-time supply chains with no flexibility or ability to find alternative sources in case of disruption. The economic mantra has generally been one of 'squeezing assets', including human assets, and squeezing value chains to maximize quarterly returns whilst leaving no margin for slack in case of system shock. Since the 2008 financial crisis, too many countries have had an obsession with efficiency to the extent that investing in infrastructure has been neglected. Cost-cutting has been given a higher priority than capacity building.

As a result of the economic and social disruptions caused by the pandemic, many international organisations, as well as governments, are now starting to advocate building resilience much more strongly into global economic systems. For example, the World Economic Forum's focus on the 'Great Re-set' (WEF, 2020), which also recognises that in the context of the three interconnected pillars of sustainable development in the UN's 2030 Agenda (United Nations, 2015b), resilience needs to be seen not only in narrow economic terms, but also as social and environmental resilience. It is clearly more 'efficient' over the medium- to long-term, also in narow economic terms, to focus on the joined-up resilience of all three pillars of the social, the economy and the environment, because if one fails all fail.

Successfully addressing turbulence also requires a **resilient society in which social structures and conditions are both diverse and interconnected**. Diversity is important for innovation, to promote thriving cultures and to support multiple values and identities through 'unity in diversity' (Scharfbillig,et al, 2021). However, without hybrid interconnectivity (both digital and human-to-human in multiple ways) with policies that promote inclusion, cohesion, justice and fairness, this can lead to atomisation, separateness and even conflict. Diversity should not mean inequality. For example, the huge inequalities within as well as between countries illustrate the pernicious effects on societies: eroding trust, increasing anxiety and illness and encouraging excessive consumption. It is clear that across a wide range of health and social problems (physical health, mental health, drug abuse, education, imprisonment, obesity, social mobility, trust and community life, violence, teenage pregnancies, and child well-being) that outcomes are significantly worse in more unequal countries, whether rich or poor (Wilkinson and Pickett, 2009). Other research has shown conclusively that the global economy has developed to such a point that greater income is being derived simply through the continuing increase in the value of already accumulated wealth assets rather than through income derived from work in the labour market where this wealth is actually produced. According to (Piketty, 2014), trickle-down economics has been entirely discredited both as a theory and as a real phenomenon.

Figure 2. Balance between efficiency and resilience



Source: Ellen MacArthur Foundation, 2013.

Poverty and inequality have also been growing dramatically in Europe, especially since the 2007-8 financial crisis and even more since 2020 with the turbulence induced by the pandemic and the war in Ukraine. In 2020, 96.5 million people in the EU were at risk of relative poverty or social exclusion; this was equivalent to 21.9 % of the EU population. The risk of poverty or social exclusion in the EU was higher in 2020 for women than for men (22.9% compared with 20.9%). Neither is poverty evenly distributed. Romania has Europe's highest risk of in-work poverty with a rate of 18.9%. Spain and Greece follow with 13.1% and 14.1%, respectively. Additionally, the in-work poverty risk has increased from 8.3% in 2010 to 9.6 percent in 2016 (Eurostat, 2021a).

McGarvey (2022) looked at relative poverty in the UK and recognised the huge consequences of a lack of interconnectivity through the "proximity gap between the powerful and the powerless as the root cause of many of society's ills." In education, health, housing and benefits there is a fundamental and maintained distance between those who make the policies and the people at the bottom of society who are on the receiving end of them. This is a gap in experience, understanding, culture and, above all, in empathy. McGarvey shows through his analysis of why and how most politicians, as a select group of people with very limited experience of social inequality, has been charged with discussing and debating it and making the rules and structures that surround the poor and vulnerable. The gap to those at the bottom of society, with the lived experience of the ramifications of these structures, policy and decisions, has been both revealed and turbo-charged during COVID-19. This lived experience needs to be coopted in decision- and policymaking as it was in the design of the original welfare state in the UK, but this has not happened since.

Digital divides directly relate to existing socio-economic divides, so that both reinforce each other (Millard 2015b), and the uneven roll-out and use of digital technology can exacerbate these inequalities (Oxfam 2016). This is not a given outcome of the technology, but instead derives from the governance, institutional framework, regulatory, labour market and wider economic system. Europe is also struggling to keep up with the digital revolution with a 2019 survey showing that 43% of the adult population in Europe had limited digital skills, and 37% had no digital skills at all. Meanwhile, young refugees and asylum seekers are experiencing a crisis of their own in Europe, finding it very difficult to integrate socially and in the workplace (European Commission, 2019a).

In **digital government**, the drive to be more efficient can push efforts to 'hardwire' processes which can reduce agility and flexibility. To counter this, for example, data-sharing approaches can provide access to information in a range of preferred standardised approaches to readily enable reuse, whilst open-source technology can be important for this reason in some 'business' contexts. Being resilient is also about planning and being prepared. for example, for a pandemic, given that even though the risk of this happening may be low the costs are extremely high if it happens. 'Continuity' is also a keyword for resilience. The Internet (ARPANET) was built so that communication could still work even if nodes were taken out due to war, sabotage, accident or natural disaster, as other nodes could continue to operate in a decentralised but interconnected network. From the oovernance perspective, it is the overall goal that is important, whether this be sustaining European values, the economy, well-being, etc., but this means choices will need to be made given that when such disruption occurs this allows for some evaluation to take place that, in turn, can improve efficiency. In the context of turbulence and disruption, it is also important to ask what sustainability means for the different actors, who wins and who loses, and who decides what is a sustainable/social/valuable interaction? The European Commission (2022d) has established resilience dashboards aimed to assess the vulnerabilities and capacities of the EU and its Member States in each of four dimensions (social and economic, green, digital and geopolitical), taking a forward-looking perspective informed by strategic foresight. The important role for resilience governance in the post-pandemic world is also explored by Brown (2022) and Bonime-Blanc (2022) with a focus on businesses.

4.2 Strategic challenge 2: Changing geographies

4.2.1 European and global geo-politics

Prior to 2020 there have been significant geo-political shifts that have impacted public governance, especially the break-up of the Soviet Union in 1989-90 and the 2001 9/11 terror attacks on New York that turbo-charged technology innovation. In particular, this latter event dramatically changed the discussion away from the rights of individuals and companies to keep their online activities away from any government scrutiny, ostensibly to uphold democracy and free-speech, towards intensive government surveillance that aimed to gather enough intelligence to counter further terror threats (Zuboff, 2019). Other important events included the change in US President in 2016 and the Brexit referendum in the UK, both heavily influenced by new social media.

However, the post-2019 shocks have perhaps been the most important for changing geo-politics and their impact on the EU. The rise of China as a world power and changes in South America, Africa, South Asia and the Pacific have also exacerbated geo-political differences and strained as well as strengthened spheres of influence and geographic blocs. In particular, the Russian invasion of Ukraine has had a considerable unifying pull on the EU and NATO and a strengthening of trans-Atlantic relations with a new US President from early 2021. According to the European Commission (2022h), **"the world is experiencing tectonic geopolitical shifts**, reinforcing the megatrends already affecting the EU. The long-term implications of Russia's military aggression against Ukraine, including for energy, food, economy, security, defence and geopolitics, will clearly affect Europe's path to achieving fair green and digital transitions. However, these and other future challenges will not divert the European Union from its long-term objectives. With the right set of policies, they can serve as a catalyst to speed up achieving them. Ultimately, this could foster our resilience and open strategic autonomy in various areas, from energy, food, security and critical supplies – including raw materials needed for the transitions – to cutting-edge technologies."

These shocks have also considerably sharpened both the perception and reality of **political differences between the so-called 'democracies' and 'autocracies', although a more nuanced dichotomy might be between countries that support a stable world order and the rule of law and those that do not**, given that this does not map directly onto whether or not a country is 'democratic' as defined by Freedom House (2022). The number of 'democratic' countries defined using this definition has fallen by half since the late 2000s and is now down to less than 20%.

The EU is attempting strategically to navigate this currently highly volatile landscape, both proactively on the front-foot as well as with intelligent reactivity when unexpected shocks appear as experienced at high tempo over the last 2-3 years, and this inevitably requires appropriate public governance responses. Thus, again this is about the mix that needs to be right and is an important element to frame within the multi-level governance setting of the EU, as examined in Section 7.4.3.

However, there are different views if the Russian invasion of Ukraine means a fundamental geopolitical shift in EU external action that catapults the bloc toward greater geopolitical assertiveness and unity. The war seems to have unlocked more progress on EU foreign and security policy in a few months than was achieved in previous decades, with High Representative Joseph Borrell declaring "the awakening of geopolitical Europe". However, while unity has tightened between member states on some issues it has splintered on others and, despite the step-change in the EU's external action, there is limited evidence so far it will project a stronger or different form of power internationally—that is, as an emergent geopolitical actor—than it did before the war. For now, there seems to be no dramatic birth of any radically new European geo-strategy. In fact, the war invites a less comfortable debate about the EU's global role and identity (Carnegie Europe, 2022).

In contrast, it is clear that **the EU's neighbourhood is increasingly hostile and unstable**, with great power competition between China and the US drawing in Europe, and climate change posing ever-growing economic, social and geopolitical challenges. The US's precipitate withdrawal from Afghanistan in August 2021 showed that even under Joe Biden the US may act in its own interests first, with minimal consultation with its allies. In such an environment, the EU needs to reconsider how it acts on the global stage in order to protect its prosperity and security. It needs to develop the governance and other tools to face these challenges effectively, boosting its diplomacy, its defence capabilities, and its instruments for tackling internal security issues (Centre for European Reform, 2022). Indeed, these other tools will certainly need to include strengthening the EU's digital posture in all its aspects, not only in cyber security but also in consumer industries and public services.

4.2.2 Re-ordering settlements and communities

The EU's Committee of the Regions (2020) examined the territorial dimensions of Covid-19 across the EU and showed that, although government responses were largely national, they resulted in very different regional impacts. The socio-economic asymmetry of consequences across Europe, countries, regions and cities is largely shaped by diverse regional characteristics that call for higher levels of place-sensitive policy responses, taking into account a region's economic structure, structural challenges, and social profile. Although much of the analysis is focused on specific regions rather than regional types, the findings show both that because Covid-19 responses vary so much, the usual urban-rural differentiation does not easily apply, but that also metropolitan areas have generally been most strongly hit, though also tend to experience quicker recovery. Sharifi and Khavarian-Garmsir (2020) report that cities that do not have a diverse economic structure are more vulnerable to Covid-19. For example, in Poland, cities going through trans-industrialism, with hard coal mining. large care centres and shrinking cities, are the most vulnerable ones. Whilst the evidence is mainly on the negative impacts, more positive developments are also seen, for example Covid-19-induced transportation restrictions and border closures disrupted food supply chains in cities but have in turn provided additional momentum to urban farming movements. It is expected that more attention will be paid to local supply chains in the post-Covid-19 era. There are also successful cases of social innovation and collaboration, such as in Naples where efforts have been made, through volunteering programs, to get people involved in local practices that contribute to meeting local food demands and also strengthen social ties during the pandemic (Cattivelli and Rusciano, 2020).

To understand processes and relations within different regional types, it is useful to consider the **three stages of the urbanisation process** and how these can repeat themselves (Aleksandrzak, 2019; Mitchell and Bryant 2020):

- 1. Initial urbanisation, or agglomeration, accompanies the shift from an agrarian to an industrial factorybased society and sees growth concentrated in urban cores.
- 2. This is later followed by a suburbanisation stage during which growth occurs beyond the urban core, at the expense of the core's population as new forms of efficient transport allow the better-off to move out of the centre to new suburbs.
- 3. The final counter-urbanisation (or de-glomeration) stage sees the growth of smaller cities and towns in nearby areas beyond the built-up suburban ring and is accompanied by population decline in the core and its immediate suburbs.

The cycle can re-start with a re-urbanisation stage that sees new growth back in the original urban core, driven by the inward movement of both counter-urbanite and suburbanite populations. Many metropolitan regions, particularly in advanced economies like Europe, experienced a counter-urbanisation period in the past, for example in the early 1970s. Since then, parts of this cycle have repeated themselves especially in the last twenty years but through somewhat different processes, this time driven by globalisation and enabled by digital technologies leading to the counter-urbanisation that is currently taking place. These distinct metropolitan cycles, often reflecting at the regional scale an inverse relationship between population growth and city size, are also charted by Cividino (2020) with metropolitan growth being highly positive before 2000 but declining progressively in the subsequent decades. The 1990s were a transitional period away from a spatially homogeneous demographic regime based on high rates of population growth strictly dependent on city size, to the regime we largely see today grounded on low rates of population growth varying over space. This seems synonymous with Mitchell and Bryant's (2020) counter-urbanisation phase and the growth of smaller cities.

Capital cities typically exhibit pockets of poverty alongside very wealthy households, while other very large cities are often characterised by older economies and are more likely to have the lowest urban incomes as former industrial areas that have been left behind economically with relatively high levels of unemployment, poverty and social exclusion. The latter are also cross-related to class, occupation, gender and ethnicity which are themselves subject to such regional variation (Eurostat, 2021 and Millard et al, 2022). On the other hand, smaller cities typically changed less during Covid-19 given their relative younger demographic profiles based on newer economies compared with the other regional types. **This current counter-urbanisation trend is seen in this growth of smaller cities beyond the traditional suburbs accompanied by population decline in the core and its suburbs**. This is being recognised in many countries, for example, the Danish Knowledge Centre for Housing Economics (2021) is charting the movement of population out of the five largest Danish Cities, including Copenhagen, to the smaller provincial cities in their hinterlands which are today the fastest growing municipalities in a development that is expected to continue to at least 2040. Similarly in France where Loumeau and Russo (2022) show that the 2017 expansion of the French high-speed rail network from

Paris to Rennes and Bordeaux induced Parisian skilled workers to move to these cities, a trend considerably boosted by the dramatic growth in tele- and hybrid-work during the pandemic (see also Section 6.3).

This dynamic is being driven by a better quality of life in the smaller cities balancing urban and rural advantages, high service levels, as well as continued good connectivity to the larger cities when desired. These movements to smaller cities are also being fuelled by some population movement from rural areas thereby further weakening rural infrastructures and increasing the digital divide, except in some relatively attractive and accessible rural areas where 'second homes' are being bought up by well-off urbanities. However, this tends to have the additional downside of squeezing out lower paid locals from the rural housing market and only bringing business into the locality at irregular intervals.

Many **rural areas are particularly disadvantaged** being characterised by lack of digital and other infrastructures, such as housing and other facilities, poor digital skills and declining economic sectors. Their population profiles tend to be ageing with many retirees moving in whilst younger people and families move to towns and smaller cities where facilities and prospects are better. Although there are many exceptions and success stories based, for example, on tourism, bio-services and nature, culture and enterprising agriculture practices, there is a need for more focus on 'rural proofing', i.e. ensuring and improving the prosperity and connectivity of rural areas. Compared with urban areas, where there are often unitary authorities, the governance of rural areas typically suffers from greater complexity through shared and scattered responsibilities amongst a large number of actors across many more governance levels (local to European) and administrative bodies (economy, agriculture, etc.), so that power-relationships are out of balance. Rural authorities also tend to have a much lower per capita tax base and larger distances to contend with. The COVID-19 pandemic exposed pre-existing weaknesses in rural connectivity and widened digital gaps, so a policy roadmap to tackle this and other gaps is needed. Digital technology is important for service delivery and policymaking, especially to address locality issues (see also JRC (2022a)).

These complex movements have considerable housing, infrastructure and socio-economic implications, for example pushing up housing costs in the smaller cities, around the central districts of national and regional capitals and in rural areas. Thisse et al (2022) report that in London the premium paid for having a home closer to the CBD has fallen by 17.1% over the past two years. In 2019, moving 10% closer to the CBD meant spending 4.6% more on housing; in 2021 this premium fell to 2.91%. In the past there has also been significant 'gentrification' where younger well-educated workers' move to the cheaper parts of the cities, create a café culture, encourage in turn the influx of wealthier house seekers, thereby changing the structure of the local housing market. In part, this increases the quality of the neighbourhoods but also changes the dynamics for those already living there, often forcing out lower paid workers to worse accommodation further away from their workplaces often living alongside immigrant groups and possibly forming ghettos and place-based racism (see also Thisse et al (2022)).

According to KPMG (2021). Covid-19 has especially accelerated the move from some larger cities towards the growth of smaller cities, though not an overall decrease in the urban population, through the adoption of **online** shopping, working from home and online gatherings rather than meeting in person in cities and towns in England. KPMG predict that many people are unlikely to return to the old ways of doing things. With fewer people coming into very large cities to work and shop, that leaves a big space in areas that were once characterised by bustling shops and offices. Those places that are most at risk are those that have little else to attract locals and visitors from further afield. In these cities there has been a loss of commuter flow from over a tenth to under a third of commuter footfall seen pre-Covid. Apart from the largest, mainly capital, cities like London, the authors contend that it is unlikely there will be a return to old commuting habits in most very large cities, with a significant proportion of those able to work from home doing so for at least part of the week or shifting to working closer to home in smaller cities. This is likely to lead to significant reductions in office space in large cities and at least a partial collapse in their central retail areas. For example, Thisse et al (2022) report that London is experiencing an exodus of high-skilled workers seeking less expensive housing and better lifestyles in residential-friendly towns and cities and that this coincides with the closure of up to 14% of central London restaurants since 2020 with similar reports from other major cities including Manhattan, Toronto, and Paris (see also Section 6.3 below, JRC (2022b) and Alberti et al (2019)).

Several other questions arise here, including in a local development context, that the privatisation of ownership often leads to the privatisation of 'profits', primarily reflected in land value appreciation, with corresponding public ownership of 'losses' (Noring, 2018; McGreal et al, 2000). This can prevent the public sector from accessing and reinvesting the revenue stemming from public asset value appreciation that local government helps to create through measures such as land use, zoning, and localised infrastructure Whereas this is an issue emerging from the ownership of physical assets, the challenges also affect the recent discussions on the

providers of digital technologies for innovating public services and the public sector, i.e., the recent debates on GovTech (Mergel et al, 2022; Kuziemski et al, 2022). Another important issue is the changing relationship between physical and digital space, as well as between physical and digital identities. There is certainly a need for a new understanding of both the public governance of the land and of cyber and augmented space and how this may be changing as society possibly moves towards digital technology generation 5.0 and the Fifth Industrial revolution, even if this happens this will not apply to everyone, whether this be by choice or inability.

4.2.3 Turbo-charged localities and communities

In the years since the economic and financial crisis of 2007-08, most European countries have experienced a **resurgence in 'localism', what some have termed a 'new localism'**. According to Katz and Nowak (2018): "power is shifting in the world: downward from national governments and states to cities and metropolitan communities; horizontally from the public sector to networks of public, private and civic actors; and globally along circuits of capital, trade, and innovation." As noted by Millard (2017c), **cities, in particular, are at the 'sweet spot'** being, in general, sufficiently large to possess significant political power, financial and other resources, whilst at the same being sufficiently small and close to their populations to understand their needs, collaborate meaningfully with them and take and implement appropriate decisions on the ground. Cities constitute the new locus of power in the 'new localism' "that is needed to solve the critical challenges of modem societies: economic competitiveness, social inclusion and opportunity; a renewed public life; the challenge of diversity; and the imperative of environmental sustainability". New localism is not a replacement for the vital roles of national governments, but instead is the ideal complement to an effective overall governance (Katz and Nowak, 2018).

There has been a clear return to an understanding that social, economic and cultural structures and competences at local level have huge value and are crucial to building wider concepts of socio-economic and environmental development in general. This understanding is, however, not a return to earlier notions of localism based on hierarchical structures, but instead closely mirrors the objectives and desired impacts of open and social innovation in meeting real social needs in new ways. This approach is increasingly focused on the daily social needs of people in their communities for work, education, health and prosperity in local contexts and in ways they themselves have some control over and so they can also contribute to strengthening how their localities work. This illustrates the dialectic between, on the one hand, a more macro, top-down and sometimes rigid structural one-size-fits-all approach with on the other hand, a more locally-embedded, nuanced as well as socially- and culturally-aware agency approach which, however, adds complexity to governance, multiplies the numbers of actors and relationships and potentially adds to the 'silo' problem. Whilst the locality has become more important in recent years, it has not yet seriously threatened the continued dominance of statism. Into this mix of systemic societal change, new condomerations of this structure-agency dialectic approach are emerging. Many are attempting to build new economic models based on shared value and social value often derived from and embedded in localities (for example, as described by Porter and Kramer, 2011), alongside an increased concern for inequality, poverty and social distress.

However, this is not only a simple model of top versus bottom as there are many layers and levels dependent upon the particular national, historical and cultural context. There can be hierarchies within hierarchies. For example, taking a 'new localism' approach also means recognising that the largely top-down structures and policies of many smart city initiatives, in which the city authorities and other organisations deploy sensors, networks, data and data analytics to improve the efficiency of urban systems, like transport, utilities, and services, important as these are, is only half the story. From this perspective on its own, there is the danger of a one-size fits all, top-down view of local development. The diverse needs of the inhabitants as individuals, households, neighbourhoods, communities, organisations and businesses, that bring a locality to life, are just as important. Thus, any adequate governance model for the locality/city must also focus on the engagement and knowledge of its citizens and encourage the processes, and especially social and cultural processes, that make these places important: those that sustain very different, sometimes conflicting, activities. Cities, in particular, are in their nature engines of diversity, so focusing solely on efficiently streamlining utilities, transport, construction and unseen city administration processes can be massively counter-productive. Instead, localities will be successful, innovative and smart because their citizens have found new ways to craft, interlink and make sense of their own and each other's assets, data and other resources (Millard, 2017c).

The **new localism attempts to retain as much as possible of the value generated locally within the locality, rather than see it seep away**, possibly out of the country and to tax havens from whence there is little chance of return investment. The definition here of 'locality' does not necessary mean within a particular administrative jurisdiction, nor does it negate the importance of seeing value retention in the context of collaboration with other localities, however defined, or with other authorities in the multi-level governance system. Indeed, the issue of value creation, use, retention and/or sharing, as well as the politics and policies that surround this, is a critical concern of a re-evaluation of multi-level governance. Awareness and action on this premise has become turbo-charged since the pandemic, partially driven by de-globalisation and more on-shoring of economic activity from which localities seek to benefit, as well as by the highly spatially differentiated impacts turbulence and shock are having. This is seen, for example, in the very significant increase since 2020 towards both food and energy localisation (see also Sections 6.1 and 6.2) where both consumer awareness and behaviour have changed towards accessing and supporting local sources and supply chains and that point to the need to rebalance national-local powers and to develop cross-cutting plans (Millard et al, 2022).

In the specific context of food and energy, the role of community infrastructures is critical, but so too are middle infrastructures to reconnect production with consumption and larger markets, thereby building resilience through intermediate markets. However, in other sectors such as tourism, work and mobility the local community role is likely to be less important given their general need for much wider markets, and certainly for collaboration between authorities in the multi-level governance system, as mentioned above. The overall thrust of this focus is about the importance of linking local and community policies across all sectors and providing new types of place leadership. The growing importance of the sustainability of local and city-region systems is inevitably linked to the topic of shortening supply chains and re-shoring, especially in the context of some de-globalisation which was already evident prior to the COVID-19 pandemic but also strongly accelerated by it and the 2022 Russian invasion of Ukraine. There is starting to be a movement to more directly reconnect producers with consumers, both physically in the locality from local trusted suppliers who are able to guarantee genuine and safe products, as well as virtually using digital technology. This involves building more transparent supply chains with the fair distribution of power among actors. These and similar developments help to strengthen the sustainability agenda and enhance resilience, especially during shocks and emergencies like the pandemic.

Policy orientations for governments in this respect have been identified by the OECD (Bulakovskiy, 2021). In the new post pandemic reality, policy-makers need to promote the creation of enabling environments and frameworks that foster the emergence and the development of local as well as national development through both demand and supply-side measures. On the demand side, the main goal would be creating local markets which do not leak wealth through a series of measures, including awareness campaigns, public procurement to prompt the integration of small enterprises, impact measurement tools to assess the relevance of such activities, and fiscal policies, namely tax incentives and subsidies. On the supply-side, these measures need to expand the number of actors and enhance the quality of their activities. These would include direct or indirect financing of local initiatives, the provision of infrastructures allowing local socio-economic ecosystems to flourish, such as incubators, the promotion of a process of skills development in accordance with the know-how required in a given sector, as well as the provision of the suitable tools and means needed to foster effective cooperation between all relevant actors. It is often at the local level that it's easier to break down governance silos and link across actors and sectors in order to realise a joined-up whole-of-government experience for users, given their smaller and shorter interaction and transaction chains and much more fine-grained knowledge, both implicit and explicit, about local conditions and contexts.

Traditionally local governance, especially in relation to local financing and economic development, have been in the competence and power of both the public sector and the private sector, each playing various roles, sometimes cooperating, sometimes competing. In recent years, however, and especially since the start of the pandemic, there has been **increasing involvement of civil society** in this mix. This is often in cooperation with both public and private actors, but also sometimes taking place without reference to them and even under their perception radar. Civil society is composed of a very large number of diverse organisations and institutions, ranging from very informal to formal, including individual citizens, families, neighbourhood groups, communities, NGOs, social entrepreneurs and the alike and where the concept of community is important, which means their roles and impact are highly diverse and dependent on the local context. Their activities encompass both direct monetary as well as in-kind initiatives, so measurement and impact assessment can be challenging. It is clear, that local public value is co-created both through the collective efforts of public and private investment as well as through the myriad socio-economic and cultural activities of civil society.

There is strong evidence that the **involvement of the civil sector can increase inclusion, trust, the quality of decisions and overall effectiveness, through the consideration of a wider set of ideas and access to additional assets** (Millard, 2015a). In some instances, this can include extra revenue for local development, as well as retaining finance and revenue within the locality through re-circulation rather than seeing this seep away. To date, however, these effects are relatively small compared to the impact the public and private sectors make, although their effect can be significant on the ground and in highly specific contexts, and it is now clearly increasing quite significantly. It might be argued that one reason why civil activities flourish and new localism has spread so widely, in the USA and the UK for example, is that there is a gap between a relatively small public sector and a relatively large private sector that the civil sector is able to address and fill, particularly by providing for the most vulnerable in society. In contrast, in most European countries, such as in Germany and Denmark, the role of civil society in addressing this gap and fulfilling such roles is relatively smaller given that the public sector tends to be larger, is often more responsive and, to some extent, more capable of providing comprehensive public services in these and similar countries (Millard et al, 2019b).

A good example of community development is the **Community Wealth Building (CWB)** movement as a system-changing approach to community governance that works to produce broadly shared economic prosperity, racial equity, and ecological sustainability through the reconfiguration of institutions and local economies on the basis of greater democratic ownership, participation, and control. It aims to link different governance levels to support local community development by attempting to retain as much as possible of the value generated locally within the locality. It creates a new model of economic development for cities and communities that offers real, on-the-ground solutions to localities and regions battered by successive waves of extraction, dis-investment, dis-placement, dis-empowerment and now by the many post-2019 shocks. CWB is based on a new configuration of economic institutions and approaches capable of producing more sustainable, lasting, and equitable economic outcomes. Rooted in place-based economics, with democratic participation and ownership, and mobilising the largely untapped power of civil society, of the local public sector and other anchor institutions (CLES, 2022).

This is done, for example, by designating **local anchor institutions** (public, private and civil) that invest and procure locally where possible, and by increasing local, employee and community ownership and control An anchor institution is a place-based organisation that is invested in its local area and cannot relocate to another part of the country. Examples include local councils, universities, colleges, local housing associations, local emergency services as well as local civil and community organisations. By their very nature, these organisations also spend substantial amounts of money that is retained within the local area. While most of their employees and volunteers are likely to live within the local area, and spend their wages there, they also have significant procurement and investment spend which can be spent locally. They have a collective interest in seeing their local area and community improve and are always looking for more opportunities to advance collaboration. For example, many attempts to promote local currencies, such as the Bristol Pound (£B) in the UK, aim to retain locally generated financial value within the locality, but with relatively unconvincing results across the broad range of economic sectors. However, in a few sectors such as food production, processing and retailing, local circulation loops have shown to be highly successful, perhaps because food lends itself particularly well to further localisation as many of its activities are viable without the need to interact with regional, national or international businesses. Thus, it is a question of smart targeting of local currencies through the adequate analysis of the business models for specific goods and services in each specific local context (Geofutures 2020).

The CWB approach stands in opposition to the prevailing model of community development that puts the accumulation of private wealth and profit above the basic needs of people—entrenching and exacerbating racial, economic, and geographic disparities. There is no one-size-fits-all model of CWB building. Rather, it is a bottom-up approach that centres democratic ownership of the economy and community self-determination. This means that each local experiment with CWB might be different, based on the local context, ecosystem, resources, and politics. What all CWB strategies have in common, however, is that they aim at improving the ability of communities and individuals to increase broad-based asset ownership, anchor jobs and resources locally, create broadly shared prosperity, and ensure local community economic stability and democratic control. Importantly, true CWB must be reparative and inclusive by design so that it delivers for those who have historically been the most excluded, marginalised and exploited. CWB is about moving in the direction of a different political-economic system, linking new bottom-up forms of development with economic and political interventions at a variety of scales to create an economy in which all can flourish.

Emerging trends related to globalisation since the 1990s—such as the relative decline of manufacturing in most developed economies, the rise of the private service sector and mounting government fiscal crises— and now more recently related to the new localism—including some on-shoring and reduced national government services resulting from these fiscal crises—both suggest the growing importance of anchor institutions to local economies. Indeed, in many places, these have surpassed traditional manufacturing corporations to become their region's leading employers. If the economic power of the anchor institutions were more effectively harnessed, they could contribute greatly to community wealth building. The earliest examples of CWB are both in the UK and USA, where the retreat of the state as well as increased local poverty, unemployment and social exclusion especially since 2020, have tended to be greater than in most EU countries that typically have larger public sectors. These include in the Ayrshire and Arran regions of Scotland where the focus is on ensuring that
wealth is locally owned and benefits the local community (Ayrshire and Arran NHS, 2019), and in Preston in England where the purpose is to provide value for the city's communities. (Preston City Council, 2022).

Although also important at national level (Hofstede Insights, 2022), locality and community are very relevant for **identity**. **behaviour**. **values and culture**. These attributes are often cross-related to class, occupation. gender, ethnicity, age and family composition, that are themselves subject to strong spatial concentration and variation (see Section 4.2.2). Local pride and identity can be deeply rooted, even when people move away from where they grew up, but identities can also be mixed without being contradictory. For example, football fans living in Milan have intense rivalry with those in Turin, and northern Italians do not always readily identify with the culture and mores of the country's south, but everyone in Italy supports the national football team. History and historical memories also play a part in defining local and community identities. Older East Germans who grew up in the DDR tend to have a decidedly different view of the war in Ukraine than the younger generation in this region or compared to those living in the former West Germany. Also, the short social and physical distances inherent in locality and local community make it easier to generate and judge positive reputation and trust, given that contextual links are more vivid, broader and multi-dimensional. This makes these difficult to replicate digitally and it is doubtful that a new cyber-localism that disconnects itself from physical proximity is possible, although there is widespread use of local social media to complement, rather than supersede, physical interaction. 'Local' and 'community' are forms of identity politics, e.g. where I come from, where I work, where I socialise, etc., for example as manifest in areas like cultural heritage. It is clear that place, identity and policy are linked, so the often profound geographic changes after the pandemic need to be addressed by new public governance. It is also particularly the case that social identity relates strongly to locality and community (Scharfbillig et al. 2021). Some of these issues are further addressed in the accompanying community perspective report (Errandonea, 2023)).

The role of digital technology in this context might also be both location- and event-based services in which the user gives permission for their location to be a factor in the experience offered when in a particular place or participating in a particular activity. Such permission by the user might also be given to be made aware of relevant services in this type of situation, for example, when entering a certain zone in a city if augmented reality is being used, thereby potentially linking human, digital and bridged realities (Millard, 2010).

4.3 Strategic challenge 3: Hybrid human-centred relationships

4.3.1 Balancing 'episodic' with 'preventative' public services

Many public welfare services were originally designed and intended to support individuals get through an occasional problem, such as an illness or becoming unemployed. Such 'acute' services remain absolutely essential but do not necessarily reflect the lives of everyone, especially the vulnerable the proportions of whom have been growing in Europe since at least the 2008 financial crisis but have again now been dramatically increased since 2020. Such episodic welfare problem solving services are also expensive especially in the time of squeezed public budgets. They are transactional and reactive which fits ill with many people's complex, and in the case of the vulnerable often fragile, lives. These services were originally designed in the 1940s and 1950s when European societies were very different from what they are today. There is thus a need for radical change in public welfare services to move towards being much more relational and proactive. The relationship between these services and the people who use them needs to be transformed to allow people to take greater control of their own health and wellbeing in the context of more continuous support in a community context. Existing ways of delivering services can sometimes disempower the people they are there to help, leaving people feeling unable to make positive changes in their lives and their communities. In the case of health and social care services, this means striking a new relationship that puts more power in the hands of patients and service users and emphasises 'working with' rather than 'doing to' (BBC Analysis, 2022).

One relatively new but still quite rare innovation is to move the balance away from services only designed for occasional support when people have a problem, such as a specific life event like a birth, marriage, loss of job, etc. There is also a need for services that recognise peoples' lives are continuous and multi-dimensional with interlinked needs that **require preventative personalised hybrid support** and that are, especially for the vulnerable, 'wrapped around the whole person'. This does not involve unwanted intrusion but is **based on sufficient time and ethically-guided elective dialogue between the service user and provider in a confidential setting** that aims to understand the full range of the user's needs given that these are typically strongly intertwined and embedded in personal and community relationships. One example is in Wigan in the north-west of England where austerity cuts since 2010 have been huge obliging the local authority to re-think such services from scratch. Over a period of six years from 2013-19, public services in Wigan have been through

a major process of transformation, based on the idea of building a different relationship with local people using an 'asset-based' working approach in which **public services seek to build on the strengths and assets of individuals, their relationships and their communities to improve outcomes** (Social Care Institute for Excellence, 2022). This also involves providing hybrid support to residents at home through digital technology as much as possible, or in convenient places in their community, keeping hospitals for when they are most needed for more acute care. Digital technology is also essential in coordinating across all the involved actors and providing easy access to documentation (Wigan Council, 2019).

In Wigan, the main focus has been on health and adult social care services, but the approach also applies to community support in general, businesses, children and young people and wellness (Local Government Association, 2022). Wigan has shown it is possible to achieve substantial financial savings at the same time as improving the community's overall health and wellbeing. Such savings are made by reducing the need for and frequency of expensive one-off acute interventions when things go wrong, as well as making efficiencies while simultaneously improving effectiveness and protecting or improving outcomes. This only happens when services are genuinely transformed and upfront investment is available to help bring about new ways of working. This is not a panacea, but it does illustrate the kind of work that is needed to shift to a new model of public service delivery in which service users, professional local authority staff, volunteers and communities are all involved as active partners. Apart from asset-based working, other core elements of Wigan's approach include: **the need for profound cultural change amongst all actors but especially the professionals**, recognising and nurturing the strengths of individuals, families and communities and to build independence and self-reliance; permission to innovate; investing in communities; and place-based working (King's Fund (2019).

Physical drop-in facilities are also available where vulnerable people have immediate access to an adviser with whom they can **build a long-term relationship** with time to talk and who has access to, and knowledge of, all services and solutions available whether within the public regime or not. In some ways this is **a 'one-stop-person' approach** which attempts to 'move on' from the already quite widespread physical 'one-stop-shops' where people can get help accessing digital services, such as in Portugal (Millard, 2017b), especially for those on the wrong side of the digital divide (see also Section 4.3.3 below). The difference in Wigan is that the support is personal (or small team based) and relational, as well as more continuous and community-based for vulnerable users requiring longer-term support.

Another important innovation of moving to hybrid and preventative human-centric public services is Buurtzorg, the biggest home care organisation in the Netherlands. It shows both inclusive and co-creative governance, developed an approach based on **self-leading teams of carers and other professionals with the slogan of putting "humanity over bureaucracy"**¹³. Buurtzorg is a pioneering healthcare organisation established in 2006 with a nurse-led model of holistic care that has revolutionised community care in the Netherlands and is now represented in about 25 other countries. It has increased from 10 to over 10,000 staff today, has **a miniscule administration** of about 30 staff overall whilst each of the now 850 self-leading teams has 10-12 **staff with ad hoc roles who decide everything for their team** (including what coffee to drink). This is an example of so-called '**mass customisation' (i.e. mass personalisation)** as all patients are individually unique care that is in constant flux to maximise their welfare. This also works out cheaper and better than any alternative with financial savings of 40% compared to other comparable models.

Buurtzorg has a **simple, flat organisational structure providing and accessing all necessary services with intimate involvement of the patients' family and community**. Its evolutionary goals are diversity and the innovative search for promising ideas – "if something works it will find a way to flower". Buurtzorg's user satisfaction rates are the highest of any healthcare organisation and staff commitment and contentedness are reflected in its title of Best Employer (4 out of the last 5 years). Buurtzorg scaled very quickly across the Netherlands and has also moved into related areas of care such as mental health, children and families, as well as supporting other Dutch and international care organisations to take on the Buurtzorg model. Digital technology is an essential enabling tool, including for collective intelligence, communication, fast decisions, eg. facilitating 'real time' information that is directly connected to the care process and reduces administrative overhead.

¹³ <u>https://www.buurtzorg.com</u>

4.3.2 'Community-based solutions'

'Community-based' solutions are, in essence, a subset of preventative public services. For example, the Wigan case also encompasses 'the Hamlet' which offers day-provision for those children and young adults with 'additional needs' such as learning difficulties who, at the time of admission, are independently competent, and aim to progress to full-time unsupported employment or voluntary work. Wigan Council has set up a "community-investment fund" as a type of local crowdfunding, with the aim to support the re-booting the social connections and relationships that have often been lost. The aim is to use this resource, together with other local assets including from the local authority, to **focus less on asking public service users "what's** the matter?" and instead setting up a dialogue focused on "what matters to you?". As one focal point of this, 'the Hamlet' is a community and provides community-based solutions for those with these additional needs in order to develop life skills, get a job or contribute to society in other ways. Apart from, but supplementing, these social services there is a shop, café, laundry, florist and a printing works, as well as a range of social activities such as singing groups for example for people with breathing issues as this can increase lung capacity by 20%. Impacts have been huge by helping to move these young people away from anger and frustration in attempting to deal with opague and narrowly focused and often unresponsive public services, where they have little say in what happens, towards a sense of hope, belonging and empowerment. Many more 'hamlets' are being set up around the country and elsewhere (BBC Analysis, 2022).

It is only by addressing the complexity of real interlinked problems, on the citizen's own terms and at their own pace in the context of their family and community relationships, that solutions are found that can also mitigate far greater problems in future which are likely to cost much more money. An often important issue is the stigma felt by many vulnerable people who, for example, are in poverty and are ashamed of admitting it, thus privacy and ongoing confidentiality are sacrosanct. Community-based solutions in this way attempt to **provide holistic solutions that 'wrap around' the whole individual by bringing many previously separate services together** using a similar approach as deployed in Wigan. This is a massive change compared to traditional public services.

However, the barriers to moving to preventative and community-based services are huge, hence the lack of widespread adoption despite their documented success and cost savings. The main issue is **the need** for massive administrative, legal and, above all, cultural change. It involves asking departments to stop being autonomous, to work together and to pool their budgets, leadership and decision-making. This is a huge ask. There is the need to dismantle traditional professions and the professional boundaries that pull in one direction and to redesign new emergent professions that pull in another direction. Making such changes at scale is hard and doubly so in the public sector because of its inherently political nature. The traditional professionals are deeply cemented through the levels of government up to and including ministries at the state level. The local professionals are dependent on these connections and lines of authority, structure, power, accountability and political will.

Even when change is attempted, it takes time which frustrates the political need for quick results and tick-box targets. All this amounts to a compelling professional and political logic that is maintaining the status quo, even against rational and evidence-based argument. It is this logic that makes public services and their political paradigms so immune to change. This is nothing new given that during the founding of the welfare state in Europe in the late 1940s and 1950s there was just as much resistance and inertia as seen today. However, this was eventually **overcome by clear leadership with a compelling vision** backed by massive early post-war popular support through inspiring stories based on **the only question that really matters: what is needed for a good and flourishing life?** The solution today probably rests on these same ingredients, but this time in the context of much more devolution and decentralisation given that preventative and community-based services are inherently local and require local co-creation and relationship building that is also necessary to generate the trust needed to lubricate change (BBC Analysis, 2022). **A new social contract is indeed needed** (Vesnic Alujevic et al, 2019) alongside a new political contract.

Digital technology and data are used to support these otherwise strongly human to human relational services on the basis of current controls and standards. As in most other governance models, however, it is also realised there are dangers of data misuse especially where volunteers and private companies are involved. The issue is what role does current and future technology have in supporting, improving and indeed extending the benefits of such services without degradation. There is of course also a diversity of approaches and views across the EU. Both between countries and within their regions, preventive services might be perceived very differently, especially because of the cultures surrounding privacy and data handling by both public and non-public entities. Some of these issues are further addressed in the accompanying community perspective report.

4.3.3 Justice, fairness and inclusion by design

Human-centred public services, and public governance more generally, need to incorporate 'design justice' **principles**, where decision-making processes include designing not only for but also with citizens and communities. The focus is particularly, though not only, on marginalised and oppressed communities, which have relatively little power, by centring the voices of those most impacted and using transparent processes through co-creation. According to Costanza-Chock (2020), **design justice takes into account the relationship between design and power by explicitly challenging, rather than reproducing, structural inequalities**. It has emerged from a growing community of designers in various fields who work closely with social movements and community-based organisations around the world. The approach attempts to counter the danger of universalist design principles and practices unfairly erasing certain groups of people—specifically, those who are intersectionally disadvantaged or multiply burdened. Design Justice goes beyond recent calls for design for good, user-centred design, and employment diversity in the technology and design professions; it connects design to larger struggles for collective liberation and ecological survival.

To implement design justice principles, digital technologies need to be developed and used to support the design of policies and governance management in ways that are fair, transparent, embed accountability, and centre the voices of citizens and civil society. This requires **questioning how technologies themselves are currently being used, designed and controlled**. For example, 'big tech' companies dominate the type of technologies we use, which are designed in ways that do not afford justice because they are designed to generate profits through reactive engagements and polarising loops and through the use of people's data, rather than designed to facilitate deliberation, understanding and decision-making, and in ways that protect people's rights. In short, digital technologies can be used in ways that enable democratic and justice-based processes but, for this to happen, **the technologies themselves also need to be democratised**.

A good example from Taiwan of using democratised technology is the POLIS software used for participatory governance through 'radical transparency' by throwing opague processes open to the light so that everyone who is affected by a decision should have a say in it. This is a way, not to measure division, but to construct consensus using the internet. However, the internet is normally part of the problem as the kinds of online spaces where political debate happens are engineered for an entirely different purpose: to capture attention and thereby to attract advertising revenue. Most social media platforms serve up information that is shocking, horrifying or crazy enough to keep people glued to their screens. And that often means amplifying the raucus politics of division and outrage rather than the subtle complexities of compromise. POLIS is designed differently. Citizens are invited into an online space for debate that the POLIS software uses to draws map showing all the different knots of agreement and dissent as they emerge. As people express their views, rather than serving up the comments that were the most divisive, POLIS gives most visibility to those finding consensus. In this way, "people spend far more time discovering their commonalities rather than going down a rabbit hole on a particular issue." "Invariably, within three weeks or four, we always find a shape where most people agree on most of the statements", says Audrey Tang who became the country's digital minister (Miller, 2020). POLIS shows that re-engineering the online space had exposed a deeper human truth. In politics, humans spend most of their time concentrating on what they disagree upon. However, if you gamify consensus, you expose points of unity that were previously hidden. By clearing away the noise and divisiveness, outcomes are created that the government can actually act on, POLIS has formed the core of around a dozen pieces of laws and regulations now implemented in Taiwan, on everything from revenge porn to fintech regulation. Many other countries are now using POLIS software (Miller, 2019).

Justice, fairness and inclusion also have a strong regional dimension. The **Just Transition Mechanism and Fund** is one of the European Union's key tools to support regions' transition towards climate neutrality by 2050, under the 2021-27 programming round within the framework of the EU Cohesion policy. It is part of the European Green Deal aiming to achieve EU climate neutrality in an effective and fair manner. Focus is on the regions and sectors that are most affected by the transition due to their dependence on fossil fuels, including coal, peat and oil shale, and on greenhouse-gas-intensive industrial processes. Its main objectives are to alleviate the impact of the transition by financing the diversification and modernisation of the local economy and by mitigating the negative repercussions on employment. In order to achieve its objective, the Just Transition Fund supports investments in areas such as digital connectivity, clean energy technologies, the reduction of emissions, the regeneration of industrial sites, the reskilling of workers and technical assistance (European Parliament, 2022c).

The challenges of the lack of justice, fairness and inclusion by design in the use of digital technology have been highlighted during COVID-19. Although the pandemic has sped up digitalisation and related changes in work,

production and the provision of services which has mitigated many of the lockdown and slowdown problems for many people, it has also **supercharged the pre-existing digital divide and further reinforced preexisting socio-economic exclusion**. According to Watts (2020) any health-care development that doesn't rapidly become available to all individuals has the unintended but inevitable consequence of fuelling health inequality. The response to COVID-19 is no exception and has inevitably thrown a spotlight on this digital divide, most obviously in everyday domestic life. Many patients who do not have COVID-19 in the UK, fearful of acquiring the infection, have been reluctant to enter hospital buildings. Attendances have shown a marked fall. Many outpatient consultations can of course be done online—but only if you have access to a computer.

Work by Eurofound (2021), shows that **the uneven pace of the acceleration of digitalisation created by the COVID-19 crisis increases inequalities and the digital divide between territories, urban and rural areas, and citizens in the EU**. Telemedicine may not provide the same quality standards as traditional healthcare systems, and data collection and sharing of personal and non-personal sensitive data during the health emergency have raised concerns where they have taken place for purposes not directly or specifically related to the COVID-19 response. In the EU, there is a geographical **urban-rural digital divide in terms of the quality and affordability of broadband networks**, which affects people's opportunities to make use of telemedicine and many other online services. There is also a digital divide between those who are used to managing digital devices and those who are digitally illiterate or have a low level of digital skills. Inequalities in health, purchasing power and digital literacy mean that the patients who stand to benefit most – such as older people, people with disabilities and those who have retired – are often also those who are least able to access and make use of telemedicine.

4.4 Strategic challenge 4: People-planet systems

Arguably **the greatest challenge that public governance has** arises from the scientific facts that **climate** change, biodiversity loss and other stresses on nature are having profound deleterious impacts on the functions of societies around the world and on the lives of their inhabitants. These functions and lives are all too often, in turn, increasing environmental stresses in a downward mutually damaging spiral that can run out of control. This is leading to numerous spin-off challenges, like poor air quality, heat island effects, increased flood risks, increased frequency and severity of extreme events, as well as rises in crime, social exclusion, inequality and a degraded infrastructures that can be directly traced to environmental stress. All these have deleterious impacts on human health, quality of life, wellbeing and security, and hit the less privileged and vulnerable the hardest of all. To address these existential challenges, public governance has a critical, if not the pivotal, role to play. There is also significant and credible evidence that by **working with nature rather than against it**, a whole range of environmental as well as economic and social gains can be realised that benefit all societies as well as nature itself (Millard et al. 2019a). Societies need to adopt both climate change adaptation and mitigation initiatives, drawing upon the ecosystem services and natural capital that nature provides, as well as going beyond these to incorporate soft engineering, bio-mimicry and industrial design approaches designed to enhance environmental resilience and reduce risks to people. Public governance needs to transform towards the facilitation of sustainable, resilient and regenerative solutions, including for example circular economies and systems (Raworth, 2017 and European Commission, 2020a) and nature-based solutions (European Commission, 2021f). These prioritise strategies derived from natural ecosystem lessons, where 'nature' is recognised as an actor in its own right that is just as important as the familiar human actors.

Such a radical public governance reorientation will need to envelop ecosystem governance, such as truly 'green' business models, eco-based policy and the promotion of bio-based industries, circular and iterative systems where there is no such thing as 'waste', such as in hydroponics and rewilding. The use of digital technologies is critical, such as in support of 3D printing which use recycled materials and ape the way spiders secrete their web, street level air pollution measurement using hand-held sensors, sharing economy platforms, etc. Where beneficial, **it is important for public governance to facilitate the mimicking of nature's successful innovations also in organisational and human development**. For example, in de-centralised and self-leading teams as adopted in the Buurtzorg business model (see Section 4.3.1 above), in business practices where many of the most successful companies mimic natural systems (Bragdon, 2016), regenerative instead of wasteful strategies, frugal innovation and an ecosystems approach to successful social and business networks. Nature, defined here as both biological and non-biological systems, should thus no longer be treated as an afterthought or add-on, but should be central to every public governance policy, strategy and action, both by drawing on nature's innovation potential as well as by testing against the needs of nature.

4.4.1 Quintuple helix

Public governance needs to acknowledge and promote the understanding that it is **both human and natural assets and knowledge, especially working together, which are the only real sources of any type of innovation, including technological and business innovation**. Natural systems often show the way for successful societal innovations, such as ecosystem development, diversity and interdependence, re-cycling and re-using assets, circular societies and economies, and learning systems through co-creation and an understanding that any under-used asset is a wasted asset. Often completely new forms of social and ecological activity and organisation will be needed, most of which are experimental and only few are currently on the drawing board.

Four actor types are already involved in public governance, i.e. as the guadruple helix: the public, private, knowledge (e.g. education and research) and civil sectors. It is imperative to add nature as the fifth actor and thereby govern the quintuple helix. Nature, as defined here, encompasses both living and non-living systems, i.e. the biosphere plus the rocks, atmosphere, oceans, etc., all already heavily transformed by humans (Caravannis et al. 2012). Nature needs to be incorporated in its own right given it is essential for all society's functions, especially given the multiple environmental crises we currently face including climate change and biodiversity loss. Giving nature a seat at the public governance table ensures that environmental as well as economic and social sustainability is also a major goal and that these three pillars are pursued together with good governance, as in the UN's Sustainable Development Goals (United Nations, 2015b), also adopted by the EU (European Commission, 2021c). Given that the four quadruple human actors rely completely on nature for all aspects of their existence, it is arguably legitimate to express the notion that nature is the 'table' and not simply just one actor sitting at it. In some contexts, therefore, it might be that a 'bevond human' approach needs to be an aim, given that we quite literally make sense of our surroundings through our touch, sight, hearing, smell and taste, but that this is in many ways different from how other sentient beings see and interpret the world. Attempting to do so is likely to facilitate a better accommodation between humans and nature.

Nature, as defined here, has been innovating biologically for at least 3.8 billion years and non-biologically since the birth of the universe, far longer than human actors. **A useful strategy when we have a societal, business or technological challenge is first to ask whether nature has already solved the problem and, if so, how?** There are already a huge number of examples across all sectors (Benyus, 2016). In many industries, this is recognised and being exploited in the so-called Fourth Industrial Revolution, characterised by a fusion of technologies that is blurring the lines between the physical, digital, and biological spheres (WEF 2016). This also needs to happen in public governance in order to facilitate the societal-ecological transition as illustrated in Figure 3.

The needs, limitations, opportunities and innovations nature represents are a critical component of any type of human endeavour. According to Carayannis et al (2012): 'The quintuple helix model is interdisciplinary and trans-disciplinary at the same time: the complexity of the five-helix structure implies that a full analytical understanding of all helices requires the continuous involvement of the whole disciplinary spectrum, ranging from the natural sciences (because of the natural environment) to the social sciences and humanities (because of society, democracy and the economy). Thus, the goal of the helix-conception is accomplished through the resource of knowledge which produces additional value for society in order to lead in the field of sustainable development."

Indeed, this is also starting to happen in business, for example a beauty company has appointed a director to represent nature on its board, giving the natural world a legal say in its business strategy. *Faith in Nature*, which sells soap and haircare products, as well as household cleaners and shampoo for dogs, says it is the first company in the world to give nature a formal vote on corporate decisions that might affect it. Working with *Lawyers for Nature* and the US-based *Earth Law Center*, as well as a pro bono team of corporate experts at an international law firm, this decision by the Edinburgh-based company taps into a growing global movement to assign nature legal rights (Kaminski, 2022). According to GARN (2022), **"Rather than treating nature as property under the law, rights of nature acknowledges that nature in all its life forms has the right to exist, persist, maintain and regenerate its vital cycles.** And we – the people – have the legal authority and responsibility to enforce these rights on behalf of ecosystems. The ecosystem itself can be named as the injured party, with its own legal standing rights, in cases alleging rights violations." Public governance has the role of ensuring that nature and the environment are also protected and nurtured through legal and regulatory systems.



Figure 3. Process of innovation towards the societal-ecological transition

Source: Adapted from UNESCO (2016)

4.4.2 From sustainability to resilience to regeneration

Figure 2 has already suggested that, in the aftermath of the COVID-19 pandemic the concept of sustainability needs to be re-interpreted as giving resilience (defined as diversity plus interconnectivity) much greater weight than traditional notions of efficiency. This is also proposed by the WEF (2020) in order to 'build back better'. Economic efficiency has been the prevailing mantra and has increasingly become a short-term goal, whilst economic and societal resilience requires a medium-to longer-term perspective. Resilience actively anticipates and prepare for both likely as well as existential risks to humanity, implying increased finance and effort in the short-term, but demonstrable financial savings and much less effort and pain over the medium- to longer-term. This argument was first put forward in stark economic terms by Stern (2007). Most economies to date, whether developed, developing or emerging, have so far failed to achieve let alone recognise the need to change the balance between efficiency and resilience (see Section 4.1.3.).

The concept of sustainability remains fundamental in the context of sustainable development and the Sustainable Development Goals (SDGs) (United Nations, 2015b) that were designed for "meeting the needs of the present without compromising the ability of future generations to meet their own needs" (Brundtland, 1987). There is clearly a time element implied here so earlier concepts of 'sustainability' need re-interpreting, as above, rather than discarding. **Traditional notions of sustainability have tended to imply only keeping what we have through conservation and preservation, rather than what we really need.** This is at least partially due to what we, at the present time, are able to understand as the 'natural state' of the environment that many equate with when they were young or what their parents and grandparents remember through storytelling and reminiscence, rather than going much further back in time.

This is termed the '**shifting baseline syndrome**', i.e. the gradual change in the accepted norms for the condition of the natural environment due to lack of past information or lack of experience of past conditions (Monbiot, 2014). This may be partially an issue of **cognitive dissonance** especially when seen in terms of cultural awareness. Older persons may look back at the idyllic 'natural' environment of their childhood as their baseline. The problem is this environment was already much worse than their grandparents' and forebears' baseline given that each generation shifts its baseline of understanding away from what might otherwise be seen as what is best or what is in fact needed. In many ways, this mirrors a huge challenge as it is clear that much of what is done in the context of public governance is severely constrained by "**this is the way we do things around here**" attitudes which act as an often insurmountable barrier to doing things better, as explored in Section 4.3.2. Clearly such **challenges are just as much psychological and cultural as they are organisational and structural** and are seen very clearly in the dissonance many politicians and other actors exhibit in their attitudes and actions regarding the climate and environment crises. This also implies the need for the current older generation to ensure that the younger and coming generations do not fall into the same trap, but rather understand that **nature should be the first one to thrive in order that human beings can survive**.

There are, however, questions being asked as to whether it is "time to ditch sustainability" and move on to 'resilience' and 'regeneration' rather than re-interpreting sustainability to encompass these new insights. According to Acaroglu (2022) there are calls for sustainability to be left behind and for a new approach to be adopted. These arguments contend that sustainability is not enough, that it is only about "doing less harm" or getting simplistic efficiency gains. Others say it's problematic because for years, companies have been doing basic Corporate Social Responsibility (CSR) or triple bottom line reporting, and now there is a rise in superficial ESG (Environmental, Social, and Governance) frameworks. There are also criticisms asserting that sustainability is now a 'greenwashing' ploy that has lost its direction and meaning, when it is itself actually proof that people don't understand the science or technical approaches that sustainability entails.

Acaroglu (2022) instead shows that sustainability is the overarching concept that encompasses so many of these ideas. At one end of the spectrum there's harm minimisation and efficiency gain, but at the other end, there's the circular economy, doughnut economics, resilience and regeneration. These are all built on the original framework for rewriting the rules of production and consumption to move to a better future with the intent to not make the same mistakes of the past. Movements build and evolve over time; they are an emergent outcome of the system that they exist within. 'Sustainability' is therefore not just an abstract concept of "doing things better" or "doing less harm." It is a set of scientific approaches and technical skills that have been tried and tested over decades of progressive scientific and creative work to find practical ways of redesigning systems and products to be more ethical, equitable, and in fact sustainable. This has always been about getting to a destination of full systems regeneration and restoration, where humans give back more than they take from the natural world. But this can only be reached along a clearly defined pathway — which is exactly what science-based sustainability offers.

Moving along this spectrum and re-interpreting environmental sustainability as environmental resilience is an important step. In terms of nature and the environment, **resilience governance is about allowing environmental systems and ecosystem services to respond to crises, cope with shocks and stresses and rebound** (Rizzi, 2021). In this context the biosphere is seen as the foundation of the global SDG system through such structures as the Global Biodiversity Framework (GBF), the Sendai Framework for Disaster Risk Reduction 2015-2030, the UN Decade on Ecosystem Restoration 2021-2030. In Europe, environmental resilience governance is being promoted by the EU Strategy on Adaptation to Climate Change 2021 (European Commission, 2021i), and the EU Green Deal Striving for climate-neutrality by 2050. The latter focuses on the policy areas of biodiversity for protecting fragile ecosystems, 'farm to fork' for sustainable food systems, climate neutrality by 2050, eliminating pollution, sustainable mobility, sustainable industry and a cleaner construction sector (European Commission, 2019d).

Regeneration tries to go one step further. Whereas traditional sustainability is about doing things that enable humanity to only keep what it currently has through conservation and preservation, and resilience is about successful coping with and surviving environmental and other shocks that are increasing in regularity and intensity, **regeneration is in principle the attempt to (re)create what is thought to be the ideal state of nature albeit one in which humans can live advanced, enriching and flourishing lives**. Regeneration thus still seeks a compromise and balance between the needs of nature and the needs of people and does not envisage nature completely without humanity, though it does seek to regenerate nature to the maximum extent possible that accommodates human lives in this way. It does so by insisting on the idea that people are inherently a part of, and inseparable from, nature, and not apart from it, thereby representing a marked shift from traditional notions of sustainability.

Some of these concepts become philosophical issues that perhaps go beyond the remit of public governance, although we know from the examination of existing paradigms that this remit has constantly changed and will need to continue to do so, possibly in unexpected ways and perhaps as outlined in these strategic challenges. For example, it might mean that, through democratic processes, public governance should incorporate 'beyond-growth', 'counter-growth', 'no-growth' or even 'de-growth' strategies or other alternatives to the current hegemony of the growth-driven paradigm. However, this will need to come with a reality check as it will always be necessary to present actionable paradigms that might at least help society move in the right direction.

These actional paradigms do exist although are still in their early days of development with very limited real initiatives. For example, using the following definitions. A sustainable society can be seen as one that maintains its life support systems in such a way that draws from the environment over time only as much as the environment can produce in that amount of time, and no more. In contrast, **a regenerative society is one that maintains its life support systems so that its actions of drawing from the environment actually serve to help create more production, more health, more resilience, and more longevity in its**

ecosystems than otherwise would have been created without the participation of that society as a feature of the ecosystem (Meyers, 2022).

This potentially provides scope for action, for example through **regenerative design** that is already being deployed. Hernández (2022) is helping private and public organisations to build platform strategies for regeneration that develop strategies for an ecosystem of interconnected actors, who coordinate their activity with a common goal, creating the conditions for the whole to thrive and flourish. Regenerative design, which goes at least one step beyond sustainability, is the term used to describe processes that regenerate damaged ecosystems and can be used in practical settings. The purpose is the same as the purpose of life itself, to create the right conditions for all life to thrive and flourish and is inspired by how living systems work to learn from and imitate them, and thus find solutions to problems that we encounter as humans and that nature has already solved before. As mentioned above, this has also been demonstrated by Bragdon (2016) who showed that many of the most successful companies consciously mimic natural systems to achieve this success.

The regeneration design process is based on a set of principles encompassing: interconnectedness in which relational social beings constantly interact with others and the environment; emergence as the properties or behaviours which emerge only when entities interact in a wider whole; evolution that enables adaption and reorganisation in an environment that continuously evolves; and holism in which systems should be viewed as a whole and where none of them operates independently given they are interdependent and nested in other systems. This approach has been used to successfully develop co-ownership economies using digital technology, smartphones and the internet together with some conjunctural shifts using a wide range of apps and platforms that allow independent workers to be autonomous while co-generating incomes Hernández (2022). An example of regenerative agriculture is given in Section 6.1. Such examples show that public governance has an important role in facilitating the sustainability, resilience and regeneration of nature together with that of society, and that digital technology will be a necessary tool in this.

4.4.3 Dynamic systems

Leading on from the role of regeneration in linking people and planet, **complex systems theory** attempts to construct a holistic framework for accommodating these issues. It is the interdisciplinary scientific study of systems, i.e. cohesive groups of interrelated, interdependent components that can be natural or human-made. Every system has causal boundaries, is influenced by its context, defined by its structure, function and role, and expressed through its relations with other systems. A system is "more than the sum of its parts" by expressing synergy or emergent behavior¹⁴. It has been decisively demonstrated in recent years that complex systems theory is necessary to understand and influence how both human societies and nature behaviour and, especially, how they interrelate and interact with each other. It is thus critical that public governance supports and facilitates this understanding.

However, the system science approach, based on huge arrays of diverse evidence and theory testing, is difficult to comprehend for lay persons, including politicians and decision makers. Thus, good **science communication**, in this field as in any other, is critical. The difficulties this poses are well illustrated by the scientist, James Lovelock, who is most famous, or notorious, for promulgating the 'Gaia' theory that **the Earth is a self-regulating community of biotic and abiotic components**. As a maverick scientist with little early support but much resistance, he independently used experimental evidence and hypothesis testing to demonstrate that living organisms interact with their inorganic surroundings. This is through a synergistic and self-regulating, complex system that helps to maintain and perpetuate the conditions for life, including human life, on the planet. Gaia theory was first conceived in the late 1970s and laid the foundations for Earth system science and a new understanding of the interplay between life, clouds, rocks and the atmosphere (Lovelock, 1979). He also warned, in clearer terms than any of his peers, of the dangers humanity posed to the extraordinary web of relations that make Earth uniquely alive. The Gaia theory was ridiculed when first proposed, both by fellow scientists and many environmentalists, some of whom believed it heralded a "new age nonsense". Today it makes up the basis of much of climate science and is accepted by most scientists and environmentalists (Horton, 2022).

There are many relevant practical applications of dynamic systems relevant for public governance, such as the interactions between food systems and soil ecology as investigated in detail by Monbiot (2022) –s see also Section 6.1. The **interaction of biotic and human-made systems displays classic system**

¹⁴ <u>https://en.wikipedia.org/wiki/Systems_theory</u>

characteristics, including self-regulation through manifold decision-points that are, in effect, random and that reveal emergent and adaptive characteristics. Through this self-regulation, the nested systems tend to find equilibrium themselves, but if pushed beyond **critical tipping points** when certain stress thresholds are reached, total system collapse is likely until a new system equilibrium is reached. There is typically no way of returning to the original state even if the causal stresses are reversed, as also shown by related chaos theory. Traditional farming methods are already pushing the soil systems upon which they rely to the point of collapse after which they will no longer be fit for growing food. This threat is similar to the climate tipping points expected to become self-sustaining shifts in the climate system that would lock-in devastating changes, like sea-level rise, even if all emissions ended straightaway. Others include ice sheet melts, and dramatic changes to ocean currents and major forests. The problem is that precise tipping points remain largely unknown – as it, for example, applies to the well-known 1.5°C global heating limit (McKay, 2022).

In complex systems – including human societies – tipping points can occur, in which a small perturbation transforms a system. Crucially, activating one tipping point can increase the likelihood of triggering another at a larger scale, and so on. A possible positive effect is that such upward-scaling tipping cascades could accelerate progress in tackling climate change. For example, in two sectors – light road transport and power – where tipping points have already been triggered by policy interventions at individual nation scales, positive-sum cooperation, between small coalitions of jurisdictions and their policymakers, could lead to global changes in the economy and emissions. The aim of activating **tipping points and tipping cascades** is a particular application of systems thinking. It gives a different starting point for policy to the theory of welfare economics, one that can be useful when the priority is to achieve dynamic rather than allocative efficiency (Sharpe and Lenton, 2021).

Applying similar system theory dynamics to **social and economic systems** can also be useful for public governance purposes. For example, Centola et al (2018) showed that a **critical tipping point threshold was passed when the size of a committed minority reached roughly 25% of the population**. At this point, social conventions suddenly flip. Between 72% and 100% of the people in experiments swung round, destroying apparently stable social norms. This seems to show that **"the power of small groups comes not from their authority or wealth, but from their commitment to the cause**". Wilkelmann et al (2022) explored the possibility that the *Fridays for Future* climate protests could trigger this kind of domino dynamics. In 2019, Greta Thunberg's school strike snowballed into a movement that led to unprecedented electoral results for Green parties in several European nations. Survey data revealed a sharp change of attitudes, as people began to prioritise the environmental crisis. *Fridays for Future* came close, the researchers suggest, to pushing the European political system into a "critical state". It was interrupted by the pandemic, so the tipping has not yet happened (Monbiot, 2022).

A systems approach is being used by the OECD (2020d) to deal with COVID-19 and future shocks as part of its ongoing work attempting to **understand and influence the operation, management and outcomes of various combinations of economic, social and environmental systems**. A sudden shock, like the pandemic, needs to be examined in terms of the response of the whole system to the incentives that individual components face within complex, nested, and interconnected systems. When a relatively stable system that is more or less in equilibrium, such as global supply chains prior to 2020, is hit by sudden and unexpected disruption as the result of an external shock, the way the system recalibrates and self-organises, or a combination of both, is critical to what happens next. Resilience, or the ability to recover and adapt to unexpected threats, has been a focus of specific aspects of understanding these systems and the resulting impacts when disrupted, especially when interconnectivity between systems is one of the structuring and determining features of the modern world, which is becoming ever more complex and dynamic.

In this context, the COVID-19 shock has spurred the OECD to change its traditional understanding of resilience as the capacity to resist downturns and get back to the same situation as before based on an approach to risk assessment focusing primarily upon system hardness and ability to absorb threats before breaking. **The new approach to resilience focuses instead on the ability of a system to anticipate, absorb, recover from, and adapt to a wide array of systemic threats**. This provides a useful new model for governance systems, as well as others, to understand and influence how complex societal and natural systems react and interact during shocks and to help design and plan new public governance approaches for tackling them.

Most systems theory work and application today is only possible using high-speed powerful computing systems and vast amounts of data that feed into modelling and predictive programmes. These are used, for example, in relation to climate change, the operation of the food system, the effects of a pandemic and the governance systems that need to be set up to understand, regulate and address their impacts. Digital technology, and especially AI, is a fundamental tool in these endeavours.

5 The opportunities and challenges of digital technology for public governance

Some of the likely digital technology implications of the strategic challenges presented in Section 4 have been indicated, and it is clear that in most cases the full cumulative range from Generation 1.0 to 4.0 will be needed depending on particular tasks and purposes. Some of the strategic challenges and the paradigms and models used to exemplify them will also need to focus heavily on services involving intense human-human relationships often without any visible digital involvement, for example for the preventative and community-based services in Section 4.3, although it is also clear that quite sophisticated technology will typically be needed in the back offices or as supplementary service supports in such cases. However, it is also realised that in all cases there are dangers of data misuse, whether wilful or accidental, especially when non-public sector staff are involved. The issue is what role does current and future technology have in supporting, improving and indeed extending the benefits of such tasks without degrading them.

For all potential strategic challenges and paradigms, including both preventative and community-based services, current Generation 4.0 digital technology, built as it is on previous generations, is likely to be able to provide the necessary support for these types of public governance models that can both save money and improve benefits for users. At the very least, these technologies should provide the means, if not the political and professional incentives, for joining-up silos within and between back-offices and between governance levels even when the user interface does not need much technology. However, the putative Generation 5.0 characteristics, with focus on human-machine hybridity under full human control and which build on Generation 4.0 and previous generations, may be even more suitable for addressing all the strategic challenges identified, as outlined in this section. Vesnic-Alujevic et al (2019) drew similar conclusions when articulating four scenarios of the future of government. They found that technology is perceived as a strong driver by different stakeholders in society but also pointed out that, in addition to the many opportunities it brings, complex ethical and legal issues need to be dealt with by the government, businesses and citizens collaborating (see also European Commission, 2022h). A review of the foresight scenarios is under way (Vesnic-Alujevic et al, forthcoming).

5.1 A new relationship between people and technology?

5.1.1 Generation 5.0 technologies and services

Elaborations on Web 5.0 just started and the concept remains under development, but there are clear indications of what it is intended to be with labels like 'emotional' or 'sensitive' web. The stated intentions of Generation 5.0 digital technology are to change the public service emphasis from user- to human-centric with a focus on real people and real peoples' lives, so that services fit these lives, rather than the reverse, using a 'whole-oflife' mindset. The human is as important, if not more important, than the digital and services are not for but with people, given that people expect more 'human touch' through 'digital sanitisation', i.e. removing digital where this does not improve peoples' lives. In principle, all tasks will be under full human control with hybrids of human-to-human, human-machine and machine-machine relationships. This involves balancing efficiency with both effectiveness and 'affectiveness', i.e. by capturing peoples' emotions and a focus on emotional intelligence. The goal is to create emotional interactions between computers and humans, but under the full control of humans using algorithms to personalise, search, and improve experience, and to act for, or on behalf of, a person. Some of the technology enhancements in view are full virtual assistants, decentralised web app platforms and APIs to empower people to regain and retain control of their own data and identity and attempting to avoid the opposite dangers of public and/or private surveillance. It is envisaged that digital emotional intelligence will be possible, as will full digital wallets, full social personas, full digital twinning and full VR (the 'metaverse'?). Decentralised web nodes will be available providing decentralised identifiers, digital sovereignty and cryptographic security (Weston 2022, ByBit, 2022, Krøl 2020).

Generation 5.0 is also about whether humans or algorithms will be 'in control' and the many fears, on the one hand, that 'black-box' AI will take over and that humans will lose control, or whether on the other hand that AI and other advanced technologies will be under full human control and provide an increasingly brighter technology-enhanced future. This will result in '**hybrid' services, with various combinations of digital and human depending on the need, especially individual need, and context**. (See also De Nigris et al, 2020, and Manzoni et al, 2022.) None of this is straightforward, however, as outlined below. According to Chantillon (2021) "...the effect that the disruptive digital technologies can have on a particular policy domain can differ,

depending on factors related to the public administration and the context in which a public administration functions. Instead of considering it as a dichotomy, it is therefore more useful to consider this relation between digital and non-digital as a scale: In some instances the transformation will be more digital then in others, and it is most likely not a yes-or-no situation."

5.1.2 Industrial revolution 5.0

The concept of successive industrial revolutions, that indicate the wider context of economic development in which **digital technology has become today's general purpose technology underlying and supporting virtually all other technologies and everything in a modern society**, was outlined in Section 2.4. Industrial Revolution 4.0 (4IR) aligns well with Generation 4.0 technology, and now Industrial Revolution 5.0 (SIR) is being proposed alongside Generation 5.0 technology. This is now beginning to be identified as having profound implications for new governance paradigms and models. SIR should ideally enable the evolution of the modern manufacturing process in order to allow **humans and machines to perform work hand-in-hand, combining the unique cognitive abilities of workers and the accurate technical expertise of, for example, robots to ensure an innovative culture in the workforce.** "This vision recognises the power of industry to achieve societal goals beyond jobs and growth, to become a resilient provider of prosperity, by making production respect the boundaries of our planet and placing the wellbeing of the industry worker at the centre of the production process. It complements the existing Industry 4.0 paradigm by having research and innovation drive the transition to a sustainable, human-centric and resilient European industry. It moves focus from solely shareholder value to stakeholder value, for all concerned" (European Commission, 2021a).

According to the Regenesys Business School (2020), SIR can be summarised as the combination of humans and machines in the workplace. However, this is vastly oversimplified and does not even begin to explain the magnitude and complexity of the change given that it comes in the context of "**a crisis of trust in technology**." "In the Fifth Industrial Revolution, we're going to have to have... a chief ethical and humane use officer. Are we using these technologies for the good of the world? You can't do business in the Fourth Industrial Revolution without the trust of your employees and your customers and partners." The third and fourth revolutions were hard on humans and hard on the environment. Previous generations had to adapt their lifestyle to what the machines could do. The Fifth Industrial Revolution is different. Human beings are now front and centre in the production process. This source labels SIR as 'personalised' and 'conscious', and suggests it is (will be) happening in the second decade of the 21st Century and will provide deep, multi-level cooperation between people and machines, for example in robotics (see Duch Brown et al, 2021).

The above prognoses for both Generation 5.0 technology and the Fifth Industrial Revolution seem to be a radical departure from the state of almost all digital technology currently in widespread use. On the face of it, they appear to be well suited to fully addressing the strategic challenges above, including for human-facing public services. However, as indicated in Section 8.1, it is clear that political, policy, socio-economic and cultural conditions are the strongest determinants of public governance changes, whilst digital technology and data are best seen as necessary but never sufficient tools in this process.

5.2 The human-technology relationship in focus

It is already clear that digital technology, as the 21st Century's general purpose technology underlining most if not all other technologies and innovations, needs to be seen as complementing and supporting human activity rather than completely replacing it where this would lead to reduced individual or societal benefit. There is **increasing insecurity regarding the relationship between people and technology, where individuals must learn to cope with the consequences of omnipresent machines and networks of a completely new kind**. How are human values, ethics and wellbeing protected and promoted? What are the philosophical implications of blurring boundaries between the physical, biological and digital spheres, as already apparent in the Fourth Industrial Revolution (WEF, 2016)? An **ethical dimension needs to be introduced** in the exploitation, for example, of big data, artificial intelligence, bio-technologies, as well as the tensions between the citizen's right to privacy and to security systems that can deal with huge threats from massive data collection and analysis as well as routine penetrating surveillance by both tech companies and governments (Zuboff, 2019).

For example, there are numerous examples where AI algorithms can accentuate biased decision-making after being trained on biased or incomplete data, and that this is confounded by the fact that how the algorithm actually reaches its decisions which affect the lives of ordinary people are typically hidden as if in a 'black box'.

(See also Annoni et al, 2018, and De Nigris et al, 2020). For this purpose, a useful policy innovation would be to create some form of **independent intermediary between those who develop and use AI, on the one hand, and the people who are subject to decisions taken by AI on the other**. For example, this might be some sort of digital ombudsmen, independent of the market and government, with the legal power, technical expertise and the resources to investigate complaints about unclear or unfair decisions taken by AI, breaches of data privacy or of data misuse on behalf of those affected. In addition, when AI is being developed it might be useful to include a legal requirement to facilitate opening up the code to scrutiny (Millard, 2020).

With such caveats and given that the ITU estimates that over half the total global female population (52%) is still not using digital technology, compared to 42% of all men (ITU, 2019), it is important to undertake proactive and gender-sensitive national high-quality infrastructure rollout, and adequate funding for more general initiatives, such as the promotion of digital literacy. For example, it is reported that the higher the level of internet and broadband coverage, the higher is internet use even for less educated and less skilled individuals. In addition, **household internet access increases the educational attainment of individuals in a given area, even when internet coverage and GDP per capita are relatively low** (Millard, 2015b). Ensuring internet, mobile and broadband infrastructure availability is necessary, but not sufficient, for more adoption and beneficial use of digital technology. It is also necessary to create appropriate incentives, awareness, reward systems, as well as to support provider and user ecosystems, driven by high levels of cooperation and coccreation, in addition to market competition. In the case of digital technology design, there is a strong trend in some developed countries to move away from technology designed purely for specific disadvantaged groups towards '**inclusion by design**'. Technology for social development works best when technology does not set the whole agenda, and policy and procedures do not try to play catch-up for technology's sake.

Important European policy actions are taking place. The EU's 2018 launch of its GDPR (General Data Protection Regulation) is a major attempt by a public authority to address data protection, data privacy and the transfer of personal data outside the regulation's jurisdictional area. The GDPR's primary aim is to give control to individuals over their personal data and to simplify the regulatory environment for international business by unifying regulation within the EU¹⁵. (Berti Suman et al, 2020.) The EC's (2021k) **Digital Decade targets for 2030** aim to empower businesses and people in a human-centred, sustainable and more prosperous digital future in line with the Industry 5.0 policy (European Commission, 2021). This proposes a Digital Compass that evolves around four cardinal points: specialist and basic digital skills, digital transformation of businesses, secure and sustainable digital infrastructures, and the digitalisation of public services. The Digital Decade targets also propose rights and principles for digital citizenship.

5.3 The misleading notion of 'technological neutrality'

It is often presumed that technology is 'neutral', neither good nor bad, because its impacts are determined by how it is used. For example, **social media has had huge positive impacts on the lives of many**, bringing people together globally, extending an individual's horizons beyond the local and even national, enabling communities, campaigns and democratic movements to form, making both governments and large companies more transparent, and enabling families to keep in touch wherever its members live. On the other hand, **so too has social media's misuse mushroomed**, from trolling and bullying the vulnerable online, allowing pedophiles to share child pornography, to the so-called dark web where illegal and dangerous anti-social transactions take place. The democratic and mind-broadening potential of the web has also come under scrutiny, as more and more people only access material they choose to follow. They increasingly ignore, and even more worryingly, are excluded from, other content, leading to so-called filter-bubbles: the more we use search engines, the more they adapt to only feed us what they deduce we like through sophisticated algorithms (Pariser, 2011).

However, closer scrutiny shows that **technology is far from neutral, given it is designed, rolled out and deployed to serve specific societal interests, whether public, private or civil**, so how this takes place is of crucial interest. There are numerous examples where technical advances have been driven by social needs, such as the Linux free open-source operating system for computers, the M-PESA mobile phone money transfer app in Kenya which allows poor people with no bank account to make secure commercial and family transactions over long distances, and the *FixMyStreet* platform used in many more developed economies. **Most technical advances are however market-driven, aimed to increase profits rather than necessarily**

¹⁵ <u>https://gdpr.eu</u>

serving the public good. However, although it is of course true that both goals can be served simultaneouly, much new digital technology is designed to extract market value from individuals and communities rather than increase it. The big Silicon Valley tech companies are flooding the market with their own content, **crowding out local content and languages which could help develop local communities, culture, and companies**. This is resulting in much local income being sent out of the locality, and even out of the country, rather than supporting local content and enterprise. Neither does international finance often invest in local content and language, so the local context is increasingly neither supported nor even recognised as legitimate (OECD, 2011, and UNCTAD, 2020). This is coupled with the fact that every time we go online we leave digital traces and footprints, which are scooped up by tech companies and sold to advertisers who use their knowledge about our personal lives to personalise advertising. **Each of us, individually, is thus the digital 'product' to be sold to the highest bidder rather than simply the consumer of online services**, in a quite bizarre flip of traditional economic relationships (Zuboff, 2019).

Governments and regulators often struggle to understand the pace of change, let alone formulate relevant policies. This prompts the question what technology companies are accountable for and to whom? As technology companies advance, current legal and regulatory frameworks may not address their increasing concentration of power. The rise of digital connectivity also prompts increased cyber-security concerns, for example with the hacking of critical infrastructures such as electricity and transport networks, and the security, ownership and usage of the massive amount of personal data created and shared. (See European Commission, 2022c.)

Although the idea of the '**post-truth society**', 'fake news' and 'fact-free' (political) discourse are not new, their significance enabled by digital technology has reached vastly new heights since 2015. Even though all sides of the debate are guilty of misusing facts, those who attack 'experts' do this much more, for example in the climate change debate. Social media plays a significant role: Facebook, Twitter, Google, etc., say they are working together to try to develop algorithms to filter out false news, 'hate speech' and terrorist propaganda. International organisations have a massive role to play in reinstating the position of facts, evidence and experts in policy debates, but also ensuring they reach **a balance between free speech and freedom on the net**, **on the one hand, and promoting cyber security and tackling hate speech and deliberate falsification on the other**. Finland provides a successful example of a programme to counter fake news by teaching people how to recognise and confront it (CNN, 2019). This 2014 initiative is one layer of a multi-pronged, cross-sector approach the country is taking to prepare citizens of all ages for the complex digital landscape of today and tomorrow. In March 2018, Finland ranked first out of 35 European countries on media literacy in a study measuring resilience to the post-truth phenomenon. Politicians do of course need passion, commitment and vision, but they cannot succeed if divorced from the facts. (See also Lewandowsky et al, 2020, and Mair et al, 2019.)

5.4 Technology out of control?

There are also many current challenges and threats which need to be addressed, partially arising from the digitisation of government, such as **when data and information are mis-used, manipulated or distorted** without any factual or objective basis. In this context, vigilance as well as new forms of security are needed to address questions such as how do we know the data is correct, and are 'black-box' algorithms dangerous when it is not clear how they function? There are also potentially bigger challenges with big data, such as cyber-crime and warfare as well as the creation of the so-called 'dark web' and other subversive developments.

These and related issues are relevant to a re-appraisal of the way our societies are governed. According to Mair et al (2019), complexity, wicked problems, the abundance of information, the pace of change, uncertainty, misinformation, populism, polarisation as well as new governance models and digital technologies are creating the need to change how policy is made. Major issues to confront include (Mair et al, 2019):

- **Misperception and disinformation**: our thinking skills are challenged by today's information environment and make us vulnerable to disinformation. We need to think more about how we think.
- Collective intelligence: science can help us re-design the way policymakers work together to take better decisions and prevent policy mistakes.
- **Emotions**: we cannot separate emotion from reason. Better information about citizens' emotions and greater emotional literacy could improve policymaking.

- **Values and identities**: these drive political behaviour but are not properly understood or debated. (See also Scharfbillig et al, 2021 and Section 3.3)
- **Framing, metaphor and narrative**: facts don't speak for themselves. Framing, metaphors and narratives need to be used responsibly if evidence is to be heard and understood.
- Trust and openness: the erosion of trust in experts and in government can only be addressed by greater honesty and public deliberation about interests and values.
- Evidence-informed policy-making: the principle that policy should be informed by evidence is under serious attack. Politicians, scientists and civil society need to defend this cornerstone of liberal democracy.

6 Implications for three example sectors

6.1 Food system governance

6.1.1 Governing the food system prior to 2020

Starting in 1962, the **EU's Common Agricultural Policy (CAP)** has implemented a system of agricultural subsidies based on the land area farmed and has since undergone several changes to reduce the budget cost (from 73% in 1985 to 37% in 2017) and to consider rural development as part of its aims. However, costs have continued to be criticised as well as the environmental and social effects. Proposals and attempts to modemise the CAP have included social innovations at the local level, from community-supported agriculture schemes and farmers' markets to the creation of local food policy councils and urban food policies. These initiatives are highly promising in terms of sustainability (e.g. reducing environmental impacts and reclaiming value for small-scale farmers/food businesses) and in terms of reconnecting actors (e.g. producers and consumers, citizens and local policymakers) in a way that restores democracy, accountability, and trust in food systems (iPES Food, 2019).

EU and national policies have, however, been generally ill-equipped to encourage this type of experimentation. For example, local food system initiatives tend to be small-scale and/or urban-based, often making them ineligible for CAP funding. Where supportive EU policy frameworks do exist (e.g. flexibilities in public procurement and food safety rules to support small-scale farmers), the opportunities are undercommunicated, ineffectively implemented at national/local level, or subordinated to competing priorities such as boosting competitiveness in conventional markets. Supporting local experimentation, promoting social innovation, and building sustainable food systems at the territorial scale remain à la carte options rather than obligations for Member States. Though opportunities exist for local and regional actors to share best practices with one another, far fewer are created for EU policymakers to learn and shape EU-level policies and programmes to further support these initiatives on the ground (iPES Food, 2019).

The iPES Food (2019) report therefore proposed a new EU Common Food Policy setting a direction of travel for the whole food system, bringing together the various sectoral policies that affect food production, processing, distribution and consumption, and refocusing all actions on the transition to sustainability. It mapped out a new governance architecture for food systems by reclaiming decision-making processes from powerful lobbies, bringing new actors around the table, shaping policies in more democratic ways, and allowing new priorities and new coalitions of interest to emerge. This was based on four objectives sequenced over the short-, medium-and long-term, and underpinned by new ways of making policies: integrate across policy areas to put an end to conflicting objectives and costly inefficiencies; integrate across governance levels to harness grassroots experimentation and align actions at EU, national, and local levels; ensure governance for transition to overcome short-term thinking and path dependencies in a way that sectoral policies cannot; and introduce democratic decision-making through public participation in policymaking.

These initiatives were also supported by the emerging concept of local, and particularly **city-region, food systems** as the most suitable scale for improving food availability, quality and environmental benefits, for example through sustainable agri-food strategies, for example through the Milan Urban Food Policy Pact (2018). Building on these developments, Moragues-Faus and Battersby (2021) identify **three core perspectives in urban food governance**: a shift towards systemic focus on local food systems; increased engagement with the complexity of governing the different geographic scales; and a growing focus on the relationships of urban food governance with the broader urban policy agenda. The authors see the need for a stronger conceptualisation of the urban context for food; a clearer definition and articulation of the nature of governance and policy at this level; and a more engaged focus on issues of power and inequities in the urban governance system. Cohen and Ilieva (2021) show how local, as well as national and European level policy makers are

starting to acknowledge that the food system is multidimensional, that social determinants affect diet-related health outcomes, and the need to move away from focusing food programmes and policies narrowly only on food access and nutritional health.

In relation to urban food policy governance, Parsons et al (2021) point to the importance of institutions as policy-structuring forces, the need to rebalance national-local powers and to develop cross-cutting food plans. Clark et al (2021) emphasise the role of **community food infrastructures** and the importance of critical middle infrastructures to connect production with consumption and larger markets, thereby building resilience through intermediate markets. The overall thrust of this focus is about the importance of linking urban food policies with other urban policies, new types of place leadership for example through the anchor institutions and middle infrastructures of community-wealth building and 'new localism' initiatives. Thus, before COVID-19, the boundaries of food governance were already spatially focusing on the local and the city-region scale whilst also simultaneously broadening in policy terms to encompass a much wider range of issues and domains not previously considered within the purview of food policy, such as local labour, housing, and education policies.

Local food system governance involves re-scaling the food system significantly away from the conventional approach and transforming towards more sustainable and resilient food systems with significantly shortened value chains in an increasingly **circular city-regional food cycle**. Figure 4 sketches the overall re-scaling approach that, at its base, deploys circularity principles which look beyond the current 'take, make, and waste' industrial model to design out waste and pollution, keep products and materials in use, and regenerate natural systems. Circularity also boosts local commerce, jobs, social inclusion and more responsive local governance. The household, neighbourhood, city and peri-urban area are nested within the wider regional, national and global food systems. Fundamental questions include what is the hinterland/catchment area required to provide a town or city with its basic needs for nutritious, safe, secure, sustainable local and seasonal food, what the relationships with rural areas, how is this organised and governed? To answer these questions, a transformation is required from a predominantly international and planet-wide system towards a more circular city-region food system that becomes much more self-sustaining and resilient. This implies a substantial shift to the left in Figure 4 with the main emphasis on strong interrelations between the household, neighbourhood, town/city and region. A nexus approach and thinking is thus needed – a city is only as resilient as the surrounding region is in terms of water, energy, food, logistics and natural ecosystems.



Figure 4: Re-scaling the food system

Source: Adapted from Diez, et al (2019)

This is the only feasible way in which both biological and technical materials can become part of a circular food ecosystem that is, in practical terms, able to both reduce waste and increase efficiencies along the whole value chain. This shift does not imply that city-regions should or will become cut-off in terms of food from their wider national or international context, or from global interactions and trade. However, the move to the left in Figure 4 needs to be substantial. Unlike biological and technical materials, the figure depicts how **data, knowledge, know-how and learning are traded globally, for example through the sharing of good practices**. These have very low transaction and transportation costs using mainly digital but also in-person contact, so are well suited to very widespread exchange and trading from hyper-local up to global level. Such a transformation can also lead to huge environmental, social and economic benefits making the short-term transformation costs significantly worthwhile in aiming at the desired outcomes and impacts over the medium and longer-term. Moving to much more local food systems will also help to break the extreme dependency of nations (both developed and, importantly, developing countries) on food shipped from distant places. If food production becomes largely distributed, every locality could become much more autonomous, also with the assistance of new technologies like vertical farming, hydroponics and precision farming including precision fermentation (Monbiot, 2022).

6.1.2 The shock of COVID-19

The **Farm to Fork Strategy** published in May 2020 (European Commission, 2020b) aims to increase the social, environmental, and economic sustainability of the EU food system, covering all stakeholders (farmers, consumers, retailers, and processors). Since its publication, successive crises, including the COVID-19 pandemic, the war in Ukraine, and a severe drought affecting the whole continent, have underlined the urgent need for a systemic transition of EU food systems to reinforce their resilience and sustainability. On 2 December 2021, a **new CAP was agreed due to run from 2023 to 2027** (European Commission (2021j) designed to secure the future of agriculture and forestry, to achieve the objectives of the **European Green Deal** (European Commission, 2019d) and to pave the way for a fairer, greener and more performance-based CAP. It will seek to ensure a sustainable future for European farmers, provide more targeted support to smaller farms, and allow greater flexibility for EU countries to adapt measures to local conditions.

Food is arguably our most vital strategic commodity. It is central to our existence, not just for sustenance and survival but as a huge contributor to our cultural, social and economic lives. The food industry also has huge environmental impacts. So, when food systems of food preparation, consumption and diet are disrupted by large scale social and economic shocks like the financial crisis, COVID-19 and the war in Ukraine, this has huge impacts. Millard (2022) shows that **the majority of European households significantly changed important aspects of their food behaviour during the first wave of the pandemic, many of which are likely to persist into the future. Across all household types, the pandemic dramatically accelerated the previous slow trend towards more local and seasonal food delivered along short supply chains, a move to smaller independent retail outlets, and much greater food awareness and own growing of food where feasible, large decreases in both food waste and in new types of food and recipes. The transformation of the market system for food, heretofore based on large centralised organisations, long supply chains and ever-increasing globalisation, is being turbo-charged.**

However, the food system is multidimensional and impacts individual households differently, especially in relation to household composition, education, income and geography, which are themselves closely inter-dependent. Households with children tend to be the largest households with the greatest range of food tastes and needs, so are likely to be the most responsive to food system shocks. They have typically changed their food behaviour much more than other households during the pandemic. Their size often means tighter budgets and this shows in their relatively large reduction of fresh food but also largest increase in comfort food consumption, no doubt because of their children's demand for sugary foods whilst stuck at home during school shutdowns. Households with two or more adults were also relatively responsive to the shock, although these households tend to consist of much older, often retired, people many of whom are living on small incomes and are very concerned about their health and, especially since COVID-19, about the impact of food on this. Both households with children and with two or more adults also generally suffer much more than single households from food vulnerability and stress and this was magnified if the household lost income during the pandemic due to job loss, furloughing or reductions in working hours (Millard et al, 2022; Hristov et al, 2022).

Somewhat similar patterns were seen in relation to the educational level of the household, so that the higher levels are the most resilient and least likely to change their food behaviours during the pandemic. The less educated the household, the more likely it was to decrease fresh food consumption and increase processed

food consumption, although the university educated households did increase their consumption of comfort food more, probably because of their typically higher incomes and easier access to shops. The less educated households also switched less to local producers, used home delivery, unpackaged and organic food much less and engaged less often in enhanced food preparation behaviours, as well as decreasing their food waste the least. It seems likely that awareness of the links between food and health were less clear to these households, a supposition reinforced by their much lower increase in evaluating the overall importance of food. These households were also more likely to revert to foodbanks during the pandemic as well as obtain free food from other sources, a situation reinforced by the fact that many more of them lost income.

Households with children and that lost income during COVID-19 tend also to be those with lower overall household income per person before the pandemic and thus more likely to be fragile at the outset, so the pandemic made their situation relatively much worse. These households also tended to eat less healthily during the pandemic away from fresh food to more processed and sugary foods and alcohol. (Hristov et al, 2022). Overall, **more vulnerable households (however measured) suffered more during the COVID-19 shok**, but this also meant that both their behaviour and awareness around the health-food relationship also improved significantly because they were starting from a much lower base. More affluent households already had relatively high levels of food behaviour and awareness around the health-food relationship and, although this increased during the pandemic, this increase tended to be less given it was starting from a higher level.

So, in many ways, the COVID-19 food shock pushed more vulnerable households to catch-up to some extent with the already relatively high levels of food-health awareness in less vulnerable households, despite the serious difficulties they encountered. Indeed, more vulnerable households also say their changed food behavior is more likely to continue beyond COVID-19 than do less-vulnerable households. This includes, for example, greater increases in shopping with local producers and in more local shops, growing own food, using a wider range of food dishes and recipes, greater use of unpackaged food and much less food waste. Thus, a useful policy guide during this age of turbulence would be to put in place **measures to support the positive food behavioural changes of all households with a specific focus on the more vulnerable households and those with lower incomes**. Vittuari et al (2021) document how recent crises have exposed the fragility of food sovereignty in cities and confirmed the close connection urban dwellers have with food and suggest how citizens would accept and indeed support a transition toward more localised food production systems. They propose the reconstruction and upscaling of such connections using a 'think globally act locally' mind-set, engaging local communities, and making existing and future citizen-led food system initiatives more sustainable to cope growing populations.

In terms of geography, there is a strong tendency for more or less regular changes in food behavior outwards from a country's capital city centre to its rural periphery, i.e. down the metropolitan hierarchy and along the urban-rural continuum. This is directly related to food market conditions and supply logistics that decrease both quantitatively and qualitatively from the centre to the periphery in terms of density, connectivity (including digital connectivity), food availability infrastructure (although there is more own food growing in rural areas due to greater land availability) and power and decision-making (Millard et al, 2022). Thus, rural areas suffered from greater food challenges than urban areas during the pandemic, even more than can only be ascribed to income levels alone. This also reflects the need for 'rural proofing' (see Section 4.2.2). However, there is also an important counter trend with some movement of population out of both the largest, typically capital, cities and from the so-called second-tier, typically older industrial, cities towards smaller cities in or beyond the suburbs and in peri-urban and adjacent rural areas. These smaller cities are also growing in size fed by some rural depopulation as more desirable destinations than the larger cities with their higher rents and living expenses. (See also Section 4.2.2.) As the new growth poles, the smaller cities tend to exhibit much oreater social and economic cohesion and often performed better in terms of higher levels of food behaviour and resilience during the pandemic, even when compared to capital cities which often themselves have much more mixed affluent and vulnerable populations.

The importance of the **sustainability of city-region food systems is inevitably linked to the topic of shorter food supply chains**, especially during shocks, and are associated with extensive good practice evidence related, for example, to the re-connection of food producers with consumers (Grando et al 2017), social sustainability (Vittersø et al, 2019), or building transparent food supply chains with the fair distribution of power among actors (Kessari et al, 2020). In addition, short food supply chains are associated with the production of quality and safe food when consumers buy products from local trusted suppliers who are able to guarantee genuine and safe products (Baldi et al 2019). These developments highlight the vulnerability of the present largely globalised agri-food systems as well as societies in the relatively developed world. Matacena et al (2021) see this situation as an opportunity to strengthen the sustainability agenda, e.g. by pursuing the **EU's** **Farm to Fork strategy** (European Commission, 2020b) and thus enhance the resilience of regional and local food systems and empowering consumers to make informed food choices. Murphy et al (2021) mention the importance of local food supply chains for supplementing the global market and ensuring normal product flow during emergencies, whilst Vidal-Mones et al (2021) propose strengthening independence in the form of support for local and seasonal consumption. McEachern et al (2021) point out that while attention has been "predominantly focused on larger retail multiples, more attention should be paid to small, independent retailers as they possess a broader, more diffuse spatiality and societal impact than that of the immediate locale. Moreover, their local embeddedness and understanding of the needs of the local customer base provide a key source of potentially sustainable competitive advantage" and thus help underpin both urban and community resilience.

6.1.3 Food system examples

An example of relatively advanced city-region food governance in practice is the **Milan Urban Food Policy** Pact (MUFPP) supported by the Urban Agenda for the EU that focuses on the means of implementation and oovernance (European Commission, 2022f). The MUFPP is an international agreement among cities committed to develop urban food systems that are sustainable, inclusive, resilient, safe and climate friendly, that provide healthy and affordable food for all, thereby improving food security and nutrition as core requirements for sustainable urbanisation. MUFPP was signed on October 2015 by more than 100 cities from all over the world representing 450 million people. MUFPP cities believe that urban food systems are at the core of sustainable development and a crucial part of the nexus of climate, energy, water and other natural resources, together with social inclusion and equity. The Pact and its framework for action represent a unique platform to support coordinated food policies and foster urban-rural linkages (MUFPP, 2015). In terms of governance, the MUFPP Monitoring Framework (MUFPP. 2018) encourages interdepartmental and cross-sector coordination internal to city governments working to integrate urban food policy considerations into social, economic and environment policies, programmes and initiatives, such as, inter alia, food supply and distribution, social protection, nutrition, equity, food production, education, food safety and waste reduction. Such inter-departmental and cross-sector institutional mechanisms or bodies (food bodies, units or teams), will enhance dialogue and coordination, policy integration, impacts, and efficiency gains by 'breaking down institutional silos'.

Analysis of various successful examples of such coordination mechanisms shows that key government actors include authorities that are responsible for: agriculture, health/nutrition, social protection, economic development, markets, planning, transport, and climate change. It is clear that the mere presence of an interdepartmental / sectoral coordination body will not provide sufficient indications on actual levels of coordination, results-impacts and gains. It is therefore also important to assess the functioning and effectiveness of the coordination body (e.g. is it having regular meetings; does it have sufficient human and financial resources to make sure that the coordination body/mechanism functions; does the coordination mechanism actually result in concrete collaboration initiatives and city policies; are the functioning of the coordination body, its activities, results and impacts monitored to drive analysis of lessons learned and impacts as a basis for further planning and improvements?) Successful examples also highlight that clear and strong institutionalisation of the coordination body/mechanism in the local government structures and budgets reduces the risks that changes in city administration can bring, e.g. shifts in allocation of budgets, and is key to mainstreaming food in municipal policies. Securing the food body and programmes through legislation also makes them more resilient to government changes. In order to gain broader political and public support, transparent information sharing on the roles, activities and achievements of the coordinating body/mechanism is also crucial. (MUFPP, 2018).

At European level in March 2022, just after the start of the Russian invasion of Ukraine, France's President Macro argued for greater food independence in Europe by prioritising productivity over sustainable farming goals in the EU's *Green Deal*. There is clearly a dilemma here between food productivity and security, on the one hand, and food sustainability on the other. Macron presented his five-year programme for independence in a "stronger Europe" where agriculture and industry are crucial. In line with the French 'resilience plan', Macron is proposing **greater "agricultural, industrial and creative independence**". Refocusing on production is Macron's priority, particularly as food security has become a significant challenge due to successive global crises and the "deep food crises" expected to follow. The French president wants to adjust the EU's Farm to Fork strategy, which is based on a world "before the war in Ukraine," as it could result in a "13% reduction in production". "Europe cannot afford to produce less," he said, calling for a review of the Farm to Fork objectives (EURACTIV, 2022a).

6.2 Energy system governance

6.2.1 Governing the energy system prior to 2020

In 2016, the European Commission published a proposal for a new regulation on the Governance of the **Energy Union** based on five dimensions: decarbonisation, energy security; energy efficiency; an internal energy market: and research, innovation and competitiveness (European Commission, 2016b). The proposal replaces the earlier Monitoring Mechanism Regulation, and requires Member States to adopt National Energy and Climate Plans. These replaced existing separate plans for renewables and efficiency and help streamline related obligations for planning and reporting. Monitoring progress takes place under a new annual cycle in the framework of the Energy Union, which is similar to the European Semester. The proposal puts forward requirements for the content of the plans, including a detailed and binding template for Member States to use. Importantly, it also contains the process for adopting and monitoring national energy objectives on renewable energy deployment and energy efficiency, as Member State contributions to EU level binding targets in these areas. Many of these national plans, as well as many other energy initiatives, were focused on **much more** local energy generation and use, implying that many different levels of government would need to work closely together and involve a variety of actors (Wade et al, 2008). According to Dobravec et al (2021), "a sustainable energy system based on renewables, energy-efficiency, decentralisation of energy generation and synergies between different sectors requires new energy planning methods and policies. Energy transition and climate change mitigation achievement can no longer be seen only through top-down activities from a national government." Local and regional governments thus have a crucial role in delivering relevant public policies within a multilevel governance system.

According to the European Commission (2022b), "energy communities are citizen-driven energy actions that contribute to the clean energy transition, advancing energy efficiency within local communities. Energy communities organise collective and citizen-driven energy actions that help pave the way for a clean energy transition, while moving citizens to the fore. They contribute to increasing public acceptance of renewable energy projects and make it easier to attract private investments in the clean energy transition. At the same time, they have the potential to provide direct benefits to citizens by increasing energy efficiency, lowering their electricity bills and creating local job opportunities. By supporting citizen participation, energy communities can help provide flexibility to the electricity system through demand-response and storage. Energy communities offer a means to re-structure energy systems by harnessing energy and allowing citizens to participate actively in the energy transition and thereby enjoy greater benefits." "Energy communities can take any form of legal entity, for instance that of an association, a cooperative, a partnership, a non-profit organisation or a small/mediumsized enterprise. It makes it easier for its citizens, together with other market players, to team up and jointly invest in energy assets. This in turn, helps contribute to a more decarbonised and flexible energy system, as the energy communities can act as one entity and access all suitable energy markets, on a level-playing field with other market actors." (The European Commission's JRC also provides scientific support to the design of energy communities, EU Science Hub, 2022.)

Schmid et al (2020) define community energy as "formal or informal citizen-led initiatives which propose collaborative solutions on a local basis to facilitate the development of sustainable energy technologies". There are many advantages, such as enhancing local acceptance of renewable energy, regional value creation and energy democracy and justice. As a legally constituted form of corporation with democratic membership control, constrained profit distribution, and open membership, energy cooperatives are one of the main governance forms, defined as an autonomous association of persons united voluntarily to meet their common economic, social, and cultural needs and aspirations through a jointly owned and democratically-controlled enterprise. However, such cooperatives do not function in an institutional void but also require interaction with and the support of governments, especially at the local level, although this itself also takes place in a multi-level governance system. Important issues include the decision-making processes, the implementation of policies, the involvement and participation of public and private actors in the problem-solving process and the ways (local) governments respond to community energy initiatives (Schmid et al, 2020; Schwalb and Walk, 2007).

A key challenge of such (new) types of governance is to balance between the enabling and authoritative modes of governing, and there are calls for a shift in the local governments' roles and responses to the emergence of community energy and claims that new civil initiatives ought to be recognised as "**opportunities for co-creating new formal arrangements that serve the public interest, rather than being merely viewed as 'difficult**'" (Avelino et al, 2014). An important balance needs to be struck between, on the one hand, the public authority's need to engage and to create space for private and community actors that contribute to widespread public value and, on the other hand, preserving the public interest of all citizens given that an energy

cooperative is first and foremost accountable to its members and not to the entire community. In terms of how close the relationship should be between the local authority and the energy co-operative, Hoppe et al (2015) demonstrate that close interaction and a high degree of trust are preconditions for success. In contrast, Frantzeskaki et al (2013) point to examples where the community energy organisation deliberately held back from close contact with the authority as they doubted the benefits of so doing. In start-up initiatives, however, it seems that partnering with the local authority is likely to be crucial for success (Hufen and Koppenjan, 2015), although, at the same time, the authority also needs to be considerate of the energy cooperative's independence, and that a "too pro-active a role [of the state] could produce dependency relations and crowd out civil society activism" (Healey, 2015).

Schmid et al (2020) report on an interesting feature of multi-level governance systems, at least in the context of local energy cooperatives. For example, in the vertical dimension of federalist systems as in Germany and Switzerland, local governments need to adapt to such top-down structures when considering the local context and the space for bottom-up developments by compensating for weaknesses or shortcomings at other levels of government (Ehnert et al. 2018). Schmid et al (2020) "suggest that local governments should be given sufficient financial capacities and autonomy to strengthen implementation of a decentralised energy transition that involves citizens. However, they also recognise that municipal structures alone are often insufficient and that superordinate policies, especially national subsidies, remain essential. Hence, policies at the municipal and national levels should take greater account of citizen initiatives, such as energy cooperatives, which exhibit various non-commodifiable advantages relevant to the energy transition." "Federalism works when governments at one level of the system are able to compensate for weaknesses or defects at another level" (Derthick, 2010). However, how this compensation takes place is not pre-determined but depends both on the scope of action provided to them by the federalist system but also on their willingness to act. Schmid et al (2020) note that "national and state governments can incentivise action by entrusting local governments with certain tasks or by supporting and motivating them to pursue their own energy policy", and they can directly shape the activities of civil society organisations, such as community energy cooperatives. In this context, Markantoni (2016) notes that national-level energy policy and regulation directly impact and probably motivate community energy initiatives, for example through incentives for renewal energy development like feed-in tariffs and electricity market regulation.

According to Gohari and Larssæther (2019), lessons from the Zero Village initiative in Bergen, Norway, on sustainable energy planning as a co-creative governance challenge requires different, but interlinked strategies, technologies and policies. This implies a complex array of overlapping systems that are shaped by diverse actors' interventions and requires conceptualisation of co-creative governance, and the dynamic interplays between power relations in the face of conflict of interests. It is necessary to go beyond the traditional division of governance network between private, public and academia to investigate the political structure underpinning the functionality of governance. The aim is to understand how different multi-level governance systems deal with the competing interests, asymmetrical power and the mobilisation of resources for goal achievement in the case of Zero Village Bergen. The findings show that, in the governance transition to sustainability, the different actors have needed to exercise power in the context of (shifting) power relations in an interesting illustration of a quadruple helix innovation system in which the knowledge community plays an active part. In the current wave of smart city initiatives, municipalities and universities are increasingly taking part in cross-sectoral partnerships and platforms. This example demonstrates that there is a substantial potential for role conflict between public authorities as guardians of the common interest, and the formal and informal bindings that occur when they form alliances with commercial actors who will also seek to pursue their self-interest. It is thus important to strike the balance between securing democracy and legitimacy of planning through 'due process' and the need for new forms of governance with respect to dealing with pressing energy and climate concerns.

It is, however, also clear that the knowledge community holds the potential of reducing conflict between private and public actors by creating arenas where common narratives can be developed across diverse societal interests over time. In pursuing this role, it is crucial to combine insights into the paradoxes and dilemmas of real-life cases, being aware that preconceived expectations about the role of the knowledge sector in unfolding governance processes may be challenged. As a result, instead of considering decisions as resulting from the intention and interests of independent actors, attention should be paid to the interaction patterns and the ways in which individual actors and organisations evolve over time. Reflecting on the multi-actor model (Avelino & Wittmayer, 2016), it seems that the temporal dimension of the role of actors is not given the focus it deserves. In this regard, the need for developing governance models that better capture the iterative nature of real-life planning processes should be stressed (Gohari and Larssæther, 2019).

6.2.2 Europe's energy governance shaken by geo-political conflicts

Pre-Ukraine war challenges facing the EU in the field of energy were already serious, such as increasing import dependency, limited diversification, high and volatile energy prices, growing global energy demand, security risks affecting producing and transit countries, the growing threats of climate change, decarbonisation, slow progress in energy efficiency, challenges posed by the increasing share of renewables, and the need for increased transparency. Objectives were set, including diversification of Europe's energy sources, ensuring energy security through solidarity and cooperation between EU countries, ensuring the functioning of a fully integrated internal energy market through the Energy Union, and enabling the free flow of energy through the EU via adequate infrastructure and without technical or regulatory barriers. In support of these, reductions on dependency on energy imports, cutting emissions, driving jobs and growth, decarbonising the economy and promoting research and innovation are considered essential (European Parliament, 2021b; European Commission, 2015b).

In March 2022, after the invasion, the European Commission proposed **updated plans to make Europe independent from Russian fossil fuels well before 2030, starting with gas**. This includes a series of measures to respond to rising energy prices and to replenish gas stocks for next winter. Europe's *REPowerEU* initiative seeks to diversify gas supplies, speed up the roll-out of renewable gases and replace gas for heating and power. This can reduce EU demand for Russian gas by two thirds before the end of the year. Commission President von der Leyen said: "We must become independent from Russian oil, coal and gas. We simply cannot rely on a supplier who explicitly threatens us." Executive Vice-President for the European Green Deal, Timmermans said: "It is time we tackle our vulnerabilities and rapidly **become more independent in our energy choices**. Let's dash into renewable energy at lightning speed. Renewables are a cheap, clean, and potentially endless source of energy and instead of funding the fossil fuel industry elsewhere, they create jobs here. Putin's war in Ukraine demonstrates the urgency of accelerating our clean energy transition." European Commission (2022e).

At the same time that the current shocks have shown the need for even greater focus on local energy generation and use, especially through citizen-driven energy communities (European Commission, 2022b), Osička and Černochac (2022) point out this implies more European level intervention in energy as solutions to the crisis require levels of coordination and resource mobilisation that individual member states or private actors cannot provide. The EU has the resources, knowledge base, and determination to turn the crisis into an opportunity. However, there are also dangers if this is uncoordinated or mismanaged that the European response might make the matters even worse, triggering a political crisis and eventually also a crisis of legitimacy. At global level, the WEF (2022c) proposes five strategies to navigate the shifting frontiers of the energy transition: legally enforce climate targets through domestic policies; **shift energy security planning** to a 'just-in-case' rather than the traditional 'just-in-time' model through 'dual diversification', i.e. in terms both of diverse partners and diverse sources of supply; de-risk clean energy investments; prioritise equity and justice; and learn from pandemic management to overcome behavioural barriers to energy efficiency. In order to strengthen Europe's overall economic resilience, Europe's energy challenge also needs to be seen as an economic opportunity through the *REPowerEU* initiative and to see it as part of the promise to delivery Europe's strategic independence. It must also be linked directly to making labour markets more resilient and strengthening Europe's global leadership on food security (WEF, 2022b).

6.2.3 Energy system examples

Europe's swift transition to a sustainable, low-carbon future will not happen without the engagement and involvement of citizens producing and consuming energy locally. Communities across Europe are increasingly making, consuming and selling their own energy, a trend the EU sees as vital if the bloc is to meet its climate targets. According to the latest data, 2 million Europeans are now involved in 7,000 local energy communities across the continent, with numbers growing rapidly since EU directives promoting clean energy and energy communities were introduced in 2018 and 2019. The technology is now available to design and install systems to produce, store and trade electricity, as well as control and optimise consumption (Henley, 2022). Highly centralised electricity production and distribution systems on their own will not be able to adequately handle the huge increase in energy demand, so it is necessary to decentralise by producing and consuming more energy locally with sources like solar and wind. There is also a need to boost storage and smart solutions for efficient energy management, all of which means involving ordinary citizens. The EU's *GRETA* project is thus working to define and enable what it has dubbed 'energy citizenship' by facilitating energy transition through active citizen participation. According to *GRETA*, all the research on **peer-to-peer energy**

sharing models show they are "far more accessible, democratised, collaborative and socially just"

than traditional top-down energy markets. "Fundamentally, they thrive on social interconnectedness among end-users, rather than being based on competing economic self-interests" (GRETA, 2022).

In 2016, the Edinburgh Community Solar Co-operative (ECSC) developed a community project that saw solar PV arrays installed across 24 public buildings owned by the City of Edinburgh Council. The project has been a huge success, with each site producing more than their initial estimated generation in the first year of operation. The electricity generated also provides an income to the cooperative, a proportion of which is allocated to its Community Benefit Fund, which is focused on sustainability and alleviating fuel poverty. Since 2016, ECSC has added six additional sites. At the outset, ESCS aimed to maximise the solar generated electricity at a selection of sites by installing battery storage. The batteries would ensure that more solar generated electricity is used, improving the efficiency of each site, as well as reducing their carbon footprint. The project would also provide grid services to assist the development of a smart local energy system. ECSC has an agreement with the City of Edinburgh for the purchase of any solar generation electricity used on site. The additional storage will generate further income in sales, as well as from the additional supply it will provide to the grid. All of the additional income generated will go into its community benefit fund. ECSC received funding from the Scottish Government's Community and Renewable Energy Scheme (CARES) to part pay for the battery installation. In terms of outcomes and achievements, three sites were chosen and a total of 156kWh of battery storage was installed. A full monitoring and evaluation process is currently taking place with expected benefits including more locally generated electricity which will also reduce the electricity imported from the grid, thereby lowering carbon footprints and improving energy efficiency. The batteries will be able to be aggregated with other storage sites to be remotely controlled and could provide services to the national grid that will increase ECSC's revenues. These additional monies will be allocated to the community benefit fund. (Local Energy Scotland, 2022)

6.3 Work and employment system governance

6.3.1 Governing work and employment prior to 2020

Globally and to an important extent also in Europe, the period since the 1980s up until 2019, exacerbated by the 2017-8 financial crisis, has seen taxation become less progressive and much of politics has increasingly been dominated by money and special interests. Side by side with this, **welfare states in most countries have become less generous with a labour market that has become much more volatile**. Many people began to work in informal, precarious or 'gig' economy jobs, typically based on new digital technology, leading to increasing numbers of the 'working poor' as workers only work and get paid when a customer places an order leading to a poverty trap associated with zero hours contracts. Numerous governments, including some in Europe, have hailed their low unemployment rates as major achievements but have sidestepped the issue of the poor remuneration, quality and security of many of the jobs existing today. (Ortiz-Ospina and Roser, 2016; Millard, 2020.) Prior to the COVID-19 pandemic, there was widespread unease about the predicted staggering impacts of digital technology during the Fourth Industrial Revolution on work and employment. These include (WEF 2016): job loss due to automation (Frey and Osborne 2013); automation replacing not just no- and low-skilled workers but now also managers and professionals; children entering school today can expect to end up in jobs that currently do not exist; and fewer jobs despite similar market values compared to 1990.

These issues require urgent policy action as **governments have to cover the social and fiscal costs of long-term unemployment and job insecurity** (Misuraca et al, 2018). The temporarily employed also run a higher risk of poverty than permanent workers, mainly due to lower wages. Al and robotics can threaten the quality and remuneration of work, as well as work itself. For example, from the perspective of embracing and promoting gig jobs, outsourcing work and highly flexible labour markets as this maximises efficiencies and reduces costs, websites and artificial intelligence 'chat bots' could replace up to 90% of the UK's central government's administrators, as well as tens of thousands in the health service and doctors' surgeries by 2030 – saving as much as £4bn a year. Even nurses and doctors could fall victim to machines which can outperform humans at some diagnoses, routine surgical procedures, and data collection (Reform Thinktank, 2017).

The UK's Reform Thinktank (2017) argues that public services should become more flexible by embracing a gig economy where workers support themselves through a variety of flexible jobs using online platforms. For "20% of public-sector workers" who "hold strategic, 'cognitive' roles", they will "use data analytics to identify patterns – improving decision-making and allocating workers most efficiently". Such 'contingent labour' platforms could suit hospitals and schools instead of traditional agency models, as well as organisations which experience seasonal peaks in demand such as the tax agencies at the end of the tax year. Although much of this remains

speculation, there will be significant impacts on welfare, jobs and the organisation of work. **The more transactional the job is, the more likely is there to be automation and greater inequality, as automation is fundamentally the substitution of capital for labour**. A 'race to the bottom' may create a growing 'precariat' as an emerging global class with no financial security, job stability or prospect of career progression (Gayle, 2017).

This disproportionally affects women and girls, as well as other groups who are often excluded in terms of ethnicity, disability, language, education, sexual orientation, etc. Precarious work is much more prevalent in female employment. (McDowell 2016) This contrasts sharply with the comparative stability of male employment in both blue-collar and white-collar occupations. Coupled with much of women's work traditionally being unpaid, for example in the home, this casts a new light on gendered differences in employment which require a policy response. There are also other societal issues to address, for instance care work is currently largely feminised, badly paid and often emotionally demanding. On the other hand, there may be potential benefits for traditionally female oriented jobs arising from the **increasing importance of such emotional intelligence** in the Fourth and Fifth Industrial Revolutions, as this is something machines are unlikely to be able to do, at least in the short-term.

Digital divides are directly related to existing socio-economic divides, so that both reinforce each other (Millard 2015b), and the uneven roll-out and use of digital technology exacerbates these inequalities (Oxfam 2016). This is not a given outcome of the technology, but instead derives from the governance, institutional framework, regulatory, labour market and wider economic system. Europe is also struggling to keep up with the digital revolution with a 2019 survey showing that 43% of the adult population in Europe had limited digital skills, and 37% had no digital skills at all. Meanwhile, young refugees and asylum seekers are experiencing a crisis of their own in Europe, finding it very difficult to integrate socially and in the workplace (European Commission, 2019a).

One of the most influential recent studies recognises that everyone has limited cognitive capacity and time but the unique disadvantage of the poor and vulnerable is that they are typically pushed to and beyond these limits more than others. Mullainathan and Shafir (2013) show that the poor in any society have precarious structures within which to live and work. They literally expend all their effort simply surviving from day to day or week to week, and do not have sufficient time, energy or cognitive capacity to plan for and invest in their own, their family's or their community's future. **Being poor is literally a full-time and stressful occupation purely** directed to survival. This empirical research shows that, when richer people are put in the same constrained conditions, they react in the same way as the poor and often a lot worse. When any individual's cognitive capacity is strained in the way experienced by most poor people struggling to cope, it is equivalent to driving a car whilst drunk or reducing their IQ by 14%. Most poor people coping with these conditions are, in fact, performing extremely well just by surviving and are far from being lazy, stupid or scroungers. Poor people, more than others in society, typically have to contend with a much more highly complex and unpredictable social and economic environment. Mullainathan and Shafir (2013) undertook their research in the USA and India, respectively examining relative and absolute poverty but reaching similar conclusions in both regarding cognitive capacity and how individuals behave. As noted in Section 4.1.3, McGarvey (2022) looked at relative poverty in the UK and recognised a "proximity gap between the powerful and the powerless as the root cause of many of society's ills."

Any type of innovation can create 'winners' and 'losers', and public governance needs to balance trade-offs between these and different societal interests. For example, sharing platforms supported by digital technology provide many people with a chance to find the flexible work they want, thereby improving their incomes and social contacts. However, for others, where work in the corresponding sectors is the only source of income, it can force them down a road of low pay and poor working conditions, with unpredictable and thus unstable financial prospects. The jobs most at risk are those which require a medium level of qualification. This has been apparent in wage statistics: salaries for medium qualifications are stagnating or growing more slowly than those of low or high qualifications (Frey and Osborne 2013). The middle class feels neglected by politicians and looks increasingly to the political extremes for help. The questions are: where can the new jobs be created? Can they be created as fast as jobs will be destroyed? And, even if they are created sufficiently rapidly, the transition from one job to another without a loss of income and an increase in structural unemployment is difficult.

6.3.2 The shock of COVID-19

Work and employment have arguably experienced the greatest shock and transformation of any aspect of the economy as a result of COVID-19 which has turned many of the pre-pandemic concerns,

outlined above, on their head. However, these probably remain in the background for the longer-term, especially given that **the labour substitution potential of digital technologies is likely increase even further in future**. These technologies have also dramatically changed, probably for the long-term, where and how people and organisations work. Thisse et al (2022) provide evidence that higher rates of teleworking are part of the post-pandemic new normal – a change with uneven effects across the labour force.

According to Edelmann and Millard (2021), **telework is defined as working away from the traditional workplace using digital technology** and includes all kinds of work-related activities outside the employer's premises, although in different countries there are different teleworking practices based on legal constraints, organisational practices and cultural predispositions (Peters et al, 2016). Telework is understood as representing "location-independent, technology-enabled new ways of working" (Messenger and Gschwind, 2016), that leads to the dissolution of traditional spatial, temporal or organisational boundaries of work (Kunze et al, 2020). The International Labour organisation (ILO, 2021c) describes teleworkers as "employees who use ICT tools to perform their work at home or in another location outside of the employers' premises". Teleworking lends itself particularly to those occupations where work can be carried out autonomously, with little direct supervision and where the workers can be managed on the basis of the results they produce.

Vilhelmson and Thulin (2001) distinguish different types of telework: i) full-time at home; ii) commuting-based work undertaken sometimes but not always at home; iii) mobile work performed at a variety of locations other than home or the ordinary place of work so that the worker is able to flexibly move around while still being in real-time communication and able to access the facilities they need, and iv) work in telecentres (shared work-spaces) as alternative, normally suburban-based, locations closer to where the teleworker lives. Møller-Jensen et al, 2008) showed that the two most important factors enabling telework are technology and the way work is organised. Educational level is also important, as is income. In addition, size of residence and availability of a dedicated room are significant, aa are differences in telework take-up with men more likely than women to do so, older people up to 50 years more likely than younger people, and families with two adults and children at home more likely than other household types facilitating a better balance between work and family.

Examining the incidence of telework before the 2008 financial crisis and the 2020 pandemic, the ECaTT survey (2000) across all EU10 countries found that 4.5% of the labour force teleworked with over one third of these engaged in mobile telework many of whom would be what are termed today 'gig' economy workers. By 2009, this had risen to 7.5% of the EU-27's employed population, and then up to 11% in 2019, although this figure jumps to 15% if those "who had ever teleworked" from home are included (JRC, 2020). Other data from ILO (2021b) provides 2015 data showing that before COVID-19 13.5% of employees were occasional home teleworkers with 4.8% doing so regularly, and a further 7.3% as mobile teleworkers. Very substantial jumps in all these figures occur during COVID-19, suggesting that **close to 40% of those currently working in the EU began to telework full-time from home as a result of the pandemic** (JRC, 2020). The ILO (2021b) calls this a 'telework tsunami' and provides data that show 34% of workers across the EU-27 exclusively teleworking from home during the pandemic with another 14% doing so partially, making 48% in total.

According to Palomino et al, (2020), the capacity of individuals to work under a lockdown relates to their teleworking capacity given that the pandemic had direct asymmetric effects on the labour market. Only the jobs that can be done from home ('teleworkable') are unimpeded by the lockdown. Some occupations like health care services and food are considered essential so workers are not affected by their capacity to work from home, whilst other economic activities like hospitality are closed under a lockdown and working is not at all possible. In addition, there may also be groups of employees who are less able to benefit from telework, such as those with low levels of education or workers with low skills and other vulnerable populations (Gallonier, 2020; ILO, 2021b; UNDP, 2020). Thus, the type and relative importance of work, together with the availability of good digital infrastructures, determine telework capacity and, on this basis, is highest in northern and western Europe, although the increases during COVID-19 were greatest in southern and eastern Europe as these countries were starting from a much lower base.

The data also show an almost complete shift to full-time teleworking at home during the pandemic, with almost no mobile teleworking by employees who normally have several workplaces. However, there is an increase in exclusively mobile telework mainly by so-called 'gig economy' workers in response to huge demand increases for the home delivery of goods. Considering that before the outbreak, just 15% of the employed population in the EU had ever teleworked according to the JRC (2020), large numbers of workers and employers alike currently face huge challenges in dealing with the sudden shift to telework. In many EU countries, more than half of the workers who have started working from home since the pandemic had no prior experience with teleworking. If past trends are a guide, the uneven ability to scale up telework could result in widening inequalities across countries, firms, and workers. Thus, with COVID-19, telework perforce became the normal way of working for much greater numbers of people, However, the impact of telework on work performance and productivity, although much more intense, is in general very similar to findings from 20 years earlier when in its infancy (Edelmann and Millard, 2021). The Global Workplace Analytics (2021) report that **two to three days a week seems to be the 'sweet spot' that allows for a balance of concentrative work (at home) and collaborative work (at the office)**, whilst much earlier Millard (2001) found exactly the same balance. For many, the balance of benefits and disadvantages is in general dependent on the amount of telework. Worker efficiency improves with low levels of telework reaching a 'sweet-spot' where productivity is maximised at intermediate levels. However, efficiency decreases with 'excessive telework' due to negative effects on communication, knowledge flows and managerial oversight, on increasing worker solitude and fusing private and professional life (OECD, 2020e). Specifically, **tacit and emotional knowledge that is critical in many occupations, can be lost if all interaction is digital**.

There is also a similarity between early telework impacts on the environment and those post COVID-19. Paradoxically, teleworkers in an early 2000s' study tended to drive greater distances than non-teleworkers, and thus have a larger environmental footprint. This was because the technology enabled them to be mobile whilst connected and thus able to visit customers, partners and other workplaces much more freely. Non-teleworkers, on the other hand, working in their normal single fixed workplace, often as commuters, were both more likely to regularly use public transport as well as stay in one place whilst working (Millard, 2001). In 2022, study results indicate that Covid-19 could cause a similar rebound effect reversing the supposed positive environmental impacts of working from home, since, even if the number of trips could be reduced, many shifts towards non-sustainable travel modes could occur. The promotion of telework should be combined with measures fostering sustainable travel habits to pave the way towards a future green mobility (Ceccato et al, 2022).

Other similarities between the pre- and post-COVID-19 situation which have public governance implications relate to **telework benefits** for many, such as better compatibility between work and family, more flexibility, more recreation and leisure time, saving costs on commuting, and increased performance and employee satisfaction. Teleworking seems to significantly improve the guality of life of many employees, mainly through stress reduction, and can be seen positively by employees because it "makes them more serene and concentrated" (Gallonier, 2020). Teleworking, which is today more commonly referred to as 'hybrid work' when it is not undertaken full-time, is often seen positively by employees because it can significantly improve life quality and relieve them of sources of stress (Edelmann and Millard, 2021). Because of COVID-19, the number of employees who would like to continue to work in hybrid mode is on the rise, so pressure is on organisations to respond to these demands (Kunze et al, 2020). Identified advantages for organisations are also similar before and during COVID-19, although again more intense, including greater time and cost savings given a reduction in the number of business trips and external meetings, increased employee satisfaction and lovalty, and a larger labour market through the recruitment of staff living further away. Many employers now need smaller offices for hot-desking in less central locations, thus significantly reducing the cost of premises. For these and other reasons, organisations are again showing willingness to experiment with hybrid work to achieve cost savings, with more flexible work schedules, appropriate work tasks and providing adequate equipment (Lavelle, 2020).

Also as documented since the mid-1990s, there are **challenges and difficulties associated with telework** that need to be addressed by organisations, but again these are often now much more intense than previously (Edelmann and Millard, 2021). For example, employees seem not always able to switch off from work, collaboration and communication with colleagues and co-workers can be difficult when not physically together, employees find it hard to separate work and private life, poorly equipped home workplaces may lead to health impairments and psychological problems, and organisations become highly dependent on technical equipment which does not always function well. In addition, there may also be groups of employees who are less able to benefit from telework, such as those with low levels of education or workers with low skills and other vulnerable populations.

It is already clear that the COVID-19 pandemic is leading to a strong upward acceleration in the digitalisation of society (UNDP, 2020), and that hybrid work will become a normal way of working (Lavelle, 2020). Digital transformation can be understood as the organisations' ability to leverage digital technologies in order to respond rapidly to stimuli (LI, 2020), but also to transformation of work and employment, providing the opportunity for organisations to rapidly change and digitally transform themselves (Seetharaman, 2020). Comparing how telework has changed pre-, during and post-COVID-19 telework shows the following (Edelmann and Millard 2021):

- The Pre-COVID-19 era of telework was technology-driven characterised by a mainly stable labour market but with increasing tech-based and 'gig'-economy work. Organisational change was slow, evolutionary, piecemeal top-down, with incremental innovation and internal and external structural adjustments based on largely predictable responses. During this period, telework was largely voluntary and selective, thus has a low uptake from an average of about 5% in the EU early years to about 11% in 2019. Telework was mainly home-based, though with significant mobile telework (between 1/4 and 1/3 of the total), and its uptake was mainly in highly advantaged demographics and highly specialised job types with others largely left out. There were small scale experimental moves to suburban and rural 'telecentres' and the use of 'hot-desking' in workplaces as some work became more mobile, while commuting and some agglomeration in cities of all sizes largely continued. For public sector organisational change was slow, pre-meditated, careful, and telework uptake lower than elsewhere, although some catching up with a greater distinction between front- and back-office staff was also seen (Edelmann et al, 2021).
- **The COVID-19 era** of telework was shock-driven, enabled by technology and extreme disruptive innovation. The measures imposed to prevent the spread of COVID meant that very fast, systemic and uncoordinated bottom-up and top-down organisational change was necessary with internal and external structural tipping points based on largely risk reactive responses. The labour market was highly unstable, with up to 40% of workers either unemployed or furloughed. Organisations responded to the pandemic by imposing telework, so that up to half of the working population worked from home full-time. The uptake of telework was seen across most demographics and job types, but it was mainly the highly advantaged who enjoyed the benefits, whilst the disadvantaged often only experienced disadvantages. There were major shutdowns of city and suburban centres and plummeting city centre real-estate values as many commuters were largely confined at home. (See also Section 4.2.2.) Organisations in the public sector also responded by imposing telework on employees, but there was a greater distinction between front- and back-office staff. In terms of organisational change, they became more flexible, reported that digital tools support communication, transparency, faster decision-making and that targets can be achieved just as well as in the office. As more people gained first-hand experiences of telework and for longer periods of time more intense positive and negative effects were noted. The first signs of digital culture, hybrid teams, digital collaboration were noted, but there was a clear need to adapt regulations and frameworks. The public sector labour market remained relatively stable but grew to some extent as 'bigger' government and many public sector jobs and tasks became recognised as 'essential' (Edelmann et al. 2021).
- **The post-COVID-19 era** of hybrid work can be postulated as exhibiting a combination of multiple societal and technology drivers and enablers as well as ubiguitous innovation. As argued by Perez (2010), radical change is most likely to result from a synergy between technological and organisational innovations, thus organisational change will be continuous systemic, radical and multi-directional, with internal and external structural change based on resilience-seeking responses. The overwhelming majority of the workforce will engage in different combinations of hybrid work which will become the working norm and the term 'telework' will have little meaning. Hybrid work will be part of almost all organisational and digital strategies. The re-shoring of economic activities that started during COVID-19 will accelerate as will the EU's 'open strategic autonomy' strategy. There will be much more spatial mixing of work and non-work activity as both can be combined in situ, with a renewed focus on co-working centres and hot-desking away from very large cities. Deglomeration, that was already apparent before the pandemic, will be significantly strengthened but only in terms of movements of populations from the very large cities (often with the exception of capitals) to smaller cities just beyond the suburbs (which are being termed 'zoom towns'). coupled with some continued rural depopulation but at smaller pace than before. (See also Section 4.2.2) Public sector organisations, in order to be creative and innovative (George et al, 2020) and able to combine experimentation and innovation with continuity, stability and predictability, will need to develop frameworks and strategies that support the implementation of hybrid work. They will also, in particular, need to re-think management, corporate culture and organisational strategy. As organisations move towards widespread systems thinking, public sector bodies will also see digital collaboration as the norm, making them modern employers, and attractive to the labour market (Edelmann et al, 2021).

Thus, hybrid work uptake is likely to expand across all demographics and job types, but with **the remaining threat of two classes of workers: the advantaged enjoying the benefits, with a large minority of the most disadvantaged who experience mainly disadvantages**. Many of the latter are low paid and insecure workers, including in 'gig' economy jobs, typically obliged to work in poor, crowded and often unhealthy conditions prone to COVID-19 infection. These were precisely those that many employers and governments were applauding as 'essential workers' keeping the economy and society functioning during the pandemic. For example, people working in the health and care sectors, in transport and maintenance, cleaners and caterers, in retail, food and agricultural workers, as well as drivers and delivery workers (Fleming, 2020; Palomino et al, 2020). This makes sense as these jobs constitute the '**foundation economy**', i.e. the part of the economy that creates and distributes goods and service that are necessary for everyday life. For most of these workers, physical presence is essential so there is less hybridity and they often have little flexibility over the location, timing and extent of their work. In comparison is the class of 'privileged' workers in 'privileged' sectors – finance, business services, consultancy, academia, media and much middle and upper management office work – who can insist on working from home part of the week, with Friday and Monday especially favoured to create a long weekend.

For example, Thisse et al (2022) show that the shift to teleworking in the UK has not been evenly realised across the labour force with a dramatic 400% increase in teleworking for skilled workers, but much more modest increases for unskilled workers. Because skilled workers now spend less time in the centre of very large cities, they consume fewer of the local services provided by unskilled workers. This leads to a destruction of jobs in the local services industry, forcing unskilled workers to move into the manufacturing sector. This decreases wages for unskilled workers but increases wages for skilled workers. Overall, according to Thisse et al (2022), the consequences of tele- and hybrid-working go way beyond productivity gains or losses for firms and workers. **The increase in teleworking is exacerbating urban wage inequalities, the social and physical structures of cities and the cost of housing**. These changes are likely to exacerbate urban inequalities and have a disproportionately negative effect on workers providing local services. Inter-city and long-distance commuting, i.e. not from the suburbs but from smaller cities into the central areas of very large cities coupled with teleworking, is having an impact on cities' finances. Fewer commuters – or workers who commute less often – could significantly decrease the demand for office space and translate into a shrinking local revenue base for these very large cities.

Driven by many of these trends, **work, employment and organisations of all types are changing dramatically across all sectors**. In 2019 the EU unemployment rate was 6.69%, but this rose to 7.05% in 2020 at the height of the pandemic because of the economic slowdown. Governments across Europe implemented furlough schemes that helped to prevent widespread job losses. At the peak of the pandemic in France, Germany, Italy, Spain and the UK, the programmes covered about a third of the workforce and these were not included in unemployment data (EIT, 2021). Now that most furlough schemes are over and COVID-19 restrictions largely lifted, Europe's companies are facing rising costs largely because of the war in Ukraine, as well as the lack of skilled workers which is becoming more acute than ever before. Around 193 million Europeans were pursuing a paid job in the first quarter of 2022, while some 74.5% of those aged 20-64 were in some sort of formal employment – the highest employment rate since Eurostat started publishing it in 2009. The most recent data suggests that the figure has risen further in the second quarter of 2022. At the same time, more than 3% of all available jobs are vacant – more than ever since statistical records of this data started in 2006 – meaning that around six million jobs are not filled across the EU. Correspondingly, the EU's unemployment rate sank to 6.0% in July 2022 – another record low since at least 2001, when Eurostat began compiling this data (EURACTIV, 2022b).

One reason for this combination of untypically low unemployment and high labour shortage is the so-called 'great resignation' or 'great reshuffle'. In mid 2022, many workers are thinking about quitting their job in both Europe and the United States. Inflation and the cost-of-living crisis are fuelling people's desire to quit work altogether if they have enough savings or are anyway close to retirement, or find a better paid job. Over four million American workers have quit their jobs since the post-pandemic recovery started in 2021, and this was mirrored on a smaller scale in countries in Europe. In France, the number of resignations reached a record peak in the third quarter of 2021, the highest since 2007. In the UK, the rate of people moving from job to job was at an all time high between October and December 2021. While this was initially ascribed to a cultural change in the way workers approached their work-life balance following the pandemic and increased burnout, workers' motivations seem to have fundamentally changed in reaction to ongoing economic crises. According to a recent survey, one in five respondents said they are likely to change employers in the next 12 months, with the majority of those seeking to switch jobs for a better salary. More than a third plan to ask their employer for a pay rise, whilst the percentage was even higher in the tech sector, where this applied to 44% of workers (Euronews, 2022).

Views about new hybrid work are mixed but most are relatively positive if it is well organised According to WEF (2022e), after the 'great resignation' comes the '**great renegotiation**' with hybrid work here to stay. More than four out of five survey respondents who worked in hybrid models over the past two years prefer retaining them. When organisations are plagued by burnout, mental-health issues, and record numbers of employees leaving their jobs, leaders who see in-person work as a return to normality must confront just how strongly employees appreciate flexible work models. More than two out of three employees who prefer hybrid models say they are likely to look for other opportunities if asked to return fully on-site. **But hybrid work has the potential to create an unequal playing field if not done well**. Employers must prioritise inclusion practices: work-life support, team building, and mutual respect. Marginalised groups are more likely to prefer a hybrid work model and are more likely to leave if not available (WEF, 2022e). A hybrid working model is also critical to attract a diverse workforce, e.g. working parents and women who have primary family care responsibilities, people with physical or mental health conditions, and those with housing problems due to the rising cost of living near their office (WEF, 2022f).

However, **there can also be inequitable outcomes if hybrid work is not implemented well**. For young people at the start of their working lives, not being in an office will probably entail two kinds of disadvantage being cut adrift from the collective workplace experiences that allow people to find their professional feet, and not having the domestic space to do their work effectively (Harris, 2022). There is also clear evidence of how traditional gender roles affect home working. In American research, 79% of men said they experienced "positive work effectiveness" at home, compared with only 37% of women. Whoever you are, moreover, there is a good chance that working from home will have increased your working hours. Research during the first global lockdown found that for 3 million remote workers around the world, the average working day had increased by 8.2%, or nearly 50 minutes (McKinsey, 2020). Rather than fixating purely on home working, **policy should focus on flexibility with significant worker control and as agreed with the employer**. Irrespective of where they work, it is important for people to be able to start and finish at times of their own choosing, carve out free time and ensure holidays complement the other aspects of their lives. For all employees, there ought to be both an entitlement to collective representation, and some sort of right to disconnect – to not have to deal with emails, calls and messages outside working hours. This has been adopted in France, Italy and Spain, and is now tentatively supported by the Scottish government for public sector staff (Harris, 2022).

In parallel with these developments, **'social jobs' are also likely to increase** given the historic underinvestment in social infrastructures leading to a widespread erosion of social mobility. For those born into lowincome families, prospects look bleaker among today's younger generations than in previous years. There is thus great benefit in focusing on **three foundational social institutions – education, healthcare and care** – that can benefit from key investments to re-start the engine of social mobility across economies, filling unmet demand for healthcare and childcare, as well as increasing the quality of education (WEF, 2022d). New distributed work patterns leading to greater spatial mixing of work and non-work activities have the potential to boost the local economy of suburban hubs and smaller cities. Assuming local governments can provide support, for example in the context of community-wealth building' initiatives, local services, 'artisan-based' and sharing and collaborative economy jobs could grow as part of the increasing 'new localism' trends. (See also Section 4.2.3.)

6.3.3 Work and employment examples

In addition to the potential of expanding these 'social jobs', other examples of new growing, often local, job creation activities are **social enterprises, employee-owned enterprises, 'worker cooperatives' and 'mutuals'**. (See also Section 7.3.2.7 for many more examples.) These are generally owned by their members who share the profits, are typically democratically controlled, have a small gap between pay levels, and have a wide social purpose. Most are local or regional and can take many forms, ranging from simple grants of shares to highly structured plans. Employee ownership in the US narrows gender and racial wealth gaps and gives employees significantly more wealth than the average worker. Employee-owned companies perform better commercially, but are resisted by banks, lawyers and governments (Rutgers, 2019). Countries with many employee cooperative workers rank low on inequality, high on happiness and high on the UN Social Progress Index (SPI) measuring "accessibility of basic needs, opportunity, and access to knowledge" e.g. New Zealand, France, Norway, Sweden, Denmark and Finland (ILO, 2021a). (See also Section 4.2.3.)

An example is the regional worker cooperative is the Mondragon Co-operative in the Basque Region of Spain with a mission encapsulated in its corporate values: inter-cooperation, grassroots management, corporate social responsibility, innovation, democratic organisation, education and social transformation, etc. Mondragon's activities are divided into four employment areas: finance, industry, retail and knowledge. It currently consists of 95 separate, self-governing cooperatives, around 80,000 people and 14 R&D centres, occupying first place in the Basque business ranking and tenth in Spain. Mondragon is overall highly successful in living up to its principles and competing regionally, nationally and internationally. It has been successful in creating a democratic, sustainable, and secure living environment for its members and, although it may not be the perfect

utopia many envision (e.g. there is a lack of emphasis on environmental impacts), there is room for growth and change (Tang, 2019). Its governance strikes an important balance between the need for democratic decision-making and managerial discretion, otherwise the critical difference to a private corporation is that the purpose of the firm is to benefit its members rather than its shareholders (Bamburg, 2017).

At the national level, **the Scandinavian 'flexi-curity' model** provides examples of labour market governance and a way to address the 'conditionality' problem for employment support that successfully increases the benefits for all labour market partners. The focus shifts from having and keeping a particular job, to being employable and relatively easily able to obtain a good job if the existing one is lost. On the one hand, this gives significant flexibility for employers by allowing them to hire and fire relatively easily. On the other hand, it also provides good security for employees by strong labour market support through relatively high unemployment and re-training benefits, as well as proactive interventions from government, employers and trades unions working together. Current social policy experimentation is also addressing the issue of 'conditionality'. For example, in the labour market context to obtain unemployment benefit, the unemployed need to prove they are actively seeking work, thereby requiring significant form filling, costly administration and time. The debate about the effectiveness of conditionality feeds into the debate about the value and effectiveness of the so-called 'universal basic income' (UBI) as an unconditional payment, sufficient for a very basic but still comfortable living, made to all adults regardless of whom they are or their labour market status. Results in 2020 from a two-year Finnish study show that UBI under certain conditions can improve the mental wellbeing, confidence life satisfaction and the civil engagement of citizens, as well as have a mild positive effect on employment by those of working age who were not initially working (Millard, 2020; Henley, 2020; Kangas, et al, 2021; Hasdell, 2020).

Examples of **hybrid digitally-supported work** based on the improved productivity that can be achieved, there are also examples of employers reducing the length of the working week. The CEO of WANdisco, a data acquisition company, reports that a move to a largely home-based four-day working week for employees during the pandemic had already allowed much higher productivity. This means that the work that was once being done in five days was now easily covered and it is expected that even greater levels of productivity and a boost to staff wellbeing will materialise in future. While Friday will be the default day off, workers will have the flexibility to choose an alternative day. Employees' salaries are not affected by the changes. Most company staff are now continuing to work from home with zero staff attrition rates, and this will continue (Financial Times, 2020).

Also fully enabled by digital technology, the **freelancing platform**, Upwork (Upwork, 2002), has the slogan "how work should work -- forget the old rules. You can have the best people, right now, right here." The platform's aim is to pioneer "a better way of working, help businesses find more flexibility and connect talent with more opportunities." Upwork is one of the most popular freelance marketplaces that connects millions of employers and independent contractors from all over the world. But there are both pros and cons with such platforms that have governance implications. For example, there is a clash between US freelancers and those from countries with much lower costs of living, giving rise to two major problems: the insurmountable entry barrier for new freelancers and dumping. This has led to freelancers with higher professional skills to leave the platform as they do not want to compete with the less qualified contractors who can win work by bidding much lower rates, which has caused an overall decrease of the Upwork job success rate and quality. Upwork reviews are filled with the unfortunate stories written by former customers who have been scammed, due to the fact that freelancers only have to go through basic verifications which don't include any background checks or vetting procedures. This leaves every employer questioning "Is Upwork safe, or are there better options?" (Lemon, 2022).

There has also been a rise in so-called '**digital nomads**' as remote workers, including self-employed individuals, freelancers, and employees, who may travel domestically or overseas, with the internet keeping them connected to jobs, co-workers, and clients. According to Forbes (2022), the digital nomad lifestyle has been on the rise for years, but once COVOD-19 hit, its popularity exploded. In 2021 the number of digital nomads in the U.S. was 15.5 million, and by 2025 there could be 35.7 million Americans or 22% of the workforce working as remote workers. Although some digital nomads work illegally for companies in other countries without travelling there and getting an official work permit, many countries now offer digital nomad visas so registered individuals can do so legally. In Europe, these include Portugal, Croatia, the Czech Republic, Estonia, Germany, Hungary, Greece, Iceland, Italy, Malta, Romania, Spain, and Norway. Such visas grant residency rights to foreigners to work for out-of-country employers for, on average six months to two years. For example, the Portugal Digital Nomad Visa enables non-EU and non-EEA remote workers who are employed by foreign companies based outside of Portugal to relocate and start working remotely whilst residing in Portugal, provided they make four times the national minimum wage (Global Citizen Solutions, 2022). For the countries and employers concerned, the benefit

is attracting extra tax revenue and potentially particular talent and expertise with limited bureaucracy. For the digital nomads they can reside in a country without paying normal tax, for example in Portugal no taxes on foreign income, 20% tax on income earned in Portugal compared to standard Portuguese income tax rates of up to 48%, and social contributions 10% which is considerably less than the standard rate. There are also moves at the EU level to improve **social security and coordination regulations for cross-border teleworkers**, digital nomads, highly mobile and posted workers (European Commission (2022f). However, there can be downsides with digital nomadism, for example those earning North European wages and living in the cheaper south can drive out locals who can no longer afford to live in areas favoured by 'anywhere workers'.

Several other important governance implications of hybrid and digital nomad work include **skyrocketing cyber risk** due to the rapid shift to remote work during the pandemic and the increased automation of business processes. While it may have been evident that home-working environments wouldn't necessarily replicate the more sophisticated security of an office environment, it wasn't clear that more individuals would be susceptible to phishing attacks while working away from offices. Nor did many predict the level of increase in cybercriminal activity. Responsibilities in relation to remote and flexible working conditions are also much less clear. Even with standards for remote working environments, employers will have less control over the myriad different set-ups, creating a higher likelihood for more work-related injuries and greater workers' compensation claims. Also, **working anywhere could increase risk of non-compliance with labour and tax laws**. Gaps in insurance and benefits coverage for both employers and employees could also increase, particularly without well-defined protocols, controls and management resources (WEF, 2022g).

7 Rethinking European public governance to meet new strategic challenges

Together with the role of digital technology in Section 5, this Section 7 focuses on the **good governance public values likely to be needed to successfully contribute to addressing the strategic challenges** in Section 4 and the examples provided in Section 6. However, Section 7 does not purport to be exhaustive, definitive or fully cohesive. Rather, it is offered as work in progress in exploring good governance values going forward that are additional to those identified for the period 1945-2019 in Table 3 (Column C) in Section 3.3.

7.1 Societal governance of human society enabled by nature's life-support systems

The whole point of public governance is that it is the fair, just and balanced governance of the whole of human society as enabled by nature's life-support systems. Recasting 'public' governance as 'societal' governance that is dependent upon nature thus might be a useful way more broadly and usefully to change the governance mindset needed to meet the new strategic challenges, some of which are suggested in Section 4. Such a mindset nudge has long been presaged, for example by Koolman (2003) who examined the question of different modes and orders of governance as patterns of societal governance. While 'Modern Governance' was still strongly 'government-oriented', there is benefit in broadening the perspective so that it looks at governance as societal, with public as well as non-public 'governors' participating. "While their roles may differ between societal levels and from sector to sector, the essence of the argument is that **governance of modern societies is a mix of all kinds of governing activities and structures**....These mixes can be seen as 'answers' of those societies to changing governing demands...in which public as well as private actors aim at solving societal problems or create societal opportunities, aim at the care for the societal institutions within which these governing activities take place, and phrasing the principles according to which these activities are carried out." (Koolman, 2003)

Indeed, it is the case that it is only the government (i.e. the state) that has the mandate and powers to take such a societal level perspective, especially over at least the medium-term where trade-offs and interactions are present between actors. However, although the government has much of the knowledge and many of the assets to take the lead in societal governance, it certainly does not have a monopoly, especially in particular contexts, as the other quadruple helix actors also have significant responsibility and ability to create their own specific value (see Figure 5). For example, a 'lean' governance paradigm (see Section 3.1) might indeed save some money in a narrow context over the short-term, but this could lead to overall loss of public value and thus additional costs on society in the longer term, if other actors are not able to produce the value needed in the context of a hamstrung public sector. Examples include environmental degradation, social and economic inequalities and in mainstream services like health, care and education, and these would be false economies indeed.



Figure 5: Changing models of delivering public services and public value: from siloed to an 'ecosystem' approach



Figure 5 also illustrates the need to move away from the traditional prevailing mode of governance, characterised by siloed roles and value creation with limited synergy between quadruple helix actors, on the left. Instead, we need to adopt a new and emerging ecosystem of joined-up, collaborative and co-creative quintuple helix actors that can optimise the synergy between them to maximise societal value, on the right It is clear that in some circumstances some specific actors and even large groups of actors will find some of their 'value' reduced or compromised, at least in the short term, because of government activity aiming to maximise societal value overall. Such value 'loss' by one actor will typically be counterbalanced by value 'gain' accruing to a different actor, and often takes place when societal inequality is relatively large so there is both more to lose and more to gain. Indeed, **it is one of the prime tasks of the government, which no other actor has, to minimise, ameliorate and/or compensate for such value loss, if possible and desirable, by balancing trade-offs in the wider societal interest. Change often disadvantages some and advantages other actors when inequality is significant, but there are also many examples of win-win-win situations especially when inequality is low at the outset. In all cases, however, government needs to distinguish between the unfair 'vested' interests of one group against the fair benefits of another group, normally by adopting a societal value perspective. (Millard, 2015a)**

In this context, there at least four types of role and relationship that the government needs to exercise, only made viable through the application of digital data and technology:

- Facilitate and orchestrate: arbitrate, coordinate, regulate, mediate, partner, support, etc., noting that government is not just another actor as it does have a special set of roles, and often needs to lead because of its democratic mandate. It takes account of and balances all interests in society, it cannot choose its users, it needs to ensure the quality of provisions and decisions, and it may need to be the supplier of last resort.
- Provide tools for collaboration and co-creation: government is often seen as 'outsourcing' its responsibilities to users, e.g. through online 'self-services', so it needs to mitigate the 'burden' on users to (co) create, and to provide users with guidelines, incentives, supports, advice, networks, ecosystems, etc, that themselves do not add to the administrative burden.
- Manage assets: Identify legitimate and available assets across society and help to orchestrate and deploy these (often in collaboration with asset owners) to create societal value. Unused assets may be considered as 'wasted' assets when their use does not damage social, economic and/or environmental sustainability.
- Ensure sustainable and balanced societal value. Only government can ensure this where all segments of society benefit and where trade-offs are seen as fair and proportionate. Only government provides longer term stability and continuity which other actors cannot do, and this is necessary to enable people and communities to live stable lives, as well as for the market to have confidence that unpredictable governance changes will not upset their own innovation and investment decisions.

Important issues arise from such '**balanced societal value**', given that, although government does not have a monopoly on societal value creation, it does have the prime role in ensuring that societal value is created, and that the relevant public services are reaching those that actually need them. All actors are potentially seen as a resource with knowledge and assets to contribute in creating societal value but their interests may need to be balanced, or traded-off, where these are incompatible with others, and this is primarily the role of government, both in the short and long term. Thus, there is a societal governance requirement not to sacrifice the long- for the short-term, and to be transparent about this perhaps with the help of 'quick wins' exploiting 'low-hanging fruit' in the short-term providing these do not damage the long-term. There are also a number of risks when operationalising these roles and responsibilities, including:

- Loss of control and blurred accountability, for example, of services: Who is providing a public service on a platform? Who owns this service? Who is responsible and accountable for the service and for providing the platform? Who benefits from the service? Does it reach those in need? etc.
- Quality standards more difficult to determine and maintain with many co-creative active designers and suppliers of services. Who defines the quality standards for the services and for the platforms? Who needs to assess the quality? When does the quality of a service or platform have to be (re-)assessed? Who is responsible to react in case of a service or platform not meeting the quality criteria? How? etc.
- Putting too much faith in using open government data and big data generally: How representative is a given data set? What are the known and the potential biases? How to avoid misuse through lack of competence, deliberate misuse (i.e. corruption), etc.
- Need a common sense test for algorithm-driven decisions and policies How to avoid the 'black box' syndrome. How to assess the transparency of algorithms and their use? How to ensure ethical use of existing algorithms? etc.
- Put 'hard data' in the context of 'soft data' like values, ethics and responsibility, necessary to build trust through transparency. How to set 'hard data' into cultural context? Who to communicate about the findings and embedded uncertainties? How to take account of reactions from citizens? etc.

There is a need to **step up the focus on experimentation** in this new age of turbulence, and to **be more accepting of 'failure'**. However, 'failure' is only valuable when, it is small and early, not late and big as happened to the financial system in 2008, and when there are built in innovation, learning and experimental systems (see Section 7.3.2 below). It is also important that the process of experimentation uses sound scientific approaches (see Mair et al, 2019) and is legal, ethical and non-exploitative. Good examples of such experimentation for new forms of governance are found in the International Research Society for Public Management (2022).

7.2 Re-orientating public value and public values from 2020 onward

As defined earlier in this document in Section 2.3, the following distinction between 'public values' and 'public value' is made:

- The **public values of good governance** (taking the definition used by the European Commission, 2017) refer to how public value is created thus public values can be seen as the 'means' of creating public value. This Section 7, together with the role of digital technology in Section 5, attempts to sketch the main public values likely to be needed to successfully contribute to addressing the strategic challenges in Section 4.
- Public value (taking the definition proposed by Kelly et al, 2002) refers to the actual benefits derived from public governance that accrue to all quintuple helix actors thus public value can be seen as the 'ends' or the purposes of public governance. If and when the strategic challenges in Section 4 are successfully addressed, this will create new types of public value, i.e. real benefits for society.

Table 3 in Section 3.3 enumerates both the cumulative public values and public value that the selected public governance paradigms appear to have delivered up until 2019. All nine of the European Commission (2017) good governance values summarised in Table 3, column C, were achieved to some extent, although further work will always be needed. In addition, **it is proposed to add open governance, sustainability governance, and locality/community governance as public values where at least partial progress has been made up to 2019**. From 2020 onwards new types of good governance public values will be needed to be built on top of those already identified and as summarised in this Section 7.

In terms of public value benefits, these are summarised in Table 3, column D, for the period up to 2019. According to Kelly et al (2002), these are mainly at two levels, i.e. first, "services that provide the vehicle for delivering public value through actual service encounters for users or clients and the distribution of fairness, equity and associated values for citizens". The second level is "outcomes that commonly overlap with services but should be considered separately as they encompass much higher order aspirations such as national security, poverty reduction, public health, etc.", to which different combinations of services collectively contribute depending on the political and policy context. (Specific public service examples are also given in Table 2.) From 2020 onwards, public value benefits created will be derived directly from any success in addressing the strategic challenges identified in Section 4, i.e. to be able to live with turbulence and to maximise the benefits and minimise the downsides of changing geographies, of hybrid human-centred relationships and of people-planet systems. Much more evidence on the period up to 2019 is provided in the accompanying the state-of-art report

7.3 A new European societal contract for delivering public value

As the basis for a **European social contract** – or perhaps better described as a **societal contract** as it needs to go far beyond only the 'social' sphere – for delivering public value through good governance values, there has been a cumulative development in European public sector reform. The following two sub-sections examine the pre- and post-pandemic situation.

7.3.1 Good governance public values prior to 2020

In 2019, Vesnic Alujevic and Scapolo (2019) outlined the contours of what they termed a new social contract by exploring a rethink according to the needs of today's society, what elements need to be adjusted to deliver value and good to people and society, what values are needed to improve society, and how a new sense of responsibility can be obtained. Their report identified nine policy options and actions that should form the basis for this new contract that are here further elaborated including from other pre-2020 sources:

7.3.1.1 Democracy and power relations

There is a need to diversify power structures, create clear strategies towards full adoption of open government and deploy digital governance. The European Commission encapsulated its approach to open government in 2013 with its 'vision for public services' (European Commission, 2013b). This was followed in 2018 through an 'open governance framework' (European Commission, 2018), resting on the rule of law, consisting of open data, open service and open process where each of these three interlocking components is open by default:

- Open data, both from government itself as well as other actors where appropriate (e.g. crowd-sourced data), suitably aggregated so individuals and organisations cannot be identified and making this readily available in machine-readable formats, is essential for facilitating collaboration, co-creation and policy making. A barrier is that for many users this requires new capacities, skills and incentives, so government needs to provide much more support and many more incentives.
- Open service, through co-creation with service users, for example to design and deliver highly personalised services rather than one-size-fits-all common services. The use of alerts, invitations, prompts, as well as typical life events, user profiles and locations, are all steps towards full personalisation.
- **Open process** ensures that all legitimate actors are able to participate in the policies, decisions and arrangements of the public sector as long as this participation is itself open and enhances public value.

7.3.1.2 Participatory culture and deliberation

There is a need to include citizens in informed deliberation and decision-making, use citizen engagement to cocreate solutions not just react to them, and build a better skilled and equipped public administration with appropriate resources. Gouillart and Hallett (2015) outlined a series of five steps for co-creation in government in order to open up the public sector value chain to citizens, frontline employees, and other stakeholders:

- Identify target communities. Select communities whose members will take part in that effort.
- **Build engagement platforms**. Provide targeted communities with physical spaces, virtual platforms or a combination of both, where community members can engage with each other.
- **Foster interactions among stakeholders**. Participants use the new engagement platform (or platforms) to enable new kinds of relationships.

- **Enable new experiences**. Ensure that new interactions lead to valuable experiences for all stakeholders –experiences that intrinsically improve the quality of their lives.
- Assess new value. Verify that the sponsoring organisation has generated new value-measurable economic value, in particular-as a result of its effort. (The idea here is that an organisation should be able to compute a return on the investment made in its co-creation project.)

7.3.1.3 Political trust

There is a need to develop new participatory governance mechanisms along with publicly funded EU public service media and contribute to raising the trust of citizens through better inclusion and the creation of the European public sphere. Basic to political trust is that the government is trustworthy which ultimately derives from an **ethical culture, transparency and accountability**. Transparency involves the right and access to information, for example in relation to budgets and procurement; and accountability refers to checks and balances such as answerability, responsiveness and integrity. Transparency also leads to accountability through mechanisms and tools that make it clear what a government should do and how well it is doing it at all levels. Both can be promoted by 'freedom of information' legislation that also allows questions to be asked and responses provided within a specified time period. Although such freedoms can be mis-used, they contribute directly to trust in government as well as to better governance in the longer term. There is a need to achieve a balance between transparency and privacy not just for users but also for politicians and civil servants who need adequate private space for decision-brainstorming before decisions once made are publicised together with full transparent disclosure of the decision logic and the sources used in making the decision.

7.3.1.4 Regulation

There is a need to develop technology regulation based on European values with a supranational approach, to include human rights that also reflect the present and future and to avoid the digital divide between citizens who can use technology and those who don't. Regulation is also necessary in terms of technology standards, service levels and quality and to ensure level playing fields that are clear, stable and predictable in order to promote investment and job creation.

7.3.1.5 Public-private relationship

It is necessary to increase collaboration between the public and the private sectors to face both future technology and public service challenges and to open more interactive spaces for the exchange of ideas and expertise. In most countries the public sector is normally the biggest and most powerful actor, but does not have a monopoly on resources, the ability to innovate or on delivering public services, although it does often have the prime role in ensuring that such services are made available. Civil society and the commercial sector also possess increased capacities, tools and willingness to participate alongside the public sector in addressing societal challenges ranging from the local to the global Public service outcomes are thus determined by the interplay between decision makers in the public sector, often cooperating with civil society and private companies, as well as closely listening to, and co-creating with, users. Such collaboration rests on the notion that the 'government's core is the people' and the ethos of 'government for and by the people' (European Commission, 2001).

7.3.1.6 Public services

Focus should be on redesign according to the needs of, and accessibility by, all citizens, to overcome inequalities in service access through co-creation with citizens and to deploy new technologies for the improvement of certain services under the condition that the state protects citizens' private data. An emerging imperative is thus to co-create value in public service delivery through public-private partnerships as well as public-civil and public-citizen partnerships. **Principles for digital public services** are user centricity, trustworthiness and security, inclusion and accessibility, interoperability, openness and transparency and innovative culture (European Commission, 2016a). Barcevičius et al (2019) find that the public services most impacted by digital technology include the application of physical robots (a very specific digital technology) in elderly care (a very specific service), healthcare and long-term care more generally, public safety and security, and smart city services.

7.3.1.7 Education and literacy

There is a need to ensure a better fit with the needs of citizens and job markets in the future, to link digital data literacy to a better understanding of the digital environment, to link policy and media literacy to a better understanding of policy processes and inclusion in civil and informed debates, and to ensure futures literacy contributes to more resilient societies. **Digital divides**, for example in terms of gender, education and ethnicity, need to be tackled. However, research also shows that the higher the level of internet and broadband coverage, the higher is internet use even for less educated and less skilled individuals. In addition, household internet access increases the educational attainment of individuals in a given area, even when internet coverage and GDP per capita are relatively low (Millard, 2015b). Ensuring internet, mobile and broadband infrastructure availability is necessary, but not sufficient, for more adoption and beneficial use of digital technology. It is also necessary to create appropriate incentives, awareness, reward systems, as well as support provider and user ecosystems, driven by high levels of cooperation and co-creation, in addition to market competition. In the case of digital technology design, there is a strong trend in some developed countries to move away from technology designed purely for specific disadvantaged groups towards '**inclusion by design**'.

7.3.1.8 Big data and artificial intelligence (AI)

To maximise the use and value of big data and AI, it is necessary to develop stronger legal, ethical and standards frameworks and new policy perspectives, to focus policy-making on the public value of technologies, to ensure that digital companies comply with new European and national laws, and develop participatory technology that is more socially robust. See Sections 5.2, 5.3 and 5.4 above.

7.3.1.9 Redesign and new skills for public administration

There is a need to redesign and constantly re-evaluate the skills needed in public administration, to develop more creative and innovative solutions adapted to challenges and ensure that **agile forms of working** lead to positive outcomes and benefits for citizens (e.g. greater efficiency, better focus on citizens' needs and the faster identification of challenges and response to changes). The development of capable cadres of civil servants is key to effective public governance including digital public services. Traditionally, a focus on middle management has been key, but this layer is tending to reduce in size under budgetary pressure so **focus should also be on the quality of mainstream public servants, especially those in frontline service-interface positions**. They need to be empowered, just as much as the users of public services, through appropriate training, applications and processes. This will also serve to promote creativity, experimentation and innovation in a continuous search for improvement. In this context, digital is becoming an increasingly essential tool in service provision, so public servant skills training in this area is also critical. The JRC's provision of competence frameworks for policymakers and researchers working on public policy is a useful support in this (Schwendinger et al, 2022.)

7.3.2 Good governance public values from 2020 onward

Given the pivotal historical cleavage point Europe has been experiencing since 2020 as it becomes firmly embedded in the new age of turbulence, **a new societal contract is needed that combines being radical and realistic and is firmly aimed at Europe's changed situation**. The OECD (2020f) has responded to COVID-19 by developing evidence-based policy responses that are similar to, but build on and go further than, Vesnic Alujevic and Scapolo (2019) given the new situation. These might also constitute an upgraded good governance values blueprint for crises more generally, and are here further elaborated including from other relevant sources:

7.3.2.1 Democracy and power relations

The pandemic and the other crises currently being experienced, including the current geo-political clashes between the so-called democracies and autocracies, have re-opened the discussion both about what democracy really is and how can it be operationalised. "**Open democracy** envisions what true government by mass leadership could look like."(Nathan Heller, New Yorker¹⁶) What does a new model of democracy look like that opens up power to ordinary citizens and strengthens inclusiveness, responsiveness, and accountability in modem

¹⁶ https://press.princeton.edu/books/paperback/9780691212395/open-democracy
societies? How can we organise multi-level democracy in Europe? On the one hand, there is – and indeed a strong – need for very local democracy and governance, whilst on the other hand, we need citizen voices at regional, national and EU levels as well. This is not primarily a question of technology, though technology plays a critical role, for example by using AI, blockchain, etc., but is even more a question of political will, civil society co-creation, organisational efficiency, transparency and offline action coupled with open behavioural science to get people involved. We should not put all our democratic eggs in the technology basket. There is a strong need **blended/hybrid democracy**, for example as proposed by Hélène Landemore in her 2020 book "open democracy" and the need to go far beyond traditional representative democracy by creating new institutions that allow ordinary citizens to "take (back) power" Landemore (2020). Heller (2020) talks about the **future of democracy as "politics without politicians"** and uses Landemore's quote that "if government is for the people, why can't the people do the governing?".

Monbiot (2019) has called for '**political rewilding**' through '**radical trust**' in which government trusts people rather than only expecting that people should trust government, as is the normal approach to trust. In an analogy with the rewilding of nature, attempts to control politics from the top inevitably fail leading to vulnerability and breakdown, given that both nature and politics are highly complex and such control is always too simplistic, coercive and damaging. In all systems, complexity tends to be resilient, while simplicity tends to be fragile. Keeping nature in a state of arrested development in which most of its natural processes and its keystone species are missing makes it highly susceptible to climate breakdown and invasive species. But rewilding – allowing dynamic, spontaneous organisation to reassert itself - can result in a sudden flourishing, often in completely unexpected ways, with a great improvement in resilience. Monbiot (2019) argues that the same applies to politics. Mainstream politics, controlled by centralised party machines, has sought to reduce the phenomenal complexity of human society into a simple, linear model that can be controlled from the centre The political and economic systems it creates are simultaneously highly unstable, lacking in dynamism and susceptible to collapse. They become vulnerable to the toxic, invasive forces of ethno-nationalism and supremacism. But localisation, especially down to cities, towns and communities, can be part of a concerted political rewilding process in which a thousand flowers can bloom, as shown in many examples in Sections 423 and 4.3.2, as well as in in the accompanying the state-of-art report. Clearly, central coordination, law-making, regulation and standardisation are important in providing a framework in which this can happen, underlining again the governance challenge of achieving the appropriate balance in a specific context.

Additional examples perhaps illustrate the possibilities. **Citizens' assemblies** are a form of deliberative democracy: processes through which citizens can engage in open, respectful and informed discussion and debate with their peers on a given issue, informed by expert evidence, facts, and possible solutions. A typical citizens' assembly is made up of a representative group of around 50 to 200 citizens, who are chosen at random from the general public, like a jury. The selection of members is normally stratified to ensure that participants are as representative as possible of the general population according to certain criteria – usually gender, age, ethnicity, geographical location, and social background. Members of the assembly normally meet over a number of weekends – from two weekends to twelve or more – to learn about, deliberate upon, and make recommendations in relation to a particular issue or set of issues. This is typically supplemented by online digital material and discussion. The topics debated by an assembly are generally set by government or parliament, though members can sometimes choose their own agenda (Electoral Reform Society, 2019).

Nesta (2019b) contends that 'mass involvement' in shaping the future can move beyond citizen assemblies. There are currently two damaging trends hindering mass involvement in solving current and future societal challenges: first fatalism which assumes that ordinary people are powerless in shaping their future and, second, so-called 'elite futurism' usually emanating from business, big government and 'experts'. Although experts are definitely needed, these barriers can be countered by "opening up and democratising the future". There are many tools available which can make it easier to forge a consensus on actions now that may have a big impact in the long run. There are many methods for mobilising far larger numbers of people in thinking about the future rather than relying just on professionals and experts, though the latter are absolutely still necessary, but not with a monopoly on power and voice. For a long time, participatory futures methods relied on group workshops, interviews, and in-person discussions, in much the same way that many public engagement exercises still do in the physical world. In recent years, however, this has begun to change. At the same time, digital technologies are expanding the reach of futures exercises enabling more and diverse citizens to play, create and participate virtually in future worlds. This is generating ideas and sharing information involving new players such as artists. designers, game makers, writers and psychologists. Over the last decade, this has led to an explosion of new ways of thinking about or experiencing the future, a phenomena that can be described as 'mutant futures' because of the combination of approaches involved that can currently be grouped into five categories: play, immerse, sense, create and deliberate (Nesta, 2019b).

New techniques are making it easier than ever for governments to draw on the knowledge and expertise of those outside of government to improve planning, policymaking and implementation of government programmes. **Collective intelligence** is a new term to describe something which is in some respects old, but in other respects changing dramatically thanks to advances in digital technology. It refers to the ability of large groups - a community, region, city or nation - to think and act intelligently in a way that amounts to more than the sum of their parts. It encompasses other movements - from open data to civic tech - but links them to the broader question of how governments make decisions on our behalf. An analysis of global collective intelligence initiatives finds that activities fall into four broad categories supported by new digital tools: better understanding of facts and experiences; better development of options and ideas; better, more inclusive decision-making; and better oversight of what is done from monitoring corruption to scrutinising budgets (Nesta (2017). Another high value example is the POLIS software pioneered in Taiwan that focuses on finding consensus rather than division between citizens on public policy issues (see Section 4.3.3).

7.3.2.2 Trust and trustworthiness

A new paradigm of public trust is needed based on high levels of integrity, fairness and openness in institutions and on government's competence, i.e. its responsiveness and reliability in delivering public services and anticipating new needs as they arise especially in crisis situations. Hale (2022) contends that **trust between people – even more than in government or institutions – is key to limiting damage in a pandemic**, as shown by the Oxford COVID-19 Government Response Tracker. In 2019, the Global Health Security Index published a report ranking countries on their preparedness for pandemics. The US scored highest, followed by the UK. Two years later, both countries rank among those with the greatest loss of life per capita from COVID-19. A large part of the answer is trust, but this is not mainly about trust in governments or institutions, but rather how much people think they can trust other citizens whom they don't already know, i.e. societal trust Trust in other people is so important because many aspects of fighting a pandemic – such as social distancing – require collective action.

It is also important to rethink trust in the age of digitisation given there is some belief that trust will become obsolete with the advancement of technology. Blockchain, the internet of things (IoT), cloud computing, and machine learning differ significantly from earlier generations of technologies in their enhanced capacity to make decisions autonomously and act 'smartly' while also enabling unprecedented interconnections between humans and machines and interoperability among a wide range of different technologies. Trust remains a central facet in acting as a lubricant for collaboration, but **digital technology does change trust** in terms of what form organisational trust takes, how it is produced, and who needs to be trusted.

First, **trust is becoming much more impersonal** given that in many, though by no means all, contexts, there is a radical shift away from conventional interpersonal collaboration in which the identity of the other party provides a focal point for initiating or sustaining a trusting relationship. An increasing number of transactions occur with individuals we never meet in person or have only an indirect tie to. For example, in the blockchain where no party can unilaterally change information, participants do not need to trust the integrity of their partner, but they do need to believe in the reliability and thus the trustworthiness of the blockchain system.

Second, **the way trust develops is changing**. While, traditionally, trust has mainly emerged from past and expected future interactions between collaborating parties, it is now more closely tied to the characteristics of the entity being trusted which can render it trustworthy or not. For example, machine learning algorithms, which have been widely employed in such industries as insurance to ascribe trust scores to people based on their social categories. At the same time, trust also becomes more institution-based, now being created through technological systems. Protocols and code-based algorithms establish unambiguous rules and create specific expectations for transactions. Third, there is a shift in who needs to be trusted. Despite trust being more impersonal, there remains a need to trust certain human actors, although the targets of such personal trust shift from the other party in the transaction to three sets of entities. There is a need to trust i) the entities that designed the system (e.g., developers) not to introduce loopholes or personal biases, whether intentionally or unintentionally: ii) individuals and organisations who provide data and information for the algorithms; and iii) data holders—those who aggregate, analyse, and profit from the information they control—to not abuse the data or violate privacy agreements (attributed to Lumineau et al, 2022.) In other words, how can these entities appear as trustworthy by their users and how can they live up to this in practice?

7.3.2.3 Risk governance and crisis management

Risk governance and crisis management is needed for critical infrastructure, scientific advice and anticipating future shocks. **One of the biggest obstacles is legacy and the resistance to change** and this applies both

to the technological and human dimensions. The entrenchment of a **'risk adverse culture**' and 'business as usual procedures' remains strong within government at all levels, creating an inherent barrier to the introduction of new processes, products, services and methods. A culture of 'open innovation' amongst civil servants should be encouraged, ensuring a cultural mindset that is flexible, adaptable and responsive to user feedback. Although **some 'bureaucracy' is necessary**, for example to ensure decisions are made according to statutory and other rules, greater discretion is also needed. This should be based on big data, user feedback, open engagement, common sense, as well as ethical principles, and should take place transparently and openly to allow scrutiny by society at large. (See also public sector innovation, experimentation and learning below.)

7.3.2.4 Regulation

New ways of looking at regulation are needed, for example in terms of regulatory quality, joined-up regulation, removing administrative barriers and risk-based regulation using AI. According to the World Bank (2017), **Risk-Based Regulation (RBR)** should be introduced in order to achieve public policy objectives by targeting activities that pose the highest risk to the public well-being, and in turn lowers burdens for a variety of lower-risk sectors and firms. However, a different and perhaps complementary approach is taken by Hidvegi et al (2021) who contend, using AI as an example and based on consultation with numerous civil society and other actors, that regulation should be based on **human rights** rather than risk assessments.

7.3.2.5 Public sector innovation, experimentation and learning

COVID-19 is seen as a catalyst for government transformation. According to Cottam (2022), in this century the need is for imagination, for creative team-based abilities, for spiritual enquiry and above all to acquire the capability for learning that can underpin both rich lives and strong communities with new forms of economy at their heart. Learning from previous technology revolutions, and drawing on work by Perez (2010), each of the four quadruple group of human actors needs a new cadre to bring about transition (Cottam, 2022):

- The state (public sector): a new generation of leaders who will dare to forge new alliances and design new frameworks.
- New industrialists (private sector): business leaders who, walking in the footsteps of enlightened forbearers, will challenge their peers believing that a new era is only possible with the design of new social systems and in particular new norms for labour.
- Organic intellectuals: those who can produce new ideas inspiring global imaginations in all disciplines (educational and research/knowledge institutions).
- Organised civil society: artists, movement makers, labour unions, activists, those who bring creativity and above all lived experience.

According to Cottam (2022), **the state is critical as the prime mover**, able to lay out the framework, the guidelines for investment and the principles that must govern a collective project such as transition. The impasse we see in the current age of turbulence is rooted in the state's own imprisonment in industrial structures and mindsets. But this transition cannot be imposed from the top: it will be rooted in new forms of horizontal institutions and relationships. This means **the local state** should play a critical role and is starting to become more active and claim for itself greater agency and power. Despite the lack of financial resources and often a lack of access to networks, those in the local state understand local realities.

However, much of the dynamism in localities comes, not from a formalised and relatively stable local state institution, though this is absolutely essential, but from the greater informality of communities, groups, citizens and movements that transcend formal boundaries The relative transient and fluid nature of the latter is also absolutely essential but there needs to be much better and heightened awareness, recognition, mutual learning and co-creation between the more formal and informal, in which digital interconnectivity also plays a critical role, similar to the relationship between agency and structure (see Section 7.4 below). Both **tacit (uncodified) and explicit (codified) knowledge** are together powerful combinations for learning, identifying good practices of successful transformations, and thereby for widespread co-creation, replication, scaling and knowledge sharing.

A good example of this is the reciprocal relationship between federal and local government structures, on the one hand, and with less formalised local civil cooperatives and activities in governing local energy systems on the other, as outlined in Section 6.2.1. Indeed, such co-creation is necessary to recognise and move beyond the cognitive barriers imposed by the '**shifting baseline syndrome** '(see Section 4.4.2). As demonstrated by Monbiot (2014), this describes resistance to change in the accepted norms about how we see and do things

and may be partially an issue of cognitive dissonance around the cultures of doing. In this sense, some of these concepts become philosophical issues that perhaps go beyond the remit of public governance, although we know from the examination of existing paradigms that this remit has constantly changed and will need to continue to do so, possibly in unexpected ways and perhaps as outlined in the strategic challenges in Section 4.

Many established persons and institutions often suffer from a shifting baseline syndrome when they fall back upon the notion of **"this is the way we have always done things around here"**. This may work if it provides some stability, continuity and predictability (see Section 7.4.2.2), but is likely not to work at all when confronted by massive shocks as currently being seen and, indeed, may do more damage than provide any benefit. Such cognitive dissonance often persists in spite of ready knowledge and understanding of empirical evidence about what works and what does not. It may require the influx of a younger generation or the intrusion of a massive societal shock to shift the baseline. On the other hand, there may be good reason not to follow the example of many in Silicon Valley who attempt to "move fast and break things". A better motto would be **"move fast and restore those things that demonstrably work and repair those things that don't"**, though this clearly does not have the same memorable clang. So, **balancing the need for governance stability with the need for governance flexibility, agility and innovation is, perhaps, the most difficult governance task of all (Millard, 2015a).**

7.3.2.6 Behaviour, values, principles and culture

It is important to build behaviours and cultures across the whole public sector conducive to responsible leadership, professionalism and the values of integrity, transparency and accountability. In turn, these principles need to rest on the rule of law, justice, respect for human rights, and the security of both persons and property. In terms of policy-making, Mair et al, 2019) recognise that advances in behavioural, decision and social sciences demonstrate that humans are not purely rational beings and that understanding and accounting for this brings new insights into political behaviour which also helps address some of the current crises of democracy. Ethical behaviours can be promoted through a **code of ethics** incorporating principles like transparency, nonviolence, resilience, participation, solidarity, identity and innovation. This should help to create values of commitment to serve the public and to counter adversarial attitudes by public servants to citizens. Corruption and lack of accountability lead to bad governance and typically contribute to a vicious cycle involving demotivated public administrations, and a lack of willingness and of capacity to innovate and modernise public services. In terms of open government, transparency and co-creation there can also be a (perceived) risk of losing power by empowering others. However, when others are empowered, this itself requires a need for balancing between the role of public administration to serve the others (and reducing the public governance burden for them) as compared to asking them to be part of development and delivery thereby putting an extra burden on them. It is clear that not all public services neither can nor should require co-creation given this is resource intensive so should be restricted to those challenges where additional actors are needed and desirable.

An example of this is **citizen science** which uses co-creation approaches between citizens (the general public) and scientists deploying bona fide scientific processes through an open and inclusive approach. The aim is to empower and enable citizens to support and be part of the exploration, shaping and development of the different aspects of scientific activities that use scientific methods and results to meet societal challenges (ECSA, 2021). Citizen science can range from the short-term collection of data to the intensive use of leisure time to delve deeper into a research topic together with scientists and/or other volunteers, to ask questions. and to get involved in some or all phases of the research process (ECSA, 2020). It can both accelerate and produce new scientific knowledge; it can help decision-makers monitor the implementation and regulatory compliance of their policies; it can increase public involvement and understanding of science and help citizens feel that they also own these policies; and it can enable faster evidence-informed reactions to events and the exploitation of new opportunities through new sources of both quantitative and qualitative knowledge (Shanley et al, 2019; Schade et al, 2020). Citizens can often reach parts of the real-world evidence base that are otherwise very difficult to reach, for example with better territorial and fine-grained data coverage. This will simultaneously increase their own scientific awareness and capabilities as well as improve the science and its contribution to better evidence-based outcomes and impacts. An important point is that citizens can determine themselves when, where and how much they wish to be involved.

7.3.2.7 New innovation frameworks and mission strategies

Collaboration, diversity and a range of voices, skills, competencies and resources, form the basics of successful innovation. In order to meet the challenge of the EU's 2021-27 policy programmes (European Commission, 2021e) and the UN's 2030 Agenda for Sustainable Development (United Nations, 2015b), new forms of

innovation, beyond but building on digital technology and top-down driven innovation, are required. For example, a new approach is **open innovation** that is designed to ensure that all can be involved, where there are no supposed monopolies of innovation talent and potential, and where the solutions become owned by as many people as possible which results in greater acceptance, trust and impact, such as through co-creation. The innate innovativeness and activism of civil society, communities and citizens as the ultimate beneficiaries of innovation, need to be liberated. Just as greater democratisation and participation are needed at the local level through increased devolution and decentralisation, it is important to re-invigorate active localism and entrepreneurship in villages, towns, cities and regions. Prosperity and inclusion thrive when the state genuinely hands power to local people, local entrepreneurs and local organisations.

Every locality needs to innovate and implement its own strategy based on its own needs, capacity and resources, systematically enabled and supported by the state through setting up **local open innovation and entrepreneurship hubs** where these do not already exist. These will provide opportunities and incentives for the full range of innovative activities, ranging from community projects, social innovation and entrepreneurship, through repair cafés, small scale manufacture and fab-labs, to fully commercial start-ups and high tech activities. In order to serve the local social and business communities, partnerships could be established as locally appropriate together with local anchor institutions like libraries, educational institutions and commercial training and apprenticeship facilities. Acting as open innovation labs these can assist everyone who has a good idea or needs to get advice. The local focus could also aim to identify champions and network good practices by sharing experiences and scaling as widely as possible. Regional clusters can be established, or further supported and extended where these exist.

Experimentation and evidence show, for example, that publicly owned local anchor institutions (see Section 4.2.3), including activist local councils, as well as local educational and health organisations, underpin longlasting prosperity. Furthermore, this can be greatly assisted by regional cooperative banks attuned to local conditions and providing low interest loans to facilitate investment in the area. The state's role is to provide an enabling framework of policies, regulation, incentives and funding for these local initiatives. One primary aim could be to retain and reinvest as much as possible of the wealth created locally back into the local economy and society through community wealth building that sees these institutions buying as much as possible from local businesses and organisations. Other innovative examples include experimenting with and supporting local basic income schemes and supporting employee-owned enterprises. It might also be useful to enable and encourage inclusive and transparent complementary local currency or token systems that are fully voluntary and additional in order to complement, not replace, traditional currencies in order to avoid financial lock-in.

The role of the state includes strategies and regulation that focuses on mainstreaming of public investment and public procurement to prioritise green and especially local solutions and that gives the public interest a direct and formal role in innovation and industry. Adopting a **'mission-driven' strategic approach**, as advocated by Mazzucato (2019) and as part of the Horizon Europe Research and Innovation Programme (European Commission, 2021l), was first used successfully by President Kennedy to get to the moon in the 1960s. A mission defines the overall strategic direction and clearly lays out the beneficial impacts of innovation and the economy, without tightly regulating how this is done. This involves a **three-step 'tight-loose-tight' approach**: first establish clear ambitious but realistic goals, such as a net-zero carbon neutral economy by 2035; second open the field for any innovations attempting to meet these goals, as long as they are ethical, non-exploitative and legal; and third tightly assess and evaluate their actual impacts in order to adapt, leam lessons and improve.

Government policy and regulation, supported by mission-driven public procurement, could equally focus on public and environmental goods as well as private and commercial goods. It is necessary to find the right balance between these goods, often seen as being in competition, and seek win-win rather than win-lose tradeoffs. The government remains the largest actor in the economy since they have the largest purse, which was in clear evidence during the pandemic, so it could nudge and direct economic growth towards mission-specific ends through its spending, active intervention and framework-setting.

7.3.2.8 Public and governance management and integrity

Both executive capacity (in terms of steering capability, implementation and institutional learning) and executive accountability (participatory competencies) need to be expanded to ensure **longer-term resilience-based planning, preparation and prevention**. New resilience-enhancing structural reforms for improving the quality of public services and getting back on the path of sustainable and inclusive growth can contribute to the EU's recovery from the COVID-19 crisis. The European Commission is providing tailor-made and technical Structural Reform Support to help EU countries design and implement reforms as part of their efforts to support

job creation and sustainable growth (European Commission, 2022j). Governance integrity is also important for curtailing corruption and ensuring open transparent reporting of budgets and of public management, green budgeting, budgetary oversight and independent fiscal institutions. This includes the need to focus on countering illicit trade, trafficking, counterfeiting and fake products and services. It is also important to focus more intensely on infrastructure and public procurement, including through transparent inventories and open processes.

7.3.2.9 Future proofing

According to Moffitt (2022), from 2020 there is a need to augment planned change and innovation, which has been the norm since about the 1950s and characterised by consumerism, automation, entrepreneurship and commercialisation. A shift is needed to 'future-proofing' in order to tackle the new age of VUCA, i.e. where the world is volatile, uncertain, complex and ambiguous. The author terms this as 'Change 5.0' influenced by the growing importance of scenario planning, intrapreneurship, agility, co-creation, new business models. ecosystems, circular economy, augmented intelligence, Industry 4.0, and growth hacking. The 'Change 5.0' model needs to respond to VUCA with what Moffitt (2022) terms TANCES, i.e. technology, agility, need (customer), cross-industry, expectations {culture} and societal (challenge). This is mainly a business perspective. but also has relevance for societal governance and the specific role of the government actor as it recognises the dramatic changes and challenges of the new age of turbulence (VUCA) and also offers a response (TANCES). Clearly, one of the signal roles of the government actor in societal governance is to adopt an approach something like TANCES, whilst also balancing this with a strong commitment to continuity, stability and predictability (see Section 7.4.2.2). The government actor has a much more complex set of tasks than business actors. Part of its role is to provide a societal governance framework in which relevant business and economic activities could readily adopt TANCES, as long as this provides value for society as a whole, in order to respond to VUCA. One way in which government can do this is by adopting a 'mission-driven' strategic approach as discussed in Section 7.3.2.7.

7.3.2.10 Policy coherence, co-ordination, alignment and evaluation

Policy coherence, co-ordination, alignment and evaluation are required for joined-up resilient and regenerative responses. This will also need to use, for example, KPIs in policy performance and policy outcomes aligned with the three substantive SDG pillars: economic, environmental and social, as well as domestic action taken by governments sensitive to international responsibilities. An interesting recent example in New Zealand, that is not widely rolled out, is making government rules and legislation machine-consumable for tracking public value impacts. This enables much more accurate predictions of how a proposed new policy will affect the total constellation of existing laws and thus create public value in society at large at very low cost Government is empowered to avoid implementing policies that subsequently will be shown to be either counterproductive and/or overall more damaging than beneficial. This involves breaking down the law into its component parts using digital code to show as accurately as possible what is likely to happen to whom under which conditions. Thus, law-making proposals can be gueried before being finalised at much greater speed and at much lower cost. For example, if a proposed law in one entity involves some people losing their jobs they will need to received benefits from another entity and this in turn is likely to have other knock-on effects elsewhere. Thus, different policy scenarios and their overall public value impacts can be examined in advance in terms of their benefits and downsides for the public sector as a whole, as well as for wider society, and the proposed law adjusted accordingly. In addition, making machine-consumable government rules available to other actors, such as citizens and businesses, could also greatly increase transparency, responsiveness, accountability and overall public value. (New Zealand Government, 2018). Fundamental aspects of all these efforts should include always aligning and validating policies with the needs of, and opportunities provided by, nature (see Section 4.4.1), as well as breaking down silos (see Section 7.4.4). The JRC's provision of **competence frameworks** for policymakers and researchers working on public policy is also a useful support in this (Schwendinger et al, 2022.)

7.4 Common features and framework conditions

7.4.1 It is the mix that matters

Some important issues arise from examining the progression of public governance paradigms and their use of digital technology. First, it is the mix that matters; any given system of governance will have varying effects depending on its context, so the need for diverse inputs is paramount while acknowledging that institutions are not a panacea for societal challenges. Second, participation and dialogue need to be given much greater priority

in a governance approach, i.e., even when experts are setting policy, dialogue needs to take place with all actors, especially those who are the direct targets of the policy. Participation is not just a normative goal but is also a means to greater efficiency and more effective outcomes, and will help to overcome overly simplistic and normative thinking. Third, developing the parameters of governance is often limited by formalised approaches; in reality, social terms and categories (such as, elections, class and democracy) are inherently 'fuzzy' without natural boundaries, and this makes analysis complex and difficult (Bevir, 2013).

Public governance usually refers to one or both of two related topics: state institutions and service delivery. Since 'governance' became a policy agenda, the main approaches to it have been dominated by formal theories derived from economics and sociology: (a) economic theories generally focus on micro-level and suggest markets are efficient ways of securing equilibrium; and (b) sociological theories generally stress mid-level contexts, describe new times, and promote networks and partnerships as responses to those times. The danger, however, is that much of the discourse has become overly descriptive as well as too keen to see cause and effect relationships. It is important to **beware of so-called 'causes'** as these are often in direct conflict with concrete human activity and experience. For example, the sociological theories of governance sometimes encourage essentialist views of organisations, markets and networks, for example that networks are always innovative and that markets are always efficient. Such posits may not always hold as they are only contingent generalisations at best, as well as being over simplistic and often lull policy makers into thinking in terms of certainties of outcomes that are far from guaranteed. Concepts like markets and networks are only heuristic ideal types, not found in reality in pure form. Instead, governance paradigms should recognise the importance of 'concrete human activity' and real 'human lives' (Bevir, 2013).

The multi-level governance of the EU (see Section 7.4.3) and the complex and varied socio-economic and cultural histories of European nations, coupled with their diverse and often volatile political realities, has led to the public governance paradigms sketched in Figure 1 building on each other, layer by layer. This is not a linear progression where one replaces the other but represents a process of partial overlapping, sedimentation and co-evolution over time in which all paradigms still have relevance and impact today, although some are dominant and others less significant. In this context, it is the precise mix that matters in each context, time and place.

7.4.2 Finding balance

According to Vibert (2001), coherence is about identifying the qualities that make a system of government a good one. Coherence is basically concerned with achieving a series of balances between competing interests and requirements: between simplicity and complexity, between stability and change, and between structure and agency, as well as the rights and responsibilities each actor has in these contexts. Finding balance is especially important in the possible emergence of a third generation of the EU's multi-level governance system by supplementing the many public governance mixes that already perform well, but also moving towards various forms of 'generative public governance' (see Section 7.4.3). Finding balance can also draw upon **negotiation theory** based on decision analysis, behavioural decision-making, game theory and negotiation analysis. Negotiation is a strategic discussion that resolves an issue in a way that both parties find acceptable. Individuals should make separate, interactive decisions; and negotiation analysis considers how groups of reasonably informed individuals should and could make joint, collaborative decisions. Such theories are interleaved and should be approached from a synthetic perspective.

7.4.2.1 Balancing simplicity with complexity

The public sector is unavoidably complex. It confronts a range of complex dilemmas, needs to fulfil complex tasks in a myriad of complex circumstances in relation to many different actors, and it has to steer towards complex and often contradictory policy goals. This means it can often become obtuse and unintelligible to citizens, as well as to civil servants themselves. Despite this complexity, **simplicity is also necessary**, **particularly in the 'front-office'**, i.e. the public interface. An ideal balance may be to allow complexity in the 'back-office' whilst building simplicity in the 'front-office', but in other situations such a balance may not be possible. According to Vibert (2001) simple government structures are needed to ensure the political and market choice systems properly complement each other. Simplicity makes it easier to identify what is important and what is not important, and also makes government more understandable to people.

There are **both benefits and dis-benefits of simplicity and complexity**. Can people trust something they cannot understand? Simplicity can be seen as part of the ethos of open, transparent and accessible government, particularly if this helps to ensure that the legislation, rules and regulation governing a given public service or value is sufficiently simple and understandable for most actors to appreciate and exploit. A society governed

fully or mainly by a legal and regulatory framework which is obtuse, hidden or can only be understood and wielded by experts, could be argued to be profoundly undemocratic, which is a possible danger of New Public Management. There can sometimes also be a challenge for the EU is not appearing as a 'bubble in Brusses' where many EU institutions are based, but should rather be seen as more than the sum of its parts (all citizens, member states, regions, businesses, overarching institutions, etc.) The COVID-19 crisis has sharpened awareness of the EU and most people want more. Three out of four respondents across all countries say they have heard, seen or read about EU measures to respond, and around half (52%) of those who know about EU action in this crisis say they are not satisfied with the measures taken so far. There is a strong call for more EU competences and a more robustly coordinated EU response whilst there is dissatisfaction expressed by a majority about the lack of solidarity between Member States in fighting the pandemic: 57% are unhappy with the current state of solidarity, including 22% who are 'not at all' satisfied (European Parliament, 2020).

Conversely, simplicity may compromise quality and effectiveness. In the UK, the tax system became much more complex since 1997 but as part of an attempt to make it fairer by targeting the poorest through means-testing and tax credits. Without this complexity, precise targeting could not easily take place so that both rich and poor citizens would, for example, not receive tax breaks. Fairness and efficiency seem to mitigate against simplicity. A similar problem confronts the Danish tax system but has been tackled by most citizens and businesses receiving pre-completed tax returns from government, which they can either accept or correct providing relevant evidence, based on a standardised ID number and government access to relevant personal data from employers and banks, so that complexity has been side-stepped. The tax system is complex, but this does not matter for most citizens and businesses because the government, in effect, calculates their tax for them. Clearly, high trust in government is necessary for acceptance of such a system.

7.4.2.2 Balancing stability with change

The need for public governance to transform, change, be flexible, innovative and dynamic, and be adaptable, is uncontested. However, **governments have a critical role as the only formal institution which can provide much needed continuity**, predictability, dependability and stability. This is absolutely necessary for individuals, families and communities to lead meaningful and peaceful lives. It is also necessary for business in order to give them a level playing field, and some longer-term certainty about investment and future developments. The stabilising and continuity functions of government should probably be preserved and further developed, even within a strategy of a dynamic and transforming public sector. In other words, some important questions are "what needs to be the stable frame" and "what can become agile and flexible?". Governments are often derided as 'risk averse', but this may be at least partially due the complexity of their tasks (as above) as well the need to ensure a strong measure of stability. Re-evaluating risk and embracing relevant experimentation and innovation are both certainly necessary, especially in turbulent times when governments need to act fast often with limited information. The most prominent risks for politicians are often losing the next election, rather than not achieving longer-term policy goals, a conflict which is significant challenge to democratic governance. (See also Section 7.3.2.)

In balancing stability with change, Agre (2000) makes a **distinction between institutions and organisations**. An institution is a persistent form of relationships among people so is often strongly related to culture and behaviour. Examples include horseracing, the medical system, greeting rituals, the university, the stock market, management consulting, religion, the family, the common law, public institutions and the nation-state. Institutions can vary across history and between different societies, but they are remarkable for their ability to remain relatively unchanged, sometimes for hundreds of years at a time. Institutional persistence is a dis-benefit when the institutions are unjust or inefficient, but it is a benefit when it enables people to predict the future, focus their attention, and compel others to keep their promises. Institutions generally change slowly but can change relatively fast during a profound period of disruption or shock. In contrast, organisations (such as government agencies, non-profits, and civic associations, as well as private companies) tend to be ephemeral and can change very quickly, often without any significant or even obvious cause. For example, there are distinctions between the institution of the university and particular universities as organisations, and between the institutions when unravelling public governance, especially in a turbulent period as well as in the context of digital technology.

7.4.2.3 Balancing structure with agency

The evidence presented in this report show the need to govern the balance between structure and agency and to recognise and address the duality and interrelationships between them (Millard et al, 2017; McGarvey, 2022):

- Structure: the governance and framework structures within which people, families, communities, businesses, as well as institutions and organisations of all types operate that are largely beyond their own immediate control
- Agency: the governance support and degrees of freedom for all these actors to both develop and meaningfully deploy their own agency to function successfully within these structures and over which they have, at least partially, a good measure of own control. Agency is also strongly mediated by an individual's or a group's values and identities (Scharfbillig et al, 2021).

The ideal public governance goal should be that structure and agency increase their mutual overlap, so they coincide as closely as possible. However, it is recognised that in practice public governance and other structures (both formal and informal) take longer to adapt and reform than agency demands in response to, for example, crises, shocks, inertia, ingrained cultural practices, vested interests, and even of course corruption, cronvism and paternalism. Thus, public governance needs to address both structure and agency. For example, changing the system to reduce homelessness and unemployment should go hand in hand with helping the homeless or unemployed themselves find accommodation or work. Too many social and economic policies tackle poyerty and inequality by only attempting to improve the 'agency' of the poor through training and so-called 'empowerment', instead of also tackling the root causes embedded in societal structures that prevent the poor from making the most of their own efforts (Millard et al. 2017; McGarvey, 2022). Legal and regulatory frameworks need to be conducive to maximising individual and group agency whilst also maximises the creation of public value. These frameworks need to be re-cast to ensure fair distribution of social, economic and environmental rights and benefits that are all too often undermined. Labour market issues are paramount, for example in relation to the burgeoning 'gig' economy, the problems of zero-hours contracts, and ensuring people have stable, simple and predictable structures so they can thrive in both their working and private lives. Policymakers should aim at specific outcomes and impacts and open up for process innovation as long as these processes remain ethical, transparent and legal (Mullainathan and Shafir, 2013; McGarvey, 2022)).

7.4.3 New multi-level governance roles

Multi-level governance refers to the way power is spread vertically between many levels of government from the global/European to the local, and horizontally across multiple guasi-government and non-governmental organisations and actors. In the EU's formal constitutional context, this rests upon various balances between centralisation and decentralisation and especially the subsidiarity principle that serves to regulate the exercise of the EU's non-exclusive powers. It rules out EU intervention when an issue can be dealt with effectively by Member States themselves at central, regional or local level, so the EU is justified in exercising its powers only when Member States are unable to achieve the objectives of a proposed action satisfactorily and added value can be provided if the action is carried out at the EU level (European Parliament, 2022b). Subsidiarity thus determines the level at which particular governance powers and levers should be exercised which might be decided on a case-by-case basis. There are always strains and tensions in multi-level governance systems as the different actors, constituencies and needs jockey for position and, in most cases, reach compromises, whether politically imposed or widely agreed, for mutual benefit. Sometimes, these can be relatively informal compromises that reflect the different (mostly national-level) cultural traits that administrators, policy- and decision-makers need to take into account. With the dramatic changes and shocks now impacting Europe since 2020 such adjustments become ever more pressing and where digital technology can open new possibilities.

The Periscope Project (2022) conducted empirical and theoretical research into best practice in multi-level governance during pandemics in countries across Europe during Covid-19 from March 2020 to May 2022. Findings were clear, including that **decentralised governance was critical in implementing pandemic policy and compliance** with it, and that **communities and civil society organisations played a key role** in closing the gap between statutory services and community needs, especially for vulnerable groups. This manifested itself in innovative forms of collaboration and mutuality formed at different levels of government, facilitated by favourable legal and financial environments. However, in times of great stress, overall pandemic policies and governance approaches also generated new forms of stigma, exclusion and inequality as well as exacerbating existing forms, whilst scientific evidence played only a mixed role in informing policy making and governance.

These insights show how public governance during turbulent times, and especially multi-level governance, needs to **focus on risk awareness**, **preparation and preventiveness** – especially when these risks will have massive impacts even if their probability is low. As argued in Section 4.1.3, the **necessary diversity resulting from the importance of decentralised governance structures needs to be linked through the strong**

interconnectivity of communication channels and coordination mechanisms. This also requires that civil organisations are empowered and well-funded to advocate for the needs of specific, and especially vulnerable groups. Innovative funding and legal structures need to be in place to allow for rapid redistribution of funds and for important collaborations to be sustained through periods of crisis and beyond. Attention needs to focus both on the structural barriers created by pandemic bureaucracy that exclude certain groups from uptake of, for example, vaccination, economic measures or healthcare, as well as on non-human factors, including a broad engagement with the needs of animals and plant-life and the impact of built environments on health outcomes. Investment in social listening mechanisms that allow governments to understand, adapt and co-design their policies with communities, specifically using qualitative and ethnographic data is also important, as is setting up a broad and diverse evidence base to inform policy making, facilitated by interdisciplinary collaboration among scientific research actors and channelled through strong communication mechanisms (Periscope Project, 2022).

Many instances are already apparent of building between, and linking, multiple governance levels and actors, ensuring that the critical local and community levels are also fully incorporated which has not always been the case. For example, in an analysis of transformative social innovation for sustainable rural community development, Castro-Arceab and Vanclay (2020) formulated the concept of '**bottom-linked governance**' as a multi-level middle-ground that links bottom-up community initiatives and top-down structures. This is where actors from various political levels, geographical scales and industry sectors come together to share decision-making. Community initiatives have the potential to be more widely transformative, but to do this, they need to scale-up from the local social demand level and provoke changes in the governance system at the systemic change level. The authors identify **bridging-role functions** like network enabler, knowledge broker, resource broker, transparency and conflict resolution agent, and shared vision champion.

Such 'middle-ground', 'middle-around' linking roles, intermediaries and breaking down vertical as well as horizontal silos between governance levels are clearly necessary. However, it is also important not to increase unnecessary bureaucracy, i.e. that reduces efficiency without increasing effectiveness, when adding levels or supplementing them with new roles and links. Further, there is a need to recognise that many non-government actors, especially citizens, do not know, do not wish to know and do not differentiate between the different governance levels, except perhaps if specific arrangements cause them problems. Joining-up and providing an holistic external governance face is thus very important and where digital technology clearly has an important role to play. There is still a strong top-down dynamic in Europe, and clearly this is important not least for law making, justice, coordination and standard-setting and monitoring, as well as external relations. In contrast, participatory interests promote a bottom-up dynamic even though much is governed by legal and constitutional levers at higher levels. How would data/information-sharing reflect the interactions between different governance levels and how could more efficiency as well as effectiveness be promoted? For example, to what extent is regional governance useful if local aggregations are more fluid and able to respond to collective needs? Who could the actors be to enable that and what are their clearly defined roles?

In light of the above considerations and especially in the context of the new 'age of turbulence', it would be opportune that the EU reconfigures both its ideas and practices around the most appropriate multi-level governance models now required. This could be seen as a **'third generation' of EU multi-level governance systems**, building on the first and second generations identified by Stephenson (2013). The 'first generation he characterised as the novelty of new governance forms and how they could transform basic institutional structures, while the 'second generation' steered multi-level governance towards new modes of governance and regulation (Conzelmann, 2008). However, the first and second generations both fall short on some of the paradigms highlighted in this report these, so a third generation might be characterised as one that addresses inter- and intra-institutional relationships, suggestions to achieve true transparency, accountability and inclusiveness in the entire system, also by making the most appropriate use of digital technology, ultimately aiming at increased trust between all the actors within the EU and interfacing in the best possible way with those outside.

A third generation of EU multi-level governance might also supplement the many public governance mixes that already perform well, but also move towards various forms of '**generative public governance**', i.e. that are sufficiently flexible and resilient to enable specific emergent governance forms to be created in particular contexts for particular purposes such as unexpected shocks. The main mechanism behind generative public governance is co-creation and, while co-production was originally tied to service production, co-creation has broader applications in the field of public governance and involves a broader range of actors and activities. The co-creation concept both builds on and extends the concept of collaborative governance and enables a strategic turn to a new type of 'generative governance' aimed at solving complex problems by constructing platforms

enabling the formation of arenas for co-creation that bring together a plethora of public and private actors, including citizens, in creative problem-solving processes. In this sense, generative governance envisions the broad-based participation of actors in **the co-creation of emergent and interactive responses to pressing problems and future needs** (Torfing and Ansell, 2021).

This might be illustrated in Figure 6 depicting a relatively flat but **interconnected 'fishnet' organisation of governance** composed of many actors sitting between fully top-down and fully decentralised modes. The 'fishnet' is sensitised to respond rapidly and resiliently to specific challenges in specific contexts, to address these flexibly and then return to a more stable state once a given challenge is dealt with. It is important to stress that such a generative public governance approach is most appropriate in addressing sudden societal shocks and is thus supplementary to governance modes that already perform well in addressing existing and relatively stable challenges.





Source: The author, taken from a presentation by Uffe Elbæk, Denmark, in 2016.

As mentioned above there are dangers in adding additional layers and complexity to public governance systems composed of multiple levels and forms, especially as seen from the perspective of citizens and other actors, so that a joined-up, whole-of-governance external face is definitely required. Such complexity, especially when locally-embedded and nuanced, multiplies the numbers of actors and relationships and potentially adds to the 'silo' problem. As pointed out in Section 4.2.3, however, it is often at the local level that it is easiest to break down governance silos and link across actors and sectors given their smaller and shorter interaction and transaction chains and much more shared and fine-grained knowledge. both implicit and explicit, about local conditions and contexts. It is likely that these issues can only be addressed using human-centred digital data and technology that fully enables different mixes of top-down, bottom-up, central, decanter, local, regional, national and EU units. Also critical are both territorial cross-border and crosssector governance forms which are central to the discussion on interoperability, not just digitally but also organisationally, legally, regulatory and in terms of policy. In a European context, this is likely to be taken further forward by the December 2022 proposal for the **Interoperable Europe Act** to introduce a framework of cooperation between EU Member States and EU Institutions. The aim is to secure the crossborder exchange of data and agree on shared digital solutions, such as open-source software, guidelines and frameworks, and will provide incentives for innovation in the public sector and for "GovTech" projects, based on public-private partnerships (European Commission, 2022k).

It is clear that it is the precise mix that matters in any given situation, so that co-creative contributions are allowed, supported and empowered to come from anywhere in the system. However, this must focus on the challenges that are best addressed by those actors taking responsibility, possibly as an advanced interpretation of subsidiarity and proportionality, as well as ensuring overall governance coherence rather than a wild west jungle. As also mentioned above, it is critical that any 'third generation' of EU multi-level governance does not

increase unnecessary bureaucracy but addresses inter- and intra-institutional relationships and is better able to achieve true transparency, accountability and inclusiveness in the entire system. The ultimate aim must also be increased trust between all the actors within the EU as well as interfacing in the best possible way with those outside. In this context, it is important to cater for the possible roles of all, including EU and international, actors so that the embedded regimes participate in co-creation in the most productive and inclusive manner possible. Again, this is hardly feasible without the appropriate use of digital technology.

7.4.4 Breaking down silos

A recurring theme as public governance paradigms change chronologically is the attempt to break-down the silos between governance levels, between different actors and within a given governance jurisdiction. This is necessary both to increase the number of public values of good governance (see Section 7.3.2) and **move the services that create public value from level 1, which focus largely on only individual benefits, to level 2 outcomes focusing on collective societal benefits** that can more successfully address societal challenges (see Section 2.3). The latter requires user-centric and personalised services that are networked, joined-up and integrated, first possible from about 2000, although success was extremely limited until some benefits were achieved in the context of the politics and policies underpinning the open governance paradigm. More significant success in breaking down silos and achieving level 2 more societal-wide public value first began, however, in the context of the changed politics and policies from about 2015 of the sustainability and especially the locality-community paradigms. Although digital technology innovations play a necessary role in these changes, the sufficient conditions are only forthcoming as the politics and policies change.

Thus, the silo problem remains today as one of the biggest challenges to public value creation and is not directly related to further advances in digital technology, although improvements in digital interoperability are absolutely necessary (see for example European Commission, 2022k). The policies and mindsets needed to move towards the human-centric focus on real people's lives, and on the 'emotional intelligence' promised by the current speculation around Generation 5.0 technology in the context of the 'digital sanitisation' involved in people and machines working together (see Section 5.1), will perhaps be more decisive in realising significant levels 2 and 3 public value.

8 Conclusions and recommendations

8.1 What are the determinants of changes in public governance

From the above narratives, supplemented by the accompanying state-of-the-art report on existing public governance paradigms, it is clear that **political**, **policy**, **socio-economic cultural conditions**, **especially including societal-wide shocks and crises**, **are the strongest determinants of public governance changes**. **Digital technology and data**, **on the other hand**, **are best seen as necessary but never sufficient tools in this process**. Nonetheless, it is certainly the case that most of the changes in public governance processes are dependent on digital technology in one way or another. Indeed, its role changes and magnifies from merely supportive during Generation 1.0, to enabling during Generation 2.0, then to driving in Generation 3.0 and finally to decisive in Generation 4.0, as summarised in Table 2. However, digital technology itself is largely dependent on these same political and socio-economic conditions with which it has a mutually supportive, reciprocal and complex relationship, so that it is not always possible, or indeed wise, to disentangle them and ascribe clear cause and effect roles to one or the other.

According to the literature review carried out by Barcevičius et al (2019), economic drivers on the demand side feature prominently for successful digital government transformation as the potential efficiency gains are of high importance to the public sector and to the taxpayers. On the supply side, the rapid technological developments and diffusion of electronic devices are identified as important drivers, even though they are not considered sufficient to transform governments on their own. Political, social and cultural factors receive a lot of attention, with a particular focus on the expectations of citizens for more transparency and participation in policy-making. Barcevičius et al (2019) also found that the barriers and preconditions for successful digital government transformation are complex and often not related to digital technology. In fact, the introduction of new technologies by governments is always mediated by organisational, institutional, legal, ethical and social barriers.

Overall, it is evident that all the successive public governance paradigms align to a greater or lesser extent with longer-term political and societal developments and shocks since 1945 (as shown in Table 3), as well as with digital technology developments (as shown in Table 2), especially around pivotal cleavage dates:

The early 1990s: The early beginnings of Generation 1.0 largely one-way digital technology use in public governance, supporting the traditional Weberian public governance paradigm, boosting New Public Management and then helping to underpin the Neo-Weberian paradigm.

About 2000: It began to be accepted that government could use digital technology and data to be the prime mover in proactively delivering public value benefits to wider society and not just to improve the administrative machinery in the back-office. At this time, digital technology first began to change the shape and workings of public governance, as opposed simply to being used by it. It is probably no coincidence that the 2000 cleavage date coincided with the significant jump from Generation 1.0 to Generation 2.0 technology enabling two-way participation. This facilitated both the networked and public value governance paradigms as two different political philosophies of society arising from the same societal and technological conditions.

From 2008: The financial crisis, itself triggered by the use of Generations 1.0 and 2.0 technology in the banking and finance sectors spurring globalisation, gave simultaneous rise to two quite different public governance paradigms. First, lean and austerity governance and, second, a new plethora of governance paradigms and models characterised by open governance. Both were significantly enabled by new Generation 3.0 semantic-based technology but based on quite different political philosophies of society.

About 2015: Serious moves began to break down governance silos enabled by the increasing digitisation of both government front- and back-offices. This was in the context of further increasing globalisation with its ensuing economic growth, but scarred by rising inequalities, dissatisfaction and loss of trust in government, coinciding with significant populism and 'post-truth' movements. Both were pushed and pulled by Generation 4.0 distributed and mobile technology. These developments are paired with a significant increase in the availability of data (including personal data) and challenges related to data gathering, capture, access and sharing.

From 2020: A new age of more or less continuous crisis, disruption and turbulence seems highly likely. Although the groundwork was arguably laid in the 2008 financial crisis, this was turbo-charged first by the COVID-19 pandemic commencing in 2020, then by the Russian invasion of Ukraine in 2022, as well as the even greater threats posed by the ever increasing environmental and climate crisis. New forms of public governance are undoubtedly required to meet these existential challenges, so it is also imperative that we design and implement more appropriate digital technologies. Whether the putative Generation 5.0 technologies and the Fifth Industrial Revolution, portending dramatically new relationships between humans and machines that aim to put people in control, will suffice remains to be seen (see Section 5.1).

8.2 Summary of insights from existing public governance paradigms

It is noteworthy that the types of public governance models and paradigms change in character and increase in variety over time. Chronologically, they change from a small number mainly concerned with process, administration and organisation, drawing their justification from an increasing array of 'good governance' public values, towards a much larger number with a clear public value focus that is directly conceived to address important societal challenges. This switch around the year 2000 coincides exactly with the big uptake by public governance of digital technology starting with Generation 2.0 interactive technology acting as a highly significant enabler, as well as to changing politics and policies.

From 2020, **a step change now seems to be taking place mainly influenced by a series of** turbulences that challenge previously established relationships and have major impacts on the geo-political landscape. These changes seem to be due to: (i) the huge variety of ways different countries and authorities react both politically and organisationally to their churning socio-economic, environmental, cultural and historical situations; (ii) the influence of external factors and crises; and (iii) the increasingly diverse possibilities enabled by innovative digital technology. This is so even in a Europe that has had great success over at least the last 15 years in coordinating and learning across countries, for example through the EU's eGovernment Action Plans (such as European Commission, 2016a), the 2020 Berlin Declaration on digital society and value-based digital government (European Commission, 2020f), the Lisbon Declaration of 2021 on digital democracy with a purpose (Portuguese Presidency, 2021) and the 2022 Strasburg Declaration on the common values and challenges of European Public Administrations underlining an overarching support of open source and interoperability within the public sector (Presidence Francaise, 2022).

8.2.1 Digital technology and public governance are intertwined

- Everything is inevitably influenced by digital technology, and it is better to explicitly account for that, than
 ignoring digital
- Yet, digital technology and digital data are best seen as necessary but never as sufficient tools for public governance. Their deployment is always mediated by organisational, institutional, legal, ethical and social conditions, as well as challenges such as digital exclusion.
- Considering the many and diverse interrelationships between all actors of public governance, digital data and technologies might be considered at the only possible way to reach transparency, accountability and inclusiveness.
- The complex roles of digital technology and human-technology relations need to take account of how peoples' quality of life, values and ethics are impacted by increasingly omnipresent machines, burgeoning surveillance by both public and private sectors (see, for example, Zuboff 2019), as well as the 'post-truth' society, 'black-box', impenetrable and inevitably biased AI.
- The mix of public governance paradigms in any one place and time, including across multi-governance levels, is an important contextual response to prevailing political, socio-economic and cultural differences.
 However, these responses and mixes may or may not be successful, so governing this mix is a crucial issue where the deployment of digital technology is necessary.
- To cope with the challenges of our times, we thus need public services that are digital-ready and interoperable by design – across borders, across sectors, and across different levels of administration.
- We are still in an era of understanding the best (context dependent) ways of using digital data and technologies for the public good, and a rich set of diverse experiments still yet to be completed or entirely conducted to complete an entire wave of digital transitions of public governance¹⁷.

8.2.2 Public governance evolving features and influencing factors

- Political, policy, socio-economic, environmental, historical and cultural factors, especially societal-wide and
 often international shocks and crises, are the strongest influences on public governance developments.
- The public governance paradigms sketched in Figure 1 accumulate layer upon layer resulting in a form of sedimentation over time. Each of the nine paradigms identified between 1945 and 2019 (each with one or more models) continue to have relevance today, although the more recent novel and emerging paradigms tend now to be more dominant but perhaps only because collectively they are more numerous.
- These public governance paradigms do not form a linear progression where one replaces the other but represent a process of co-evolution. According to Aristovnik et al (2022) "Due to the constantly changing environment, public governance models have transformed many times, creating differences in public governance practices among public administration institutions, with combinations of contradictory structures and principles that coexist." Differences can also be seen between central and local governance, and to elaborate on possible evolutions of the original approach especially as applied within the EU (within and across countries; see Section 7.4.3).
- It is the actual 'mix that matters' (Bevir, 2013) The mix of paradigms and models at a specific place and time depends on the context of global, national and local politics, as well as history, culture, socioeconomics, environmental factors and the political choices made. This mix may or may not be a successful arrangement, so governing this mix is a crucial issue where the deployment of digital technology is necessary.
- Co-creation, and thus generative and emergent governance paradigms, seem imminent. Both tacit and codified knowledge are a powerful combination for learning, identifying good practices of successful transformations, and thereby for widespread co-creation, replication, scaling and knowledge sharing. At least up to 2020 only promising examples are seen without a truly systemic approach at EU or national levels. It is not yet clear what a balanced approach, which applies representative methodologies to some societal challenges while investing in more participatory approaches in others, might look like. Revisiting

¹⁷ See for example, Kert, Vebrova and Schade (2022).

Ansell and Torfing (2021), these deliberations on co-creation might be further divided into co-initiation, codesign, co-implementation and co-evaluation, as a cyclic approach that departs from the traditional ladder analogy of Arnstein (1969) that tends to imply a hierarchy of activities.

— The most desirable re-balancing of power and responsibilities for future-proof public governance still needs innovation and experimentation so is yet to be found, as explored in Section 7.4.3. This will also require a cultural change within public institutions that tend to be risk averse. These risks are often seen in the short term (for example, but not only, connected to election cycles) but without considering the longer-term risks of not rebalancing power relations, and experimenting with new approaches to public governance.

8.3 Overall conclusions

The overall purpose of public governance as addressed in this report is to promote the prosperity, wellbeing and flourishing of all people and all parts of society, as enabled by nature's life-support systems. The focus is on the EU but also with a broader international perspective. The setting is (liberal) democratic systems, with their country, regional and local variations and the EU as the overarching level of governance. In order to promote such flourishing, this report maps both current and near-future societal challenges which are growing in complexity as global, European and more local needs and shocks pile up especially since 2020, and examines the public governance paradigms constructed to tackle these. The report first looks retrospectively in order to better understand why and how public governance across the EU has addressed this overall purpose since 1945 and especially in the last fifteen years since the 2008 economic and financial crisis. It then examines Europe's new strategic challenges in the age of turbulence since 2020, the public value benefits that can accrue if these are successfully addressed, and proposes the main public values needed to make this happen.

There is an immutable number of outstanding features of this succession of public governance approaches:

- It is clear that successive public governance paradigms, and their constituent models, have cumulatively been determined, as well as partially shaped, by the prevailing political, socio-economic, environmental and cultural factors, as well as how they have responded to societal-wide and international shocks. This has led to a sedimentation of public governance paradigms where all those identified are either still in play, or potentially and frequently having effects.
- 2. The creation of **an 'ecosystem' of public governance approaches** often differentially characterises each administration, multi-governance level and combination of government and non-government actors. Over time there has been an increase in the number and type of non-government actors in each ecosystem in directly addressing societal challenges, as well as in power- and role-sharing at all levels.
- 3. It will **only be possible to govern these ecosystems**, and knit them together across levels and geographies, **by deploying digital technology and data** alongside and in support of human organisations and systems. Digital technology is an essential component in enabling society to actually implement such complex and heterogeneous governance models. Thus, the technology is 100% necessary for understanding both the current status of public governance and how we can successfully grapple with both the current and new societal challenges in order to promote human flourishing, although by itself it is never sufficient.
- 4. Overall, both the strategic challenges facing public governance from 2020 onward (Section 4) and the example sectors (Section 6) show very strong trends towards localised relationships and transactions with significant human and physical presence but supported by Generation 5.0 technologies (Section 5.1.1) and Industry 5.0 (Section 5.1.2). At the larger scales, and especially internationally and globally, relationships and transactions are becoming relatively more digital than at local scales. Data from everywhere are increasingly imported into localities to help drive these localised human and physical relationships and transactions. In turn, data from these localities are exported everywhere else to drive scaled digital relationships and transactions, which is then distributed back to localities everywhere. Figure 4 shows this in the food sector.
- 5. Despite the danger of over-simplification, there does seem to be some **distinction between 'digital 'light' public governance models**, where the technology supports the model and perhaps changes the model but isn't necessary for its basic premise, **and 'digital heavy' models** which could not exist without the technology. There are many examples of both in this report. 'Digital light' examples include many of the local relationships and interactions in localities as mentioned above, such as the preventative public services in Wigan in the UK (see Section 4.3.1) and flexi-security labour market model in Scandinavia (see

Section 6.3.3). 'Digital heavy' examples include POLIS democracy software (see Section 7.3.2.1) and New Zealand's introduction of machine-readable rules and laws (see Section 7.3.2.10).

- 6. Looking at the impacts of the crises hitting Europe since 2020, including COVID-19 and the war in Ukraine, on the three example sectors (Section 6), it is relatively easy to trace and describe the pre-crisis and the current crisis situations, but what is unclear are the details of the period starting in 2023. However, it is certain this will be very different from the pre-crisis situation, so there is no going back to 'normality'. There is instead a claimed 'new normal' characterised by continuous turbulences. At the same time, there is a strong likelihood that many, though not all, of the trends and contours of the near future can already be seen in the current crisis situation, as illustrated for example in the work and employment example in Section 6.3.
- 7. Looking in more detail at impacts of the crisis-induced turbulence on the three example sectors (Section 6), both **the food and energy sectors sharply accelerated longer-term, but previously relatively weak, trends**. These include a move towards shorter value chains, more local supply and consumption and more de-globalisation and these are becoming an important part of the future ('new normal?). This is accompanied, however, by increasing activity towards even greater European cooperation. Behaviour and attitudes have generally changed in parallel with these developments. It seems there is some certainty that these trends will continue, probably unabated, assuming the current crises of geo-political turbulence and environmental degradation continue, as appears likely.

Work and employment have arguably experienced the greatest shock and transformation of any aspect of the economy as a result of COVID-19 which has turned many of the pre-pandemic concerns like mass unemployment, outlined in Section 6.3.1, on their head. However, these are unlikely to remain a problem in the background for the long-term, especially given that the labour substitution potential of digital technologies is likely to increase even further in near future. There has been very significant work localisation, but mainly through tele- and hybrid-work so economic localisation has only happened to lesser extent. in fact, this is marked more by some de-centralisation of economic activity down the city hierarchy, from very large to smaller cities, whilst there continues to be some centralisation from rural to urban areas generally. The future of work and employment is thus more uncertain than for food and energy and, although behaviour and attitudes have also generally changed in parallel, these seem more unpredictable going forwards. However, for the foreseeable future, these present COVID and war triggered trends seem likely to gather pace, assuming the current crises of geo-political turbulence and environmental degradation continue, as appears likely.

- 8. In this context there is now much **stronger recognition of the critical role of communities and localities in delivering prosperity and welfare** and thus the importance of giving them much more agency and autonomy, compatible with wider governance structures. Although there are risks of a postcode lottery which need to be addressed, such localisation can contribute massively to resilience through the diversity, interconnectivity and experimentation they foster. The research behind this report has improved understanding of communities of interest and of practice, as well as of how people identify themselves with these communities and how values and their perception change. For example, the change from a purely economically-determined focus to a greater role for social and cultural expectations of being heard, being able to participate, and being able to rely on public institutions. (These issues are examined in more detail in the accompanying report on the community perspective.)
- 9. It is increasingly accepted that **nature (both organic and non-organic)**, **as the basis of all human existence**, **is given pride of place in its own right at the decision- and policy-making table**, **and thereby must be centrally incorporated into all public governance paradigms**. If the helix analogy of societal actors is to have any credibility, it must first move from the triple to the quadruple helix by including community and civil society. Further, it is essential that it also moves to the notion of the quintuple helix with nature as the fifth actor if not, in fact, the policy-making table itself. (See Section 4.4.1.)
- 10. There is a growing focus on governance paradigms that are **value-based**, **human-centric**, **culturally-and behaviourally-sensitive**, that take full account of **variations in values and identities** (Scharfbillig et al, 2021) and that deploy increasingly multi-actor (quintuple helix) approaches.
- 11. It is also clear that the mix of both individual and ecosystems of public governance paradigms have all displayed varying degrees of success and failure. In fact, given the complexity and rapid change of society, this is inevitable and is also seen where, from the same set of technologies, societal

conditions and shocks, different political approaches produce starkly contrasting public governance paradigms and impacts. Politics and their decisions and policies make all the difference.

- 12. Despite many setbacks, undoubted 'progress' has been made since 1945 in improving societal prosperity and wellbeing, but this is now being stalled and even reversed by the huge shocks of the past few years for which most governance systems were ill-prepared. Living with this turbulence and, indeed, still improving prosperity and wellbeing while it continues, requires much awareness and honest learning. It requires the application of lessons about what works and what does not from both successes and failures which should be dispassionately delivered but have instead often been forgotten or ignored. (McKinsey, 2017)
- 13. Despite the significant de-globalisation currently taking place, the world nevertheless continues to become more interconnected, in fact hyper-connected. This lack of learning is having increasingly detrimental effects over wider and wider areas and contexts, and this threatens to become more damaging in future. Politically-inspired transparency of vision, mission and far-sightedness are valuable ingredients to add to the recipe of learning and experimentation, but political dogma and blindness, of whatever hue, are the greatest threat to both, and thus to prosperity and wellbeing on a healthy planet
- 14. Within the setting scoped by the above, this report has especially investigated **the role of digital technology and data**, given that both are unelidable constituents of our daily lives and also have a necessary role in driving change. It is critical that we understand which changes are desirable, which are not, and to provide scientific advice to support the required policy actions to improve prosperity and welfare going forward, despite the surrounding turbulence.

In the light of all the above, one of the important future activities is to learn from the results we already have, to foster a rich set of diverse experiments and to enable uptake and adoption of value-based, human-centric and co-created approaches in contexts where this makes sense.

8.4 Recommendations for action and research on present and future European public governance

The recent crises have already set back decades of public governance progress that had earlier, and despite much unevenness, led overall to the greater prosperity, wellbeing and flourishing of most people in most parts of society, and that had begun to make some progress in addressing the vulnerability of nature's life-support systems. At the same time, however, the crises since 2020 also provide opportunities to rethink how public governance and the role of digital technology can assist Europe in getting back on track **European societies are ready for new ways of thinking and accepting bold policy and public governance changes, but the next five years are critical to redirect the course of development.** We are at a critical juncture to take action if we are to realise a socially just transition towards new paradigms that again increase societal-wide flourishing. This research has identified four strategic challenges that now require urgent public governance action and research, although this is not to imply that there are no other challenges likely to arise (see Section 4):

Living with turbulence: Public governance needs to accommodate shocks and crises by moving to more resilient paradigms that move away from traditionally-determined efficiency criteria by anticipating and mitigating turbulence by prioritising diversity and interconnectivity, and reject dogmas through free and informed criticism. The goal is to be equipped to react better, more nimbly and more timely to unexpected events as well as longer term risks, including environmental breakdown, whilst still providing as much continuity, stability and predictably to society as possible. This should be the case even when it decreases some short-term efficiencies given the clear evidence of the societal damage prevented, including the improved long-term efficiencies that can instead be realised as, ultimately, this is the mandate of public sector governance. There should be increased focus on the EU's values and open strategic autonomy in essential policies, goods and services choice, also through multi-governance levels where beneficial. Cyber threats and other known risks also need continuing focus, as does seeing resilience in social and economic, as well as environmental terms, given these are all intimately interrelated.

Changing geographies: Changes to Europe's geo-political position have been turbo-charged by the recent turbulence creating new global realignments that are reshaping the global geo-political landscape economic and security arrangements, and their political manifestations. New EU public governance at all levels needs to draw upon and strengthen the EU's values and principles if these changes are to be successfully addressed. Sub-national level trends include the growing importance of small cities and larger towns within the hinterlands

of very large cities which, sometimes excepting capital cities, are themselves decreasing in importance driven by digital work, learning, shopping, etc. There is also a very strong resurgence of local economies and identities linked to efforts to retain as much as possible of locally-created value within the locality. This is seen, for example, by residents becoming more aware of their locality and community as well as the economic benefits of local production and consumption, with local governments focusing much more on local democracy, partnerships and procurement. Urban, rural and community-level geographies are also changing in directions that both accelerate existing trends and carve out new pathways, especially ones that prioritise decentralisation within strong national and EU structures, in order to ensure just transitions for all wherever they live.

Hybrid human-centred relationships: Traditional welfare state public services, first designed in the early post-1945 period, are no longer fit for purpose. The new age of turbulence, coupled with rapidly changing demographics towards greater cultural and ethnic diversity and an increasingly aged population, require public services that prioritise the prevention or slowing of social and economic misfortunes affecting individuals and communities especially manifest at local level and through community-based solutions. Such a shift can be both economically efficient as well as socially beneficial and, especially when meeting the needs of the most vulnerable, requires strong human relationships alongside hybrid service delivery mechanisms to be successful Resilient structures across all multi-governance levels are necessary to delivery such solutions with a focus on justice, fairness and inclusion by design.

People-planet systems: Arguably the greatest challenge that public governance has arises from the scientific facts that climate change, biodiversity loss and other stresses on nature are having profound deleterious impacts on the functions of societies and on the lives of their inhabitants. The needs of a flourishing society are bound together with a flourishing nature, both organic and inorganic, so nature must be centrally incorporated into all public governance paradigms. This should involve moving on, from a concept of sustainability that aims only to keep what we currently have through conversation and preservation, and instead aims to regenerate what we really need. This will also involve public governance that better understands the complex dynamic systems that intricately link human societies with natural systems.

The above strategic challenges, if successfully addressed by public governance, will provide highly significant **public value benefits** across all European societies. As originally identified by Kelly et al (2002), such public value can be evaluated both in terms of level 1 services, focusing largely on only individual benefits, as well as on level 2 outcomes focusing on collective societal benefits that can more successfully address societal challenges (see Section 2.3). In this context, a future focus of research might also address the semantic as well as a difference in practice between what the public 'wants' in terms of benefits compared to what the public 'needs'. 'Wants' clearly reflect the consciously expressed preferences of the public, whether as individuals or collectively, which may be contradictory, unethical or illegal, and tend to be derived bottom-up through expressions of agency. In contrast, 'needs' require some more objective definition of what is necessary for an acceptable standard of living or a good fulfilling life, which implies some structurally imposed decision-making or assessment process that may or may not be democratically driven. A good example of such a needs-based approach are the definitions and distinctions between 'absolute' and 'relative' poverty and how to escape from them (endPoverty, 2023).

This also requires further action and research into the **European public values and principles** as the means necessary to deliver these public value benefits. Over the last few years there has been a great push on valuebased governance through, for example, the Berlin Declaration (European Commission, 2020f) that takes the user-centricity principles a step further by strengthening the pioneering role of public administrations in driving a value-based digital transformation of European societies, and on digital rights and principles (European Commission, 2022l) that presents the EU's commitment to a secure, safe and sustainable digital transformation that puts people at the centre, in line with EU core values and fundamental rights. There is a clear need to build on these initiatives in the years to come. In support of this, as outlined in Section 7.3.2, this research has outlined ten key public values necessary for designing the new public governance needed in the age of turbulence and that build on and/or supplement those prior to 2020. These are: democracy and power relations; trust and trustworthiness; risk governance and crisis management; regulation; public sector innovation, experimentation and learning: behaviour, values, principles and culture: new innovation frameworks and mission strategies: public governance management and integrity; future proofing; and policy coherence, coordination, alignment and evaluation. These ten are likely to be insufficient to meet future public governance challenges, so on-going research will also be needed to identify how they should develop in order to optimise European public value and, specifically, address the four main strategic challenges identified above.

Action and research are also critically needed to develop and deploy the 'third generation' of EU multi-level governance systems that might be characterised as one that addresses inter- and intra-institutional

relationships, successions to achieve true transparency, accountability and inclusiveness in the entire system. also by making the most appropriate use of digital technology, ultimately aiming at increased trust between all the actors within the EU and interfacing in the best possible way with those outside (see Section 7.4.3). It also specifically moves towards various forms of 'generative public governance', i.e. that are sufficiently flexible and resilient to enable specific emergent governance forms to be created in particular contexts for particular purposes such as unexpected shocks The main mechanism behind generative public governance is co-creation that both builds on and extends the concept of collaborative governance and enables a strategic turn aimed at solving complex and sudden problems. However, there are also dangers in adding additional layers and complexity to public governance systems composed of multiple levels and forms, especially as seen from the perspective of citizens and other actors, so that a joined-up, whole-of-governance external face is definitely required. Such complexity, especially when locally-embedded and nuanced, multiplies the numbers of actors and relationships and potentially adds to the 'silo' problem. As pointed out in Section 4.2.3, however, it is often at the local level that it is easiest to break down governance silos and link across actors and sectors given their smaller and shorter interaction and transaction chains and much more shared and fine-grained knowledge, both implicit and explicit, about local conditions and contexts. None of this is hardly feasible without the appropriate use of digital technology, although the sufficient conditions for this are only forthcoming as the politics and policies change.

Overall, it is clear that it is the precise mix that matters in any given situation, so that co-creative contributions are allowed, supported and empowered to come from anywhere in the system. However, this must focus on the challenges that are best addressed by those actors taking responsibility, possibly as an advanced interpretation of subsidiarity and proportionality, as well as ensuring overall governance coherence rather than a wild west jungle.

8.5 Recommendations for action and research on the ongoing deployment and development of digital technology and data

The research outlined in this report has unambiguously shown that political, policy, socio-economic, environmental, historical and cultural factors, especially societal-wide and often international shocks and crises, are the strongest influences on public governance developments. However, **since the early 1990s such developments have also been decisively shaped and enabled by digital technology and data that have become necessary, but not sufficient conditions for the changes seen**. Moreover, it is clearly the case that the strategic challenges outlined above cannot be successfully addressed without digital technology and that it will not be possible to govern the public governance that now needed without it.

Strong research focus is thus needed on designing **human-centred digital technology** that focuses on real people and real peoples' lives where the human, human-touch and human control are more important than the digital, but where public value can be enhanced by the digital. Digital 'sanitisation', in the sense of audits that reduce or remove the digital where this does not add value but instead increases digital divides, should take place. This should also be the focus even where such reappraisals decrease some short-term efficiencies given the clear evidence of the societal damage prevented, including the improved long-term efficiencies that can be realised. This is how the public sector fulfils its mandate of creating value, through making a balanced, just, lawful human and nature centred contribution to public governance.

Such human-centred (and also human-nature) approaches to digital technology should also be the focus of research into the putative **government Generation 5.0 technologies** (see Section 5.1.1) that cumulatively build on earlier generations. This should progress in parallel with the anticipation of **Industrial Revolution 5.0**, which also cumulatively builds on earlier revolutions, and envisages humans and machines performing work hand-in-hand. Combining the unique cognitive abilities of workers and the accurate technical expertise of, for example, robots can better promote an innovative and productive culture in the workforce (see Section 5.1.2). A specific manifestation of such an interrelationship between the human and the digital is illustrated by the beginning of changes to food systems where the local remains best served by human and physical relationships and transactions, which in turn are re-distributed to localities everywhere. This is also an example of increasing food resilience through nature-intrinsic diversity plus interconnectivity (see Section 6.1.1).

Interconnectivity can also be enhanced by research supporting **interoperability by design** across borders, across actors, across organisations and across sectors and, above all, across Europe's multi-governance levels. In a European context, this is likely to be taken further forward by the December 2022 proposal for the **Interoperable Europe Act** to introduce a framework of cooperation between EU Member States and EU

Institutions. The aim is to secure the cross-border exchange of data and agree on shared digital solutions, such as open-source software, guidelines and frameworks, and will provide incentives for innovation in the public sector and for "GovTech" projects, based on public-private partnerships (European Commission, 2022k). Ongoing research is also needed into understanding the best (context dependent) ways of using digital data and technologies for the governance of society and the public good. This **requires a rich set of diverse experiments** that are ongoing and continuous to ensure that successive digital transitions of public governance support these goals. Data is a key resource in the digital economy, and control over the way it is generated, aggregated, value is extracted and equitably distributed in society is crucial. In particular, the rise of AI and IoT offer new opportunities for policy design, implementation, and assessment, as well as for providing more personalised support to those who need it and for being more participative throughout the policy cycle.

In this context, the **2022 European Data Governance Act**, based on EU values and principles are designed to bring significant benefits to EU citizens and companies by increasing trust in data sharing, strengthening mechanisms to increase data availability and overcoming technical obstacles to the reuse of data (European Commission, 2022a), needs ongoing research. The focus should be on ensuring this initiative can be a powerful engine for innovation and new jobs allowing the EU to ensure that it is at the forefront of the 'second wave of innovation based on data'.

In terms of digital governance, in addition to algorithmic bias, competition/anti-trust, privacy and surveillance, **important values such as human rights, the rule of law, trust and transparency, are critical to guide digital governance**. Ensuring data privacy and countering misinformation is essential for building trust and trustworthiness, and to facilitate technology adoption. In order to democratise digital governance, a multi-actor social dialogue on what the moral and political obligations are, as well as mechanisms to facilitate them, will be crucial to combat both socio-economic and digital inequality. Democratising technology means that the control of technology should ultimately rest with all relevant actors, and that technology should be transparent and held accountable. Digital governance systems will need to deploy tools to address the unequal distribution of income, wealth and control of technologies.

In particular, **Artificial Intelligence (AI) is revolutionising the way we live, work and govern**. There is a need to better forecast possible scenarios of how these technologies could be used, the skills they might require, and the policies and regulatory frameworks needed at the national and international levels. There is a pressing need to ensure transparency, trust and accountability in how technologies and data are managed and used. Opacity in relation to the collection and use of data threaten personal freedoms and individual rights. Algorithmic biases threaten to widen inequalities. Discriminative data collection (unintended or intended) can harm vulnerable populations without access to proper recourse mechanisms. Currently, these mostly take years and are often out of reach of many people, so new independent and effective institutions that directly represent ordinary people subject to AI decisions should be set up, for example in the form of a digital ombudsman. This is important as **AI is qualitatively different from most other technologies as it is often difficult to understand why and how it makes decisions**, even for the developers themselves and certainly by most of those applying it. Technological design can exclude people with limited connectivity or hardware access. These risks call for a transparent and robust regulatory framework with accountability mechanisms that can assess the impact of technology on people over their life cycle.

An indication of how fast AI is developing and how great the potential impact could be, is the launch of the *ChatGPT* in November 2022¹⁸, since upgraded several times and joined by competitors in a global race. By drawing on the vast amounts of data across the internet, so-called 'generative AI' can produce human-like conversations on a wide range of topics by generating text, images, music, speech, code or video that is increasingly difficult to distinguish from that made by humans alone. These developments are *significantly impacting European* society and accelerating the digital transformation of various industries, such as transportation and healthcare. *While the increasing adoption of AI can bring a vast range of potential benefits, it also presents new challenges and risks that need to be addressed*, not least disinformation and misinformation, large scale threats to jobs as we know them, and significantly greater threats arising from data bias that is increasingly difficult to detect. In response, the European Commission (2023) has put forward an ambitious strategy aimed at fostering AI excellence in Europe and ensuring that AI systems are trustworthy and protect the safety and fundamental rights of European citizens. Ongoing research and action in this area are paramount.

¹⁸ <u>https://openai.com/blog/chatqpt</u>

Finally, **contextual technology assessment** is necessary prior to adoption (e.g. through sandboxing and experimentation on real world conditions), also by incorporating measures for transparency, for example, periods for public comment, and consulting all relevant actors during the problem framing and development process. Establishing robust procurement procedures for technologies supplied by private corporations for the purpose of public governance can increase trust and accountability by using *specific contractual clauses and audits to address 'black box' proprietary systems*. Inclusive assessment procedures help answer questions such as: "What is the identified problem?", "Is technology the best solution and use of resources?", "Is this technology sustainable?", "Who receives the benefits and who is harmed?", and "What accountability procedures need to be in place to grant its lawful, just, and respectful use in compliance with human and nature rights?".

It is undisputable that digital technologies may transform virtually every process, system and structure of government and of public governance – resulting in redefinition of power, roles, responsibilities, and effect on all aspect of life. Nevertheless, they also create issues and trade-offs that merit careful consideration and preparation before a full-blown change is introduced.

References

Acaroglu, L. (2022). Is it Time to Ditch Sustainability?, Medium. (<u>https://medium.com/disruptive-design/should-we-ditch-sustainability-5420b417ed8)</u>.

Adamovic, M., Bisselink, B., De Felice, M., De Roo, A., Dorati, C., Ganora, D., Medarac, H., Pistocchi, A., Van De Bund, W. and Vanham, D. (2019). Water--Energy Nexus in Europe, Magagna, D., Bidoglio, G., Hidalgo Gonzalez, I. and Peteves, E. editor(s), EUR 29743 EN, Publications Office of the European Union, Luxembourg, 2019, ISBN 978-92-76-03386-8, doi:10.2760/285180, JRC115853.

Agre, P.E. (2000). Information and Institutional Change, Communication Studies, Spring 2000.

Aleksandrzak, K. (2019). Urbanization – components and stages of urbanization, i.e. the city's life cycle, 24 October 2019. (<u>https://geographicforall.com/en/urbanization-components-and-stages-of-urbanization-i-e-the-citys-life-cycle</u>).

Albers R.A.W., Bosch P.R., Blocken B., Dobbelsteen A.A.J.F., van Hove L.W.A., Spit T.J.M., van de Ven F., van Hooff T. and Rovers V. (2015). Overview of challenges and achievements in the climate adaptation of cities and in the Climate Proof Cities program, *Building and Environment*, , 83, 1-10. DOI: 10.1016/j.buildenv.2014.09.006.

Alberti, V., Alonso Raposo, M., Attardo, C., et al. (2019). The Future of Cities, Vandecasteele, I., Baranzelli, C, Siragusa, A. and Aurambout, J. editor(s), EUR29752 EN, Publications Office of the European Union, Luxembourg, ISBN 978-92-76-03848-1, doi:10.2760/364135, JRC116711.

Al-Hujra, O., Al-dalahmeh, M. and Aloudat, A. (2011). The Role of National Culture on Citizen Adoption of eGovernment Services: An Empirical Study. *Electronic Journal of e-Government*. Volume 9 Issue 2 2011, (pp93 - 106).

Aljazeera (2022). World Food Programme: 20 million risk starvation as Horn of Africa drought worsens. (<u>https://www.aljazeera.com/news/2022/4/19/20-million-risk-starvation-as-horn-of-africa-drought-worsens-un</u>).

Alonso Raposo, M., Mourtzouchou, A., Garus, A., Brinkhoff-Button, N., Kert, K. and Ciuffo, B. (2021). JRC Future Mobility Solutions Living Lab (FMS-Lab): conceptual framework, state of play and way forward, EUR 30906 EN, Publications Office of the European Union, Luxembourg, ISBN 978-92-76-44741-2, doi:10.2760/999349, JRC127272.

Alston, P. (2020). The parlous state of poverty eradication -- Report of the Special Rapporteur on extreme poverty and human rights. United Nations Human Rights Council. (<u>https://chrgj.org/wp-content/uploads/2020/07/Alston-Poverty-Report-FINAL.pdf</u>).

Anderson, C. (2012). Makers: the new industrial revolution. New York: Random House.

Annoni, A., Benczur, P., Bertoldi, P., Delipetrev, B., De Prato, G., Feijoo, C., Fernandez Macias, E., Gomez Gutierrez, E., Iglesias Portela, M., Junklewitz, H., Lopez Cobo, M., Martens, B., Figueiredo Do Nascimento, S., Nativi, S, Polvora, A., Sanchez Martin, J.I., Tolan, S., Tuomi, I. and Vesnic Alujevic, L. (2018). Artificial Intelligence: A European Perspective, Craglia, M. editor(s), EUR 29425 EN, Publications Office of the European Union, Luxembourg, 2018, ISBN 978-92-79-97219-5, doi:10.2760/936974, JRC113826.

Ansell, C., and Torfing, J. (2021). *Public Governance as Co-creation: A Strategy for Revitalizing the Public Sector and Rejuvenating Democracy*. (1 udg.) Cambridge University Press. Cambridge Studies in Comparative Public Policy, <u>https://doi.org/10.1017/9781108765381</u>.

Aristovnik, A., Murko, E. and Ravšelj, D. (2022). From Neo-Weberian to HybridGovernance Models in Public Administration: Differences between State and Local Self-Government. *Administrative Sciences* 12: 26.

Arnstein, S. R. (1969). A ladder of citizen participation. *Journal of the American Institute of planners*, 35(4), 216-224.

Arthur, W. B. (2014). All Systems will be Gamed: Exploitive Behavior in Economic and Social Systems. Santa Fe Institute Working Paper.

Avelino, F., Bosman, R., Frantzeskaki, N., Akerboom, S., Boontje, P., Hoffman, J. and Wittmayer, J. (2014). The (self-)governance of community energy: Challenges & prospects (DRIFT practice brief no. PB 2014.01). Rotterdam, the Netherlands: Dutch Research Institute for Transitions.

Avelino, F. and Wittmayer, J.M. (2016). Shifting power relations in sustainability transitions: a multi-actor perspective. *Journal of Environmental Policy & Planning*, 2016. 18(5): p. 628–649: <u>http://doi.org/10.1080/1523908X.2015.1112259</u>.

Ayrshire and Arran NHS (2019). Community Wealth Building, putting the interest of people, place and planet at the heart of what we do. (<u>https://www.nhsaaa.net/services-a-z/community-wealth-building</u>).

Baldi, L., Bertoni, D., Migliore, G. and Massimo, P (2019) How alternative food networks work in a metropolitan area? An analysis of Solidarity Purchase Groups in Northern Italy. *Agricultural and Food Economics* 7(1), <u>https://doi.org/10.1186/s40100-019-0139-3</u>.

Baldini, G., Barrero, J., Chaudron, S. et al. (2020). Cybersecurity, our digital anchor, Nai Fovino, I., Barry, G, Chaudron, S., Coisel, I., Dewar, M., Junklewitz, H., Kampourakis, G., Kounelis, I., Mortara, B., Nordvik, J. and Sanchez Martin, J. editor(s), EUR 30276 EN, Publications Office of the European Union, Luxembourg, ISBN 978-92-76-19958-8, doi:10.2760/967437, JRC121051.

BBC Analysis (2022). The case for public service reform – is there a better way to deliver public services, podcast, 28 February 2022. (<u>https://www.bbc.co.uk/programmes/m0014x7v</u>).

Bulakovskiy M. (2021). Building Local Ecosystems for Social Innovation. A Methodological Framework, OECD Local Employment and Economic Development (LEED) Papers.

Bamburg, J. (2017). Mondragon through a Critical Lens: Ten Lessons from a visit to the Basque Cooperative Confederation. *Fifty by Fifty: Employee Ownership News*, 3 October 2017. (<u>https://medium.com/fifty-by-fifty/mondragon-through-a-critical-lens-b29de8c6049</u>).

Barcevičius, E., Cibaitė, G., Codagnone, C., Gineikytė, V., Klimavičiūtė, L., Liva, G., Matulevič, L., Misuraca, G. and Vanini, I.(2019) Exploring Digital Government transformation in the EU - Analysis of the state of the art and review of literature, EUR 29987 EN, Publications Office of the European Union, Luxembourg, 2019, ISBN 978-92-76-13299-8, doi:10.2760/17207, JRC118857.

Bason, C. (2010). Leading public sector innovation: co-creating for a better society, the Policy Press, Bristol, UK

Bauwens, T., Hekkert, M. and, Kirchherr, J. (2020). Circular futures: What Will They Look Like? Ecol. Econ. 175, 1–14. <u>https://doi.org/10.1016/j.ecolecon.2020.106703</u>.

BBC (2022). Ukraine war: World Bank warns of 'human catastrophe' food crisis (<u>https://www.bbc.com/news/business-61171529).</u>

Behavioural Insights Team (2020). A menu for change: using behavioural science to promote sustainable diets around the world: https://www.bi.team/wp-content/uploads/2020/03/BIT_Report_A-Menu-for-Change_Webversion_2020.pdf.pdf

Benington, J. and Moore, M.H. (eds.) (2011). Public value in complex and changing times. In Public Value: Theory and Practice. Basingstoke: Palgrave Macmillan, pp. 1–30.

Benyus, J.M. (2016). A Biomimicry Primer -- Biomimicry 3.8, Resource Handbook, Biomimicry.net & AskNature.org. (<u>https://biomimicry.net/b38files/A Biomimicry Primer Janine Benyus.pdf</u>).

BEPA -- Bureau of European Policy Advisers (2010). Empowering people, driving change. Social innovation in the European Union. Luxembourg: Publications Office of the European Union.

Berti Suman, A., Micheli, M., Ponti, M., and Craglia, M. (2020). Emerging models of data governance in the age of datafication. Big Data & Society, 7(2), 1-15. <u>https://doi.org/10.1177/2053951720948087</u>.

Bevir, M. (2013). A theory of governance. University of California Press, 2013.

Bharosa, N., De Winne, C.P.I., Van Wijk, R. and Janssen, M.F.W.H.A. (2012). Lean government: Critical success factors for XBRL-based business-to-government reporting, *European Journal of ePractice*, (18) 2012. (https://repository.tudelft.nl/islandora/object/uuid%3A764fbebc-009d-4632-a058-c2160ef3d1ee).

Boffey, D. (2019). 'Veggie discs' to replace veggie burgers in EU crackdown on food labels, *Guardian Newspaper*, 4 April 2019. (<u>https://www.theguardian.com/food/2019/apr/04/eu-to-ban-non-meat-product-labels-veggie-burgers-and-vegan-steaks</u>).

Bonime-Blanc, A. (2022). The Invasion of Ukraine Turbocharges the Need for Effective ESG and Resilience Governance, National Association of Corporate Directors, March 15, 2022. (https://blog.nacdonline.org/posts/turbocharges-esg-resilience-governance).

Börzel, T.A., Karen Heard-Lauréote, K. (2009). Networks in EU Multi-Level Governance: Concepts and Contributions. *Journal of Public Policy*, Vol. 29, No. 2 (Aug. 2009), Cambridge University Press.

Bragdon, J.H. (2016). Companies that mimic life, Greenleaf Publishing, Saltaire, UK.

British Library (2010). The Big Society online: harnessing technology for social change. Social Welfare Portal, 29 December 2010. (<u>https://www.bl.uk/collection-items/big-society-online-harnessing-technology-for-social-change</u>).

Brown, F.Z. (2022). Governance for Resilience: How Can States Prepare for the Next Crisis?, Carnegie Endowment for International Peace, Working Paper, May 2022. (<u>https://carnegieendowment.org/files/</u> Brown Governance for Resilience final.pdf).

Brundtland Commission Report (1987). Our common future: report of the World Commission on Environment and Development, United Nations, New York.

Bryson, J.M., Crosby, B.C., and Bloomberg. L. (2014). Public value governance: Moving beyond traditional public administration and the new public management. *Public Administration Review* 74: 445–56.

Bulakovskiy M. (2021). Building Local Ecosystems for Social Innovation. A Methodological Framework, OECD Local Employment and Economic Development (LEED) Papers.

Bulley, D., and Sokhi-Bulley, B. (2014). Big Society as Big Government: Cameron's Governmentality Agenda. British Journal of Politics and International Relations, 16(13), 452-470. <u>https://doi.org/10.1111/j.1467-856X.2012.00547.x</u>.

Butler, S. (2021). Uber drivers entitled to workers' rights, UK supreme court rules. *Guardian Newspaper*, 19 February 2021. (<u>https://www.theguardian.com/technology/2021/feb/19/uber-drivers-workers-uk-supreme-court-rules-rights</u>).

ByBit (2022). Web 5.0: What It Is & Why It Will Be Better Than Web 3.0, Jun 23, 2022. (<u>https://learn.bybit.com/blockchain/what-is-web-5</u>).

Cagnin, C., Muench, S., Scapolo, F., Stoermer, E. and Vesnic Alujevic, L. (2021). Shaping and securing the EU's Open Strategic Autonomy by 2040 and beyond. EUR 30802 EN, Publications Office of the European Union, Luxembourg, 2021, ISBN 978-92-76-41021-8, doi:10.2760/727114, JRC125994.

Carayannis, E.G., Barth, T.D. and Campbell, D.F. (2012). The Quintuple Helix innovation model: global warming as a challenge and driver for innovation. *Journal of Innovation and Entrepreneurship*, 1(1), 1-12.

Carnegie Europe (2022). Geopolitical Europe, by Richard Youngs, 28-07-22. (<u>https://carnegieeurope.eu/2022/07/28/awakening-of-geopolitical-europe-pub-87580</u>).

Castells, M. (2005). Network Society: From knowledge to Policy. Washington, DC: Johns Hopkins Center for Transatlantic Relations.

Castro-Arceab, K. and Vanclay, F. (2020). Transformative social innovation for sustainable rural development, *Journal of Rural Studies*, Volume 74, February 2020, 45-54.

Cattivelli, V., and Rusciano, V. (2020). Social Innovation and Food Provisioning during Covid-19: The Case of Urban-Rural Initiatives in the Province of Naples. Sustainability 2020, Vol. 12(11): 4444, https://doi.org/10.3390/SU12114444.

Ceccato, R., Baldassa, A., Rossi, R., and Gastaldia, M. (2022). Potential long-term effects of Covid-19 on telecommuting and environment: An Italian case-study, *Transp Res D Transp Environ*. 2022 Aug; 109. (<u>https://www.ncbi.nlm.nih.gov/pmc/articles/PMC9355418</u>).

Centola, D., Becker, J., Brackbill, D. and Baronchelli, A. (2018). Experimental evidence for tipping points in social convention, *Science*, 8 Jun 2018, Vol 360, Issue 6393, pp. 1116-1119: doi:10.1126/science.aas8827.

Centre for European Reform (2022). The Geopolitical EU. (<u>https://www.cer.eu/hot-topics/geopolitical-eu</u>).

Chantillon, M. (2021). A governance framework facilitating a digital transformation of the public administration. PhD Thesis. KU Leuven, Belgium. (<u>https://lirias.kuleuven.be/retrieve/637247</u>).

Chesbrough, H.W. (2003). Open Innovation: The new imperative for creating and profiting from technology. Harvard Business School Press, Boston. Cisco. (2007). The Connected Republic 2.0 -- New Possibilities & New Value for the Public Sector: A Point of View from the Cisco Internet Business Solutions Group (IBSG), August 2007.

City of Copenhagen (2016). Climate adaptation and urban nature: development catalogue, prepared by the Technical and Environmental Administration in collaboration with SLA Architects. (<u>https://issuu.com/sla_architects/docs/bynatur_booklet_uk_small</u>).

Cividino, S., Halbac-Cotoara-Zamfir, R., and Salvati, L. (2020). Revisiting the "City Life Cycle": Global urbanization and implications for regional development. *Sustainability*, 12(3), 1151, <u>https://doi.org/10.3390/su12031151</u>.

Clark, J. K., Conley, B., and Raja, S. (2021). Essential, fragile, and invisible community food infrastructure: The role of urban governments in the United States. *Food Policy*, 103, 102014; <u>https://doi.org/10.1016/j.foodpol.2020.102014</u>.

CLES, National organisation for local economies (2022). What is Community Wealth Building?. (<u>https://cles.org.uk/community-wealth-building/what-is-community-wealth-building</u>).

CNN (2019). Finland is winning the war on fake news. What it's learned may be crucial to Western democracy. (<u>https://edition.cnn.com/interactive/2019/05/europe/finland-fake-news-intl</u>).

Cohen, N., and Ilieva, R. T. (2021). Expanding the boundaries of food policy: The turn to equity in New York City. *Food Policy*, *103*, 102012, <u>https://doi.org/10.1016//j.foodpol.2020.102012</u>.

Committee of the Regions (2009). Committee of the Regions' White Paper on Multilevel Governance. (2009/C 211/01). Brussels, 17-18 June 2009. (<u>https://eur-lex.europa.eu/LexUriServ/LexUriServ.do?</u> uri=0J:C:2009:211:0001:0027:EN:PDF).

Committee of Regions (2020). Opinion on From farm to fork – the local and regional dimension, 141st CoR plenary session, 8-10 December 2020. NAT-VII/005. (<u>https://cor.europa.eu/en/our-work/Pages/OpinionTimeline.aspx?opId=CDR-594-2020</u>).

Conzelmann, T. (2008). A new mode of governing? Multi-level governance between cooperation and conflict, in T. Conzelmann and R. Smith (eds), Multi-level Governance in the European Union: Taking Stock and Looking Ahead, Baden Baden: Nomos.

Cordella, A. and Bonina, C.M. (2012). "A public value perspective for ICT enabled public sector reforms: A theoretical reflection", Government Information Quarterly, 29, 512-520.

COSIE (Co-creation of service innovation in Europe) (2021). How to shift towards co-creative governance? Learning co-creation in public service, a MOOC e-learning course on co-creation. (https://sola.kau.se/cosie/project-level-findings/co-creation-as-a-relational-service/how-may-governing-of-change-towards-co-creative-services-take-place).

Costanza-Chock, S. (2020). Design Justice: Community-Led Practices to Build the Worlds We Need, Information Policy Series, MIT Press.

Cottam, H. (2022). The Work Project: Imagining Transition, UCL Institute for Innovation and Public Purpose, 31 May 2022. (<u>https://medium.com/iipp-blog/the-work-project-imagining-transition-d3d56c4e7b4e</u>).

Council of Europe (2022). What is Democracy?. (<u>https://www.coe.int/en/web/compass/democracy</u>).

Craglia, M., Scholten, H.J., Micheli, M., Hradec, J., Calzada, I., Luitjens, S., Ponti, M. and Boter, J. (2021). Digitranscope: Key findings, Publications Office of the European Union, Luxembourg, ISBN 978-92-76-30424-1, doi:10.2760/169341, JRC124113.

Danish Agency for Digitisation (2013) Digital welfare: empowerment, flexibility and efficiency. Common Public Sector Strategy for Digital Welfare, 2013-2020. (<u>https://www.rm.dk/siteassets/sundhed/faginfo/center-for-telemedicin/english/documents/strategy for digital welfare 2013 2020.pdf</u>).

Danish Knowledge Centre for Housing Economics (Boligøkonomisk Videnscenter (2021). Demografi, socioøkonomi og boligstrukturer i danske kommuner, May 19, 2021. (<u>https://www.bvc.dk/faglige-udgivelser/demografi-sociooekonomi-og-boligstrukturer-i-danske-kommuner</u>).

DECODE (2017). Co-creation framework, methodologies and templates, H2020 R&I Project. (<u>https://decodeproject.eu/publications/decodes-co-creation-framework-methodologies-and-templates.html</u>).

DECODE (2020). Project Communication, exploitation plans, events report and overall project impact, H2020 R&I Project. (<u>https://decodeproject.eu/publications/project-communication-exploitation-plans-events-report-and-overall-project-impact.html</u>).

Democracy Collaborative (2022). Community Wealth Building. (<u>https://democracycollaborative.org/programs/</u><u>cwb</u>.

De Nigris, S., Gomez-Gonzales, E., Gomez Gutierrez, E., Martens, B., Iglesias Portela, M., Vespe, M., Schade, S, Micheli, M., Kotsev, A., Mitton, I., Vesnic Alujevic, L., Pignatelli, F., Hradec, J., Nativi, S., Sanchez Martin, J.I, Hamon, R. and Junklewitz, H. (2020). Artificial Intelligence and Digital Transformation: early lessons from the COVID-19 crisis, Craglia, M. editor(s), EUR 30306 EN, Publications Office of the European Union, Luxembourg, 2020, ISBN 978-92-76-20802-0, doi:10.2760/166278, JRC121305.

Derthick, M. (2010). Compensatory federalism. In B. G. Rabe (Ed.), Greenhouse governance: Addressing climate change in America (pp. 58–72). Washington, DC: Brookings Institution Press.

Dessart, F.J., Marandola, G., Hille, S.L. and Thøgersen, J. (2021). Comparing the impact of positive, negative, and graded sustainability labels on purchase decisions, European Commission, 2021, JRC127006.

Dewies, D., Denktaş, S., Giel, L., Noordzij, G. and Merkelbach, I. (2022) Applying Behavioural Insights to Public Policy: An Example From Rotterdam. *Global Implementation Research and Applications*. Volume 2, pp. 53–66, Springer. (https://link.springer.com/article/10.1007/s43477-022-00036-5).

De Vroey, M. (2015). Milton Friedman and the Monetarist Debate (from Part I - Keynes and Keynesian Macroeconomics) in "A History of Macroeconomics from Keynes to Lucas and Beyond", Cambridge University Press. (<u>https://www.cambridge.org/core/books/abs/history-of-macroeconomics-from-keynes-to-lucas-and-beyond/milton-friedman-and-the-monetarist-debate/E72475E0F070682E26D11F6CB6014FFC)</u>.

Dey, S., Saha, S., Singh, A.M. and McDonald-Maier, K. (2022). SmartNoshWaste: Using Blockchain, Machine Learning, Cloud Computing and QR Code to Reduce Food Waste in Decentralized Web 3.0 Enabled Smart Cities. *Smart Cities* 2022, *5*(1), 162-176: <u>https://doi.org/10.3390/smartcities5010011</u>.

Diez, T., Millard. J, Menichinelli, M. and Sorivelle M.N. (2019) The fabrication city, in book Albert, S (ed) "Innovative solutions for creating sustainable cities", Cambridge Scholars Publishing, Newcastle upon Tyne, UK.

Dobravec, V., Matak, N., Sakulin and C. Krajačić, G. (2021) Multilevel governance energy planning and policy: a view on local energy initiatives. *Energ Sustain Soc* (2021) 11:2. (<u>https://doi.org/10.1186/s13705-020-00277-</u> \underline{y}).

Dodson, S. (2008). Was software responsible for the financial crisis? *Guardian Newspaper*, 16 October 2008. (<u>https://www.theguardian.com/technology/2008/oct/16/computing-software-financial-crisis</u>).

Duch Brown, N., Rossetti, F. and Haarburger, R. (2021). Evolution of the EU market share of robotics: Data and methodology. EUR 30896 EN, Publications Office of the European Union, Luxembourg, 2021, ISBN 978-92-76-43794-9, doi:10.2760/292931, JRC124114.

Dunleavy, P., and Hood, C. (1994). From old public administration to new public management. *Public Money & Management* 14: 9–16.

Dunleavy, P. and Margetts. H. (2006). New Public Management is Dead: Long Live Digital Era Governance", Journal of Public Administration Research and Theory, July 2006.

Dunleavy, P. (2022). Information regimes in government bureaucracies and 'digital decompression'. Paper presented to the 'Public Administration and Technology' Panel, UK Political Studies Association Conference, University of York, 12 April 2022. (<u>http://eprints.lse.ac.uk/114488/1/Dunleavy information regimes in government bureaucracies published.pdf</u>).

ECaTT Consortium. (2000). ECaTT Final Report, European Commission Research project.

Economics Observatory (2021). Zoomshock: how is working from home affecting cities and suburbs?. (<u>https://www.economicsobservatory.com/zoomshock-how-is-working-from-home-affecting-cities-and-suburbs</u>).

ECSA, European Citizen Science Association (2021). Vision and mission. (<u>https://ecsa.citizen-science.net/about-us</u>).

ECSA, European Citizen Science Association (2020). Citizen science declaration: Our world – our goals: Citizen Science for the Sustainable Development Goals. (<u>https://survey.naturkundemuseumberlin.de/sites/default/files/uploads/Citizen%20Science%20SDG%20Declaration.pdf</u>).

Edelmann, N., Schoßböck, J., and Albrecht, V. (2021). Remote Work in Public Sector Organisations: Employees' Experiences in a Pandemic Context. Paper presented at the Digital Government Research (dg.o 2021) Omaha, Nebraska.

Edelmann, N. and Millard, J. (2021). Telework development before, during and after COVID-19, and its relevance for organizational change in the public sector. In 14th International Conference on Theory and Practice of Electronic Governance (ICEGOV 2021), October 06–08, 2021, Athens, Greece, ACM, New York, NY, USA: https://doi.org/10.1145/3494193.3494252.

EDS (Electronic Data Systems Corporation) (2005), Delivering modern services strategy: EDS input to first stages of eGovernment Unit consultation in the UK, London, England.

Ehnert, F., Kern, F., Borgstr€om, S., Gorissen, L., Maschmeyer, S., and Egermann, M. (2018). Urban sustainability transitions in a context of multi-level governance: A comparison of four European states. Environmental Innovation and Societal Transitions, 26, 101–116.

EIT, Economist Intelligence Unit (2021). Europe's labour market after furlough, 8 September 2021. (<u>https://www.eiu.com/n/europes-labour-market-after-furlough</u>).

Electoral Reform Society (2019). Background on Citizens' Assemblies 2.8 July 2019. (<u>https://www.electoral-reform.org.uk/latest-news-and-research/parliamentary-briefings/what-are-citizens-assemblies</u>).

Ellen MacArthur Foundation (2013). Towards the circular economy. (<u>https://www.ellenmacarthurfoundation.org/assets/downloads/publications/Ellen-MacArthur-Foundation-Towards-the-Circular-Economy-vol.1.pdf</u>).

Elliot, L. (2020). Eurozone economy shrinks by record 12.1% due to coronavirus. *Guardian Newspaper*. 31 July 2020. (<u>https://www.theguardian.com/business/2020/jul/31/eurozone-economy-shrinks-by-record-121-due-to-coronavirus-crisis</u>).

ENoLL, Europeam Network of Living Labs. (2022) Citizen-driven network empowering everyone to innovate. (<u>https://enoll.org</u>).

Errandonea, L. (2023). *Exploring the impact of digital transformation on public governance - A community perspective*, Schade, S. (Editor), Publications Office of the European Union, Luxembourg, 2023, doi:10.2760/679503, JRC133342.

Erumban, A. and Jong, S. (2006). Cross-country differences in ICT adoption: A consequence of Culture?, *Journal of World Business*, vol. 41, no. 4, pp 302–314.

EURACTIV (2022a). Macron wants to 'adapt' EU Farm to Fork to the post-Ukraine war world, By Hugo Struna, 18 March 2022. (<u>https://www.euractiv.com/section/agriculture-food/news/macron-wants-to-adapt-eu-farm-to-fork-to-the-post-ukraine-war-world</u>).

EURACTIVE (2002b). Labour shortages felt all over Europe, 13 October 2022. (<u>https://www.euractiv.com/section/politics/news/labour-shortages-felt-all-over-europe</u>).

Eurofound (2021). COVID-19 and digitalisation, 15 December 2021. (<u>https://www.euractiv.com/section/agriculture-food/news/macron-wants-to-adapt-eu-farm-to-fork-to-the-post-ukraine-war-world/and https://www.eurofound.europa.eu/data/digitalisation/research-digests/covid-19-and-digitalisation).</u>

Euronews (2022). No end in sight for the 'Great Resignation' as inflation pushes workers to seek better-paid jobs, 22 May 2022. (<u>https://www.euronews.com/next/2022/05/25/no-end-in-sight-for-the-great-resignation-as-inflation-pushes-workers-to-seek-better-paid</u>).

European Commission (2001). European Governance A White Paper. COM (2001) 428, Brussels, 25.7.2001. (<u>https://op.europa.eu/en/publication-detail/-/publication/11c3e337-9cf5-4603-a518-cacb77207e3b</u>).

European Commission (2010) "A European strategy for smart, sustainable and inclusive growth, 2010-2010. (<u>http://ec.europa.eu/eu2020/pdf/COMPLET%20EN%20BARROS0%20%20%20%20%20-%20EN%20version.pdf</u>).

European Commission (2013a). Powering European Public Sector Innovation: towards a new architecture, Report of the Expert Group on Public Sector Innovation, Directorate-General for Research and Innovation Union.

European Commission (2013b). A vision for public services, prepared by DG CONNECT after an expert workshop and open public consultation. (<u>https://digital-strategy.ec.europa.eu/en/news/vision-public-services</u>).

European Commission (2014). Study on eGovernment and the Reduction of Administrative Burden, Luxembourg, Publications Office of the European Union.

European Commission (2015a) Nature-Based Solutions & Re-Naturing Cities: Final Report of the Horizon 2020 Expert Group on 'Nature-Based Solutions and Re-Naturing Cities', Directorate-General for Research and Innovation. (<u>https://publications.europa.eu/en/publication-detail/-/publication/fb117980-d5aa-46df-8edc-af367cddc202</u>).

European Commission (2015b). Energy union, 25 February 2015. (<u>https://energy.ec.europa.eu/topics/energy-strategy/energy-union_en</u>).

European Commission (2016) "Communication on EU eGovernment Action Plan 2016-2020 -- Accelerating the digital transformation of government", 19 April 2016. (<u>https://ec.europa.eu/digital-single-market/en/news/communication-eu-egovernment-action-plan-2016-2020-accelerating-digital-transformation</u>).

European Commission (2016b). Governance of the Energy Union and Climate Action. (<u>https://climate.ec.europa.eu/eu-action/climate-strategies-targets/progress-made-cutting-</u>emissions/governance-energy-union-and-climate-action en).

European Commission (2017). Toolbox 2017 edition - Quality of Public administration. (<u>https://ec.europa.eu/social/BlobServlet?docId=18556&langId=en</u>).

European Commission (2018). Digital Government Transformation. Presentation by Jean-Francois Junger, DG CONNECT H4, to the College of Europe, March 2018.

European Commission (2019a). Bridging the digital divide for a more inclusive Europe, Fact sheets. (<u>https://home-affairs.ec.europa.eu/pages/publication/bridging-digital-divide-more-inclusive-europe en</u>).

European Commission (2019b). Co-creating innovative public services for citizens and businesses. CORDIS Results Pack on digital government: a thematic collection of innovative EU-funded research results, September 2019. (<u>https://digital-strategy.ec.europa.eu/en/library/digital-government-co-creating-innovative-public-</u> services-citizens-and-businesses).

European Commission (2019c). Joint Research Centre, Rudkin, J., Kimbell, L., Stoermer, E., et al., *The future of government 2030+ : a citizen centric perspective on new government models*, Publications Office, 2019: <u>https://data.europa.eu/doi/10.2760/145751</u>.

European Commission (2019d). A European Green Deal -- Striving to be the first climate-neutral continent (<u>https://ec.europa.eu/info/strategy/priorities-2019-2024/european-green-deal_en</u>).

European Commission. (2020a). Circular Economy Action Plan for a cleaner and more competitive Europe, European Green Deal. Brussels. (<u>https://ec.europa.eu/environment/circular-economy/pdf/new_circular_economy_action_plan.pdf</u>).

European Commission. (2020b). Farm to Fork strategy -- for a fair, healthy and environmentally-friendly food system. (<u>https://food.ec.europa.eu/system/files/2020-05/f2f_action-plan_2020_strategy-info_en.pdf</u>).

European Commission. (2020c). Europe's moment: Repair and Prepare for the Next Generation, Brussels, 27.5.2020. (<u>https://eur-lex.europa.eu/legal-content/EN/TXT/?gid=1590732521013&uri=COM:2020:456:FIN</u>).

European Commission. (2020d). A renewed trade policy for a stronger Europe: Consultation Note, 16 June 2020. (<u>https://trade.ec.europa.eu/doclib/docs/2020/june/tradoc_158779.pdf</u>).

European Commission (2020e). The EU's Cybersecurity Strategy for the Digital Decade, Joint communication to the European parliament and the Council, 16-12-22. (<u>https://eur-lex.europa.eu/legal-content/ga/TXT/?uri=CELEX:52020JC0018</u>).

European Commission (2020f). Berlin Declaration on Digital Society and Value-based Digital Government, Shaping Europe's Digital Future, 8 December 2020. (<u>https://digital-strategy.ec.europa.eu/en/news/berlin-declaration-digital-society-and-value-based-digital-government</u>).

European Commission (2021a). Industry 5.0: Towards more sustainable, resilient and human-centric industry. January 2021. (<u>https://ec.europa.eu/info/news/industry-50-towards-more-sustainable-resilient-and-human-</u>

<u>centric-industry-2021-jan-07</u> en#:~:text=The%20report%20%E2%80%9CIndustry%205.0%E2%80% 9D%20published,centre%20of%20the%20production%20process).

European Commission (2021b). Digital Public Administration Factsheets, National Interoperability Framework Observatory (NIFO) project. (<u>https://joinup.ec.europa.eu/collection/nifo-national-interoperability-framework-observatory/digital-public-administration-factsheets</u>).

European Commission (2021c). The EU and the United Nations – common goals for a sustainable future. (<u>https://ec.europa.eu/info/strategy/international-strategies/sustainable-development-goals/eu-and-united-nations-common-goals-sustainable-future_en</u>).

European Commission (2021d). Transparent and inclusive governance through innovation and collaboration for civil society, project with Uzbekistan during 2021. (<u>https://international-partnerships.ec.europa.eu/policies/programming/projects/transparent-and-inclusive-governance-through-innovation-and-collaboration-civil-society_en).</u>

European Commission (2021e). EU Action Plan on Integration and Inclusion 2021-2027. (<u>https://ec.europa.eu/migrant-integration/news/one-year-eu-action-plan-integration-and-inclusion-2021-2027 en#:~:text=Today%2C%2024%20November%202021%2C%20marks,that%20promote%20inclusion%20for%20everyone</u>).

European Commission (2021f). Evaluating the Impact of Nature-based Solutions: A Handbook for Practitioners. (<u>https://op.europa.eu/en/publication-detail/-/publication/d7d496b5-ad4e-11eb-9767-01aa75ed71a1</u>).

European Commission (2021g). EGovernment benchmark 2021, 12-11-21. (<u>https://op.europa.eu/en/publication/d30dcae1-436f-11ec-89db-01aa75ed71a1</u>).

European Commission (2021h). European Quality of Government Index 2021. (<u>https://ec.europa.eu/regional_policy/en/information/maps/quality_of_governance/#:~:text=The%20European%</u>20Quality%20of%20Government_in%20their%20region%20of%20residence).

European Commission (2021i). Forging a climate-resilient Europe - the new EU Strategy on Adaptation to Climate Change, COM(2021)82 – Communication:.

European Commission (2021j). Agriculture and rural development: the new common agricultural policy, 2023-27, 2 December 2021. (<u>https://agriculture.ec.europa.eu/common-agricultural-policy/cap-overview/new-cap-2023-27 en</u>).

European Commission (2021k). Europe's Digital Decade: digital targets for 2030, 9 March 2021. (<u>https://ec.europa.eu/info/strategy/priorities-2019-2024/europe-fit-digital-age/europes-digital-decade-digital-targets-2030 en</u>).

European Commission (20211). EU Missions in Horizon Europe. (<u>https://research-and-innovation.ec.europa.eu/funding/funding-opportunities/funding-programmes-and-open-calls/horizon-europe/eu-missions-horizon-europe_en</u>).

European Commission (2022a). European Data Governance Act, Shaping Europe's digital future. (<u>https://digital-strategy.ec.europa.eu/en/policies/data-governance-act</u>).

European Commission (2022b). Energy communities -- Citizen-driven energy actions that contribute to the clean energy transition, advancing energy efficiency within local communities. (<u>https://energy.ec.europa.eu/topics/markets-and-consumers/energy-communities en</u>).

European Commission (2022c). Cyber Defence: EU boosts action against cyber threats, 10 November 2022. (<u>https://ec.europa.eu/commission/presscorner/detail/en/IP_22_6642</u>).

European Commission (2022d). Resilience dashboards -- The social and economic, green, digital, and geopolitical resilience dashboards. (<u>https://joint-research-centre.ec.europa.eu/resilience/resilience/dashboards/20provide/20a.geopolitical</u>).

European Commission (2022e). REPowerEU: Joint European action for more affordable, secure and sustainable energy, 8 March 2022. (<u>https://ec.europa.eu/commission/presscorner/detail/en/IP_22_1511</u>).

European Commission (2022f). Cross-border teleworkers, digital nomads, highly mobile and posted workers. Unblocking the amendment of the Social Security Coordination Regulations, 21 October 2022, DG Employment, Social Affairs & Inclusion. (<u>https://ec.europa.eu/social/main.jsp?langld=en&</u> catld=1098&eventsId=2019&furtherEvents=yes). European Commission (2022g). Urban Agenda for the EU. (https://futurium.ec.europa.eu/en/urban-agenda).

European Commission (2022h). Twinning the green and digital transitions in the new geopolitical context, 2022 Strategic Foresight Report, Brussels, 29.6.2022, COM(2022) 289 final. (<u>https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A52022DC0289&gid=1658824364827</u>).

European Commission (2022i). New Cohesion Report shows that differences between EU regions are narrowing thanks to EU support, Press Release, 9 February 2022. (<u>https://ec.europa.eu/commission/presscomer/detail/en/ip_22_762</u>).

European Commission (2022j). Structural Reform Support, The Directorate-General for Structural Reform Support (DG REFORM). (<u>https://commission.europa.eu/about-european-commission/departments-and-executive-agencies/structural-reform-support en</u>).

European Commission (2022k). Interoperable Europe Act Proposal, 30 November 2022. (<u>https://commission.europa.eu/publications/interoperable-europe-act-proposal en</u>).

European Commission (2022l). Digital Rights and Principles: Presidents of the Commission, the European Parliament and the Council sign European Declaration, Press Release, 15 December 2022. (<u>https://ec.europa.eu/commission/presscorner/detail/en/ip_22_7683</u>).

European Commission (2023). Trustworthy artificial intelligence for a fair and democratic Europe, JRC Portfolio 15, European Centre for Algorithmic Transparency, 24 February 2023. (<u>https://joint-research-centre.ec.europa.eu/jrc-science-and-knowledge-activities/trustworthy-artificial-intelligence-ai en</u>).

European Digital Forum (2015). Government of the future: how will digital technology change the way we live,workandgovern.DigitalInsights,SpecialEdition,3/2015.(https://www.researchgate.net/publication/293334332Government of the Future How Digital TechnologyWill Change the Way We Live Work and Govern).

European Parliament (2020). Uncertainty/EU/hope: public opinion in times of COVID-19, Public opinion survey commissioned by the European Parliament, June 2020. (<u>https://www.europarl.europa.eu/at-your-service/files/be-heard/eurobarometer/2020/public opinion in the eu in time of coronavirus crisis/report/en-covid19-survey-report.pdf</u>).

European Parliament (2021a). Europe's Digital Decade and Autonomy, Directorate-General for Internal Policies,October2021.(2021)695465EN.pdf).

European Parliament (2021b). Energy policy: general principles, October 2021. (<u>https://www.europarl.europa.eu/factsheets/en/sheet/68/energy-policy-general-principles</u>).

European Parliament (2022a). EU strategic autonomy 2013-2023: From concept to capacity, Briefing, EU Strategic Autonomy Monitor, July 2022. (<u>https://www.europarl.europa.eu/RegData/etudes/</u>BRIE/2022/733589/EPRS_BRI(2022)733589_EN.pdf).

European Parliament (2022b). The principle of subsidiarity, Fact Sheets on the European Union, 05-2022. (<u>https://www.europarl.europa.eu/factsheets/en/sheet/7/the-principle-of-subsidiarity</u>).

European Parliament (2022c). Just Transition Fund, Fact Sheets of the European Union, March 2022. (<u>https://www.europarl.europa.eu/factsheets/en/sheet/214/just-transition-fund</u>).

European Systemic Risk Board (2022). Fiscal support and macroprudential policy - Lessons from the COVID-19 pandemic. (<u>https://www.esrb.europa.eu/pub/pdf/reports/esrb.2022.11.21.note.on.fiscal.support.and.macroprudential.policy~e5abc993e9.en.pdf</u>).

European Union Law (2022). EU governance. 2022-01-05. (<u>https://eur-lex.europa.eu/EN/legal-content/glossary/eu-governance.html#:~:text=The%20expression%20'European%20governance'%</u> 20designates,closer%20to%20the%20EU%20institutions).

Eurostat, European Commission. Statistical Office of the European Union. (2018). Methodological manual on territorial typologies: 2018 edition.

Eurostat, European Commission. Statistical Office of the European Union. (2021a). Living conditions in Europe - poverty and social exclusion. (<u>https://ec.europa.eu/eurostat/statistics-explained/index.php?title=Living</u> conditions in Europe - poverty and social exclusion#:~:text=In%202020%2C%2096.5%20million%20

people,21.9%20%25%20of%20the%20EU%20population.&text=The%20risk%20of%20poverty%20or,%25% 20compared%20with%2020.9%20%25).

Eurostat, European Commission. Statistical Office of the European Union (2021b). Digital economy and society statistics - households and individuals. (<u>https://ec.europa.eu/eurostat/statistics-explained/index.php?title=</u> <u>Digital economy and society statistics - households and individuals</u>).

Eurostat (2021c). Individuals using the internet for interaction with public authorities. (<u>https://ec.europa.eu/</u><u>eurostat/databrowser/view/tin00012/default/table?lang=en</u>).

Eurostat, European Commission. Statistical Office of the European Union (2021d). Eurostat regional yearbook, 2021 edition.

EU Science Hub (2022). Pilot living labs at the JRC. (<u>https://joint-research-centre.ec.europa.eu/pilot-living-labs-jrc_en</u>).

FERMA, Federation of European Risk Management Associations (2022). European Risk Manager Report 2022, 14/09/22. (<u>https://www.ferma.eu/publication/european-risk-manager-report-2022</u>).

Fernandez Llorca, D. and Gomez Gutierrez, E. (2022). Artificial Intelligence in Autonomous Vehicles: towards trustworthy systems, European Commission, JRC128170.

Fleming, S. (2020). This is how COVID-19 is affecting informal workers. World Economic Forum. 29 July 2020. (<u>https://www.weforum.org/agenda/2020/07/coronavirus-impact-informalworkers-world-bank</u>).

Forbes (2022). Why The Digital Nomad Lifestyle Is On The Rise, 17 July 2022. (<u>https://www.forbes.com/sites/carolinecastrillon/2022/07/17/why-the-digital-nomad-lifestyle-is-on-the-rise/?sh=7bbdb1dc4934</u>).

Financial Times (2022). UK tech group WANdisco adopts four-day working week, 7 February 2022. (<u>https://www.ft.com/content/82fe2509-960b-43be-bf92-4482f5d944d4</u>).

Frantzeskaki, N., Avelino, F., and Loorbach, D. (2013). Outliers or frontrunners? Exploring the (self-) governance of community-owned sustainable energy in Scotland and the Netherlands. In E. Michalena and J. Hills (Eds.), Renewable energy governance. Understanding the complexities and challenges of RE implementation (pp. 101–116), Berlin, Germany: Springer.

Freedom House (2022). Freedom in the World, 2022: The Global Expansion of Authoritarian Rule, February 2022. (<u>https://freedomhouse.org/sites/default/files/2022-02/FIW_2022_PDF_Booklet_Digital_Final_Web.pdf</u>).

Frey, C.B. and Osborne, M.A. (2013). The future of employment: how susceptible are jobs to computerization? Oxford University Press.

Futuregov (2020). Redesigning early action, prevention and targeted support in Dagenham and Barking, London, UK. (<u>https://wearefuturegov.com/case-study/barking-dagenham-childrens-social-care</u>).

Gallonier, V. (2020). Teleworking: a better quality of life for the employee but risks of isolation and of losing contact with the employer. (Memorandum 4). World Pandemic Resarch Network.

Garcia-Sanchez IM, Cuadrado-Ballesteros B, Frias-Aceituno J (2013). Determinants of Government Effectiveness, *International Journal of Public Administration*, Volume 36, 2013 - Issue 8.

GARN, Global Alliance for the Rights of Nature (2022). <u>https://www.garn.org/about-garn</u>.

Gascó-Hernández (2014). Open government: opportunities and challenges for public governance, Springer Science + Business Media, New York.

Gayle, D. (2017). Robots 'could replace 250,000 UK public sector workers', *Guardian Newspaper*, 6 February 2017. (<u>https://www.theguardian.com/technology/2017/feb/06/robots-could-replace-250000-uk-public-sectorworkers</u>).

Geofutures (2020). Sustainable Currency? A business analysis of the Bristol Pound, more than money, July 2020. (<u>https://bristolpound.org/wp-content/uploads/202101-geofutures-report-on-b.pdf</u>).

George, B., Verschuere, B., Wayenberg, E., and Zaki, B. L. (2020). A Guide to Benchmarking COVID-19 Performance Data. *Public Administration Review*, 80(4), 696-700. doi:<u>https://doi.org/10.1111/puar.1325</u>.

Gilman, H.R. (2016). Engaging Citizens: Participatory Budgeting and the Inclusive Governance Movement within the United States, Ash Center for Democratic Governance and Innovation, Harvard Kennedy School, Occasional Papers, January 2016. (<u>https://ash.harvard.edu/files/ash/files/participatory-budgeting-paper.pdf</u>).

Global Citizen Solutions (2022). Digital Nomad Visa Europe – Everything You Need to Know. (<u>https://www.globalcitizensolutions.com/digital-nomad-visa-europe</u>).

Global Worplace Allaince (2021). Latest Work-At-Home/Telecommuting/Mobile Work/Remote Work Statistics: https://globalworkplaceanalytics.com/telecommuting-statistics.

Goharia,S. and Larssætherb, S. (2019). Sustainable energy planning as a co-creative governance challenge. Lessons from the Zero Village Bergen. *International Journal of Sustainable Energy Planning and Management*, Vol. 24 2019 147–154. (<u>https://journals.aau.dk/index.php/sepm/article/view/3353/3073</u>).

Gouillart, F. and Hallett, T. (2015). Co-Creation in Government. *Stanford Social Innovation Review*, Spring 2015. (<u>https://ssir.org/articles/entry/co_creation_in_government#bio-footer</u>).

GOV 3.0 (2021). Scientific Foundations Training & Entrepreneurship Activities in the Domain of ICT-enabled Governance, EU project, 2017-21. (<u>https://www.gov30.eu</u>).

Grando, S., Carey, J., Hegger, E., Jahrl, I., and Ortolani, L. (2017). Short Food Supply Chains in Urban Areas: Who Takes the Lead? Evidence from Three Cities across Europe. Urban Agriculture and Regional Food Systems, 2(1), 1–11.

GRETA (2022). GReen Energy Transition Actions, funded under the Horizon Europe Programme, SOCIETAL CHALLENGES, Secure, clean and efficient energy:. (<u>https://cordis.europa.eu/project/id/101022317</u> and <u>https://projectgreta.eu/project/</u>).

Hackemoon (2001). Web 1.0 to Web4: A Brief History of The Evolution of Internet Technologies. 9 May 2021. (<u>https://hackemoon.com/web-10-to-web4-a-brief-history-of-the-evolution-of-internet-technologies-tl64341x</u>).

Hale, T. (2022). We've found one factor that predicts which countries best survive Covid, The Guardian Newspaper, 24 March 2022. (<u>https://www.thequardian.com/commentisfree/2022/mar/24/countries-covid-trust-damage-pandemic</u>).

Halmos, A., Misuraca, G. and Viscusi, G. (2019). From Public Value to Social Value of Digital Government: Co-Creation and Social Innovation in European Union Initiatives, In: 52nd Hawaii International Conference on System Sciences, 08-11 January 2019, Grand Wailea, Maui, Hawaii, USA, PROCEEDINGS OF THE HAWAII INTERNATIONAL CONFERENCE ON SYSTEM SCIENCE, 2019, ISSN 0073-1129 (online), p. 2974-2983, JRC115346

Harris, J. (2022). Working from home has entrenched inequality – how can we use it to improve lives instead?, *Guardian Newspaper*, 9 January 2022. (<u>https://www.theguardian.com/commentisfree/2022/jan/09/working-from-home-inequality-pandemic-inequalities-uk</u>).

Harrison, T.M., Pardo, T.A., Cook, M. (2012). Creating Open Government Ecosystems: A Research and Development Agenda/ Future Internet 2012, 4(4), 900-928.

Harsh, M. (2013). Informal governance of emerging technologies in Africa', in Christiansen, T. and Neuhold, C. (Eds), International Handbook on Informal Governance, Cheltenham, Edward Elgar.

Harvey, F. (2022). Ukraine war threatens global heating goals, warns UN chief, *Guardian Newspaper*, 21 March 2022. (<u>https://www.theguardian.com/environment/2022/mar/21/ukraine-war-threatens-global-heating-goals-warns-un-chief</u>).

Hasdell, R. (2020). Universal basic income a cross-synthesis of reviews, Stanford University Basic Income Lab, July 2020. (<u>https://basicincome.stanford.edu/uploads/Umbrella%20Review%20BI_final.pdf</u>).

Healey, P. (2015). Citizen-generated local development initiative. Recent English experience. International Journal of Urban Sciences, 19(2), 109–118.

Heller, N. (2020). The future of democracy: politics without politicians. *The New Yorker*, February 19, 2020. (<u>https://www.newyorker.com/news/the-future-of-democracy/politics-without-politicians</u>).

Henley, J. (2020). Finnish basic income pilot improved wellbeing, study finds. Guardian Newspaper, 7 May 2020. (<u>https://www.theguardian.com/society/2020/may/07/finnish-basic-income-pilot-improved-wellbeing-study-finds-coronavirus</u>).

Henley, J. (2022). Energy citizenship: Europe's communities forging a low-carbon future. *Guardian Newspaper*, 3 September 2022. (<u>https://www.theguardian.com/environment/2022/sep/03/energy-citizenship-europes-communities-forging-a-low-carbon-future</u>).

Hernández, L. (2022). Helping private and public organizations in building platform strategies for regeneration, Platforms for Regeneration. (<u>https://luciahernandez.co</u>).

Hidvegi, F., Leufer, D. and Massé, E. (2021). The EU should regulate AI on the basis of rights, not risks *,Accessnow Blog.* 17 February 2021. (<u>https://www.accessnow.org/eu-regulation-ai-risk-based-approach</u>).

Hofstede Insights. (2022). The six dimensions of national culture, (<u>https://hi.hofstede-insights.com/national-culture</u>), and country comparisons data (<u>https://www.hofstede-insights.com/country-comparison</u>).

Hood, C. (1991). A Public Management for All Seasons, *Public Administration*, 69 (Spring), 3-19.

Hoppe, T., Graf, A., Warbroek, B., Lammers, I., and Lepping, I. (2015). Local governments supporting local energy initiatives. Lessons from the best practices of Saerbeck (Germany) and Lochem (the Netherlands). Sustainability, 7(2), 1900–1931.

Horton, H. (2022). James Lovelock, creator of Gaia hypothesis, dies on 103rd birthday, *Guardian Newspaper*, 27 July 2022. (<u>https://www.thequardian.com/environment/2022/jul/27/james-lovelock-creator-of-gaia-hypothesis-dies-on-103rd-birthday</u>).

Howaldt, J., Butzin, A., Domanski, D., and Kaletka, C. (2014). Theoretical Approaches to Social Innovation - A Critical Literature Review. A deliverable of the project: "Social Innovation: Driving Force of Social Change" (SF DRIVE). Dortmund: Sozialforschungsstelle.

House of Commons Science and Technology Committee, UK (2016) Robotics and artificial intelligence. (<u>http://www.publications.parliament.uk/pa/cm201617/cmselect/cmsctech/145/145.pdf</u>).

Hristo, H., Millard, J., Pravst, I. and Janssen, M. (2022). European household spending and socio-economic impacts on food behaviour during the first wave of COVD-19. Frontiers in Nutrition. 2022;9: 869091. DOI: 10.3389/fnut.2022.869091. (<u>https://www.frontiersin.org/articles/10.3389/fnut.2022.869091/full</u>).

Hufen, J. A. M., and Koppenjan, J.F.M. (2015). Local renewable energy cooperatives. Revolution in disguise? Energy, Sustainability and Society, 5(1), 161.

ILO, International Labour Organisation (2021a). Cooperatives and the wider social and solidarity economy as vehicles to decent work in the culture and creative sector. (https://www.ilo.org/global/topics/cooperatives/sse/WCMS 825146/lang--en/index.htm).

ILO, International Labour Organisation (2021b). Remote work and the right to disconnect in Europe. (<u>https://www.ilo.org/wcmsp5/groups/public/--europe/---ro-geneva/---ilo-paris/documents/meeting</u> <u>document/wcms</u> 767509.pdf).

ILO. (2021c). Working from home: From invisibility to decent work. Retrieved from Geneva. (<u>https://www.ilo.org/wcmsp5/groups/public/---ed_protect/--protrav/---travail/documents/publication/wcms_765806.pdf</u>).

IMF (2012) A Relative Question, *Finance & Development*, December 2012, Vol. 49, No. 4, International Monetary Fund. (<u>http://www.imf.org/external/pubs/ft/fandd/2012/12/ravallion.htm</u>).

IMF (2022). War sets back the global recovery, World Economic Outlook, Full Report, April 2022. (<u>https://www.imf.org/-/media/Files/Publications/WE0/2022/April/English/text.ashx</u>).

International Research Society for Public Management (2022). Designing for new forms of Governance – how to leverage on bottom-up and top-down initiatives for public service system change, 3 February 2022. (<u>https://blog.experientia.com/designing-for-new-forms-of-governance</u>).

iPES Food (2019). Towards a common food policy for the European Union: the policy reform and realignment that is required, International Panel of Experts on Sustainable Food Systems. (<u>https://www.ipes-food.org/ img/upload/files/CFP ExecSummary EN.pdf</u>).

Ingram, G.K. and Hong, Y.-H. (2012). Value Capture and Land Policies, *The Lincoln Institute of Land Policy*, Bolton.

Involve Foundation (2018). The citizens' assembly behind the Irish abortion referendum, 30 May 2018. (<u>https://involve.org.uk/resources/blog/opinion/citizens-assembly-behind-irish-abortion-referendum</u>).

ITU, International Telecommunications Union (2019). New ITU data reveal growing Internet uptake but a widening digital gender divide. Geneva, 5-11-19. (<u>https://www.itu.int/en/mediacentre/Pages/2019-PR19.aspx</u>).

IUCN (2020). IUCN Global Standard for Nature-based Solutions – A user-friendly framework for the verification, design and scaling up of NBS, first edition. The International Union for the Conservation of Nature and Natural Resources. (<u>https://portals.iucn.org/library/sites/library/files/documents/2020-020-En.pdf</u>).

Janssen, M. and Estevez, E. (2013). Lean government and platform-based governance: Doing more with Less, Government Information Quarterly. Vol. 30. Supplement 1, pp. S1-S8.

Jerch, R., Kahn, ME and Li, S. (2016). Efficient Local Government Service Provision: The Role of Privatization and Public Sector Unions, *NBER Working Papers*, No. 22088, March 2016, *National Bureau of Economic Research, Cambridge*, Mass, USA.

JRC (2020). Telework in the EU before and after the COVID-19: where we were, where we head to, The European Commission's Joint Research Centre, JRC120945.

JRC (2022a). Rural development. (<u>https://joint-research-centre.ec.europa.eu/scientific-activities-z/rural-development_en</u>).

JRC (2022b). The future of cities, Urban Data Platform Plus. (https://urban.irc.ec.europa.eu/thefutureofcities).

Kahneman, D., Sibony, O. and Sunstein, C. (2021). Noise: A Flaw in Human Judgment. Hachette Book Group, May 2021.

Kaminski, I. (2022). Eco beauty company 'appoints nature' to its board of directors, *Guardian Newspaper*, 22 September 2022. (<u>https://www.theguardian.com/environment/2022/sep/22/eco-beauty-company-faith-in-nature-board-directors</u>).

Kangas, O., Jauhiainen, S., Simanainen, M., Ylikanno, M. (2021). Experimenting with Unconditional Basic Income: Lessons from the Finnish BI Experiment 2017-2018. Elgaronline. (https://www.elgaronline.com/view/edcoll/9781839104848/9781839104848.xml).

Kasdan, D.O. (2018). Toward a theory of behavioral public administration

Katz, B. and Noring, L. and Verdis, S. (2017). Governing city infrastructure: Who drives the urban project cycle (<u>https://www.brookings.edu/research/governing-city-infrastructure-who-drives-the-urban-project-cycle</u>).

Katz, B. and Nowak, J. (2018). The New Localism: How Cities Can Thrive in the Age of Populism, the Brookings Institute, January 9, 2018.

Kelly, G., Mulgan, G., Muers, S. (2002). Creating public value: an analytical framework for public sector reform, prepared for the UK Strategy Unit, Cabinet Office, October 2002.

Kert, K., Vebrova, M. and Schade, S. (2022). Regulatory learning in experimentation spaces, European Commission, 2022, JRC130458.

Kessari, M., Joly, C., Jaouen, A., and Jaeck, . M. (2020). Alternative food networks: good practices for sustainable performance. Journal of Marketing Management, 36(15-16)1417–1446, https://doi.org/10.1080/0267257X.2020.1783348.

Kim, S. (2015). National culture and public service motivation: investigating the relationship using Hofstede's five cultural dimensions. *International Review of Administrative Sciences*, December 29, 2015 Research Article, Sage Publishing: <u>https://doi.org/10.1177/0020852315596214</u>.

King's Fund (2019). A citizen-led approach to health and care: Lessons from the Wigan Deal, 26 June 2019. (<u>https://www.kingsfund.org.uk/publications/wigan-deal</u>).

Koczetkow, B., Klimczuk, A. (2022). The Context of Public Policy on the Sharing Economy. In: Česnuitytė, V, Klimczuk, A., Miguel, C., Avram, G. (eds) The Sharing Economy in Europe. Palgrave Macmillan, Cham. https://doi.org/10.1007/978-3-030-86897-0_3.

Kooiman, J. (2003). Societal Governance. In: Katenhusen, I., Lamping, W. (eds) Demokratien in Europa. VS Verlag für Sozialwissenschaften, Wiesbaden. <u>https://doi.org/10.1007/978-3-663-09584-2_11</u>.

Koppenjan, J. (2015). New public governance: a framework International Summer school on Smart networks and Sustainable partnerships, Snekkersten, Denmark, 27 June 2015. (<u>https://www.cbs.dk/files/cbs.dk/joop - slides.pdf</u>).

Kougias I., Szabó S., Scarlat N., Monforti F., Banja M., Bódis K., Moner-Girona M., Water-Energy-Food Nexus Interactions Assessment: Renewable energy sources to support water access and quality in West Africa. (2018). Luxembourg, European Commission, 2018, EUR 29196 EN, ISBN 978-92-79-84034-0, doi:10.2760/1796.

KPMG (2021). The future of towns and cities post COVID-19. January 2021, UK. (<u>https://home.kpmg/uk/en/home/insights/2021/01/future-of-towns-and-cities-post-covid-19.html</u>).

Król, K. (2020). Evolution of online mapping from web 1.0 to web 6.0, *Geomatics, Land management and Landscape* No. 1, 2020, 33–51. (<u>https://gll.urkedu.pl/zasoby/74/GLL-1-3-2020.pdf</u>).

Kunze, F., Hampel, K., Zimmerrmann, S. (2020). Homeoffice in der Corona-Krise – eine nachhaltige Transformation der Arbeitswelt? Policy Paper N° 02 16. Juli 2020. (<u>https://www.progressives-zentrum.org/wp-content/uploads/2020/07/Studie Home-Office-in-der-Corona-Krise.pdf</u>).

Kuziemski, M., Mergel, I., Ulrich, P. and Martinez, A. (2022). GovTech Practices in the EU, EUR 30985 EN, Publications Office of the European Union, Luxembourg, ISBN 978-92-76-47234-6, doi:10.2760/74735, JRC128247.

Landemore, H. (2020). Open Democracy: Reinventing Popular Rule for the Twenty-First Century, Princeton University Press.

Lavelle, J. (2020). Gartner CFO Survey Reveals 74% Intend to Shift Some Employees to Remote Work Permanently, 23.11.2020. (<u>https://www.gartner.com/en/newsroom/press-releases/2020-04-03-gartner-cfo-surey-reveals-74-percent-of-organizations-to-shift-some-employees-to-remote-work-permanently2</u>).

Lee, G. and Kwak, Y.H. (2012). An Open Government Maturity Model for social media-based public engagement, Government Information Quarterly, Volume 29, Issue 4, October 2012, Pages 492-503.

Leidner, D. and Kayworth, T. (2006). A Review of Culture in Information Systems Research: Toward A Theory of Information Technology Culture Conflict, *MIS Quarterly*, vol. 30, no. 2, pp 357-399.

Lemon (2022). Upwork review. how legit is Upwork? "Lemon is the exclusive community of startup sidekicks". (<u>https://lemon.io/upwork-review</u>).

Lewandowsky, S., Smillie, L., Garcia, D., Hertwig, R., Weatherall, J., Egidy, S., Robertson, R.E., O'connor, C., Kozyreva, A., Lorenz-Spreen, P., Blaschke, Y. and Leiser, M. (2020). Technology and Democracy: Understanding the influence of online technologies on political behaviour and decision-making. (2020), EUR 30422 EN, Publications Office of the European Union, Luxembourg, 2020, ISBN 978-92-76-24089-1, doi:10.2760/593478, JRC122023.

Li, J. (2020). The digital transformation of business models in the creative industries: A holistic framework and emerging trends, *Technovation*, vol. 92, p. 102012, 2020.

Lipsey, D (2005), Too much choice, Prospect. December 2005, pp. 26-29.

Local Energy Scotland (2022). Edinburgh Community Solar Cooperative. (<u>https://localenergy.scot/casestudy/edinburgh-community-solar-cooperative</u>).

Local Government Association (2022). 'The best thing that ever happened'.

Long, C.K., Agrawal, R., Trung, H.Q., Pham, H.v. (2021). A big data framework for E-Government in Industry 40, August 16, 2021, published by De Gruyter Open Access (<u>https://www.degruyter.com/document/doi/10.1515/comp-2020-0191/html</u>).

Loumeau, G. and Russo, A. (2022), Second-hand gentrification: Theory and evidence from high-speed rail extensions, CESifo Working Paper 9992.

Lovelock, J. (1979). Gaia: A New Look at Life on Earth. Oxford University Press.

Lumineau, F., Schilke, O., Wang, W. (2022). Rethinking trust in the age of digitisation, 5 December 2022, London School of Economics. (<u>https://blogs.lse.ac.uk/businessreview/2022/12/05/rethinking-trust-in-the-age-of-digitisation</u>).

MacKerron, G. (2012). *Happiness and environmental quality*. PhD thesis, London School of Economics and Political Science. (<u>http://etheses.lse.acuk/383</u>).

Maes, J., Quaglia, A., Martinho Guimaraes Pires Pereira, A., Tokarski, M., Zulian, G., Marando, F. and Schade, S. (2021). BiodiverCities: A roadmap to enhance the biodiversity and green infrastructure of European cities by

2030, EUR 30732 EN, Publications Office of the European Union, Luxembourg, ISBN 978-92-76-38642-1, doi:10.2760/288633, JRC125047.

Mair D., Smillie L., La Placa G., Schwendinger F., Raykovska M., Pasztor Z., van Bavel R. (2019). Understanding our political nature: How to put knowledge and reason at the heart of political decision-making, EUR 29783 EN, Publications Office of the European Union, Luxembourg, 2019, ISBN 978-92-76-08621-5, doi:10.2760/374191, JRC117161.

Manzoni, M., Medaglia, R., Tangi, L., et al. (2022), *AI Watch, road to the adoption of artificial intelligence by the public sector : a handbook for policymakers, public administrations and relevant stakeholders*, European Commission, Joint Research Centre, Publications Office of the European Union. (<u>https://data.europa.eu/doi/10.2760/288757</u>).

Markantoni, M. (2016). Low carbon governance: Mobilizing community energy through top-down support? Environmental Policy and Governance, 26(3), 155–169.

March, H., Ribera-Fumaz, R. (2016). Smart contradictions: The politics of making Barcelona a self-sufficient city, *European Urban and Regional Studies*, 2016, Vol. 23(4) 816–830.

Marshall, AP and O'Neill (2018). The Bristol Pound: A tool for localisation? Ecological Economics, 146.

Martens, B., Aguiar, L., Gomez-Herrera, E. and Mueller-Langer, F. (2018). The digital transformation of news media and the rise of disinformation and fake news - An economic perspective; Digital Economy Working Paper 2018-02; JRC Technical Reports.

Matacena, R., Zenga, M., D'Addario, M., Mari, S., and Labra, M. (2021). COVID-19 as an Opportunity for a Healthy-Sustainable Food Transition. An Analysis of Dietary Transformations during the First Italian Lockdown. *Sustainability*, *13*(10), 5661.

Mazzucato, M. (2013) The entrepreneurial state: debunking public vs. private sector myths, Anthem, London.

Mazzucato, M. (2019). Governing Missions in the European Union, June 2019, Independent Expert Report, Europoaen Commission. (<u>https://research-and-innovation.ec.europa.eu/system/files/2019-07/ec_rtd</u> <u>mazzucato-report-issue2_072019.pdf</u>).

McAfee A and Brynjolfsson E (2017), Machine, Platform, Crowd: Harnessing Our Digital Future. New York: W.W. Norton and Company.

McDermott, P. (2012). Building open government, *Government Information Quarterly*, Volume 27, Issue 4, October 2010, Pages 401-413.

McDowell, L. (2016). Migrant Women's Voices: Talking about Life and work in the UK since 1945, published by Bloomsbury Press, London.

McEachern, M. G., Warnaby, G., and Moraes, C. (2021). The role of community-led food retailers in enabling urban resilience. *Sustainability*, 13(14), 7563, <u>https://doi.org/10.3390/su13147563</u>.

McGarvey, D. (2022). The social distance between us; how remote politics wrecked Britain. Ebury Publishing, London.

McGreal, S. Adair, A., Berry, J., Deddis, B. and Hirst, S. (2000). Accessing private sector finance in urban regeneration: investor and non-investor perspectives, *Journal of Property Research*, 17(2), pp 110-124.

McKay, D.A. (2022). Climate tipping points could lock in unstoppable changes to the planet. How close are they? *The Conversation*, October 8, 2022. (<u>https://phys.org/news/2022-10-climate-unstoppable-planet.html</u>).

McKinsey (2017). Government productivity -- unlocking the \$3.5 trillion opportunity. McKinsey Center for Government, discussion paper, April 2017.

McKinsey (2020). COVID-19 and the employee experience: How leaders can seize the moment, 29 June 2020. (<u>https://www.mckinsey.com/capabilities/people-and-organizational-performance/our-insights/covid-19-and-the-employee-experience-how-leaders-can-seize-the-moment</u>).

Meijer, A.J.; Lips, M.; Chen, K. (2019). Open Governance: A New Paradigm for Understanding Urban Governance in an Information Age. *Front. Sustain. Cities*, 2019, 1, 3.

Mergel, I., Ulrich, P., Kuziemski, M. and Martinez, A. (2022) Scoping GovTech dynamics in the EU, EUR 30979 EN, Publications Office of the European Union, Luxembourg, ISBN 978-92-76-47059-5, doi:10.2760/700544, JRC128093.
Messenger, J. C., and Gschwind, L. (2016). Three generations of Telework: New ICT s and the (R) evolution from Home Office to Virtual Office. New Technology, Work Employment, 31(3), 195-208.

Meuleman, L. (2008). Public Management and the Metagovernance of Hierarchies, Networks and Markets: The Feasibility of Designing and Managing Governance Style Combinations. Berlin/Heidelberg: Springer Science & Business Media.

Meyers, G. (2022). What is the Difference Between "Regenerative" and "Sustainable"?, *Medium*, 19 March 2022. (<u>https://medium.com/the-regenerative-transition/what-is-the-difference-between-regenerative-and-sustainable-f6c042985f11</u>).

Millard, J (2000). Making Telework Work For All. (<u>http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.133.5312&rep=rep1&type=pdf</u>).

Millard, J. (2001). Good practice in ICT work and skills, Information Society Technologies Programme of the European Commission.

Millard, J. and Kubieck, H. (2004). Reorganisation of government back-offices for better electronic public services: European good practices (back-office reorganisation), prepared for the European Commission eGovernment Unit, Brussels, January 2004.

Millard, J., Shahin, J., et al (2006). Towards the eGovernment vision for EU in 2010: research policy challenges, for the Institute of Prospective Technological Studies, Seville, Spain, European Commission, JRC.

Millard, J., Shahin, J., et al (2007). Study for the Impact Analysis of FP5 e-Government projects, under the WING Framework Contract for Impact Analysis, for the European Commission, DG INFSO, April 2007.

Millard, J. (2010). Just a-walking down the street – everyday eGovernment, Proceedings of the 4th International Conference on E-Democracy, OCG-Austrian Computer Society, Vienna, April 2010.

Millard, J. (2014), 'Development theory', in Jürgen Howaldt, Anna Butzin, Dimitri Domanski and Christoph Kaletka (eds), *Theoretical Approaches to Social Innovation - A Critical Literature Review, a deliverable of the project "Social Innovation: Driving Force of Social Change" (SI-DRIVE),* Dortmund: Sozialforschungsstelle, pp.34-58.

Millard J, Lauritzen JRK, Simon J et al (2014): Doing Social Innovation. A Guide for Practitioners, a deliverable of the project: The theoretical, empirical and policy foundations for building social innovation in Europe (TEPSIE), European Commission – 7th Framework Programme, Brussels: European Commission, DG Research.

Millard, J., Carpenter, G. (2014). Digital technology in social innovation: synthesis, gaps and recommendations, a deliverable of the project: "The theoretical, empirical and policy foundations for building social innovation in Europe" (TEPSIE), European Commission – 7th Framework Programme, Brussels: European Commission, DG Research.

Millard, J (2015a). Open governance systems: Doing more with more, *Government Information Quarterly*, 12 September 2015, <u>http://doi.org/10.1016/j.giq.2015.08.003</u>.

Millard, J (2015b) "The digital divide and the post-2015 development debate" in *Digital Divides: The New Challenges and Opportunities of e-Inclusion,* Taylor and Francis Publishing Group, Abingdon, UK.

Millard, J. (2017a). European Strategies for e-Governance to 2020 and beyond, in Ojo, A., and Millard, J. (eds) "Government 3.0 – Next Generation Government Technology Infrastructure and Services: Roadmaps, Enabling Technologies and Challenges", Springer Public Administration and Information Technology.

Millard, J. (2017b). Technology innovations in public service delivery for sustainable development, in Ojo, A, and Millard, J. (eds) "Government 3.0 –Next Generation Government Technology Infrastructure and Services: Roadmaps, Enabling Technologies & Challenges", Springer Public Administration and Information Technology.

Millard J (2017c) Cities and Urban Living at the Crossroads, in "Digital Nations – Smart Cities, Innovation and Sustainability", Volume 10595 of the Lecture Notes in Computer Science series, Springer Nature, Chennai, India.

Millard J, Holtgrewe U, Hochgerner J (2017) Addressing social demands, societal challenges and systemic change, in Howaldigital technology J, Schröder A, Butzin A, Rehfeld D (Editors) "Towards a General Theory and Typology of Social Innovation", Technical University of Dortmund.

Millard, J., Thomasen, L., Pastrovic, G., Cvetkovic, B. (2018). A roadmap for e-participation and open government empirical evidence from the Western Balkans. Proceedings of the 11th International Conference on Electric Government, April 2018, doi:10.1145/3209415.3209459.

Millard, J., Noring, L., Farinea and C., Sorivelle, M.N. (2019a). Nature-based solutions for resilient and sustainable cities, in book Albert, S (ed) "Innovative solutions for creating sustainable cities", Cambridge Scholars Publishing, Newcastle upon Tyne, UK

Millard, J., Noring, L., and Sorivelle, M.N. (2019b). The role of civil society in city governance and financing, in book Albert, S (ed) "Innovative solutions for creating sustainable cities", Cambridge Scholars, Cambridge, UK.

Millard, J. (2020). Impacts of COVID-19 on social development and implications for the just transition to sustainable development. Prepared for UNDESA's Virtual Expert Group Meeting on the "Socially just transition towards sustainable development: The role of digital technologies on social development and well-being of all", 4-7 August 2020. (<u>https://www.un.org/development/desa/dspd/wp-content/uploads/sites/22/2020/08/Impacts-of-COVID-19-on-social-development-and-implications-for-the-just-transition-to-sustainable-development-4-8-20.pdf.)</u>

Millard, J., Farina, C. and Steinhauer, J. (2021). Evaluation strategies for the regenerative and restorative circular city, in Albert and Pandey, (eds) "Place-based performance metrics in building sustainable cities", Routledge Taylor and Francis.

Millard, J. (2021). How 'zoomshock' is transforming work, geography and society. (<u>http://globalforum.items-int.com/gf/gf-content/uploads/2021/06/Jeremy-Millard-GF2021-webinar-9-6-21.pdf</u>).

Millard, J. (2022). The Geo-Demographics of European Consumers' Food Behaviour in the New Age of Disruption, in Ayman, U (ed) "A New Era of Consumer Behavior - Beyond the Pandemic, IntechOpen, doi: 10.5772/intechopen.106938."; https://www.intechopen.com/chapters/83502

Millard, J., Sturla, A., Smutná, Z., Duží, B., Janssen, M. and Vávra J. (2022). European Food Systems in a Regional Perspective: A Comparative Study of the Effect of COVID-19 on Households and City-Region Food Systems. *Front. Sustain. Food Syst.* 6:844170. doi:10.3389/fsufs.2022.844170.

Millard, J. (2023). Exploring the impact of digital transformation on public governance - Prevailing (public) governance models throughout Europe, Schade, S. and Manzoni, M. (Editors), Publications Office of the European Union, Luxembourg, doi:10.2760/219929, JRC133791.

Miller, C. (2019). Taiwan is making democracy work again. It's time we paid attention. *Wired Magazine*, 26 November 2019. (<u>https://www.wired.co.uk/article/taiwan-democracy-social-media</u>).

Miller, C. (2020). How Taiwan's civic hackers helped find a new way to run the country, Guardian Newspaper, 27 September 2020. (<u>https://www.theguardian.com/world/2020/sep/27/taiwan-civic-hackers-polis-consensus-social-media-platform</u>).

Miron, M., Tolan, S., Gómez, E. et al. (2021). Evaluating causes of algorithmic bias in juvenile criminal recidivism. Artif Intell Law 29, 111–147, <u>https://doi.org/10.1007/s10506-020-09268-y</u>.

Misuraca, G., Pasi. G. and Viscusi. G. (2018). Social Innovation and Resilience: exploring the dynamics and impacts on the digital transformation of governance & society, The Proceedings of the 10th International Conference on the Theory and Practice of Electronic Governance (ICEGOV2018), Galway, Ireland, the ACM Press.

Misuraca, G., Barcevičius, E., Codagnone, C., (Eds.). (2020) Exploring Digital Government Transformation in the EU – Understanding public sector innovation in a data-driven society, EUR 30333 EN, Publications Office of the European Union, Luxembourg, 2020, ISBN 978-92-76-21326-0, doi:10.2760/480377, JRC12154.

Mitchell, C.J.A. and Bryant, C.R. (2020) Counter urbanization, in International Encyclopaedia of Human Geography (Second Edition). Elsevier, Amsterdam.

Mobile-Age Project (2019). Effective use of open government data to transform the lives of Europe's elderly citizens, H2020 Project. (<u>https://cordis.europa.eu/article/id/386889-effective-use-of-open-government-data-to-transform-the-lives-of-europe-s-elderly-citizens</u>).

Moffitt, S. (2022). Futureproofing – the future beyond innovation, *Futureproofing*. (<u>https://futureproofingnext</u>.<u>.com/thebook/</u>).

Møller-Jensen, J., Jensen-Butler, C., Madsen, B., Millard, J. and Schmidt, L. (2008). A web-based study of the propensity to telework based on socio-economic, work organisation and spatial factors, in road pricing, the economy and the environment: Springer, 2008.

Monbiot, G. (2014). Feral: Rewilding the Land, the Sea, and Human Life, University of Chicago Press.

Monbiot, G. (2019). There is an antidote to demagoguery – it's called political rewilding, in the Guardian Newspaper, 18 December 2019. (<u>https://www.thequardian.com/commentisfree/2019/dec/18/demagogues-power-rewilding-party-trust-power-government</u>).

Monbiot, G. (2022). Regenesis – Feeding the World without Devouring the Planet, Penguin Books.

Moragues-Faus, A. and Battersby, J. (2021). Urban food policies for a sustainable and just future: Concepts and tools for a renewed agenda. *Food Policy*, 103, special issue "Urban food policies for a sustainable and just future". (<u>https://www.sciencedirect.com/journal/food-policy/vol/103/suppl/C)</u>.

Mosinegutu, V. (2022). What is Web 3.0 and How Will It Change the Internet? (<u>https://academy.coinzilla.com/what-is-web-3-0</u>).

Moore, M.H, (1995) "Creating Public Value: Strategic Management in Government", Cambridge, Massachusetts: Harvard University Press.

MUFPP, Milan Urban Food Policy Pact. (2015). The Milan Urban Food Policy Pact and the New Urban Agenda. (<u>https://habitat3.org/wp-content/uploads/press_files/cXxx6dJ8HJQDw4bYZP.pdf</u>).

MUFPP, Milan Urban Food Policy Pact (2018). Milan Urban Food Policy Pact Monitoring Framework Monitoring Framework, July 2018 version. (<u>https://www.fao.org/3/cb4030en/cb4030en.pdf</u>).

Mullainathan, S. and Shafir, E. (2013). Scarcity: why having too little means so much, Times Books, New York.

Muller, J. (2018). The Tyranny of Metrics, Princeton University Press.

Murphy, B., Benson, T., McCloat, A., Mooney, E., Elliott, C., Dean, M., and Lavelle, F. (2021). Changes in consumers' food practices during the covid-19 lockdown, implications for diet quality and the food system: A cross-continental comparison. *Nutrients*, 13(1), 1–14, <u>https://doi.org/10.3390/nu13010020</u>.

Nesta (2017). Governing with Collective Intelligence. January 2017, Nesta Innovation Foundation, UK (<u>https://media.nesta.org.uk/documents/governing_with_collective_intelligence.pdf</u>).

Nesta (2019a). Public Value -- how can it be measured, managed and grown? May 2019, Nesta Innovation Foundation, UK (<u>https://media.nesta.org.uk/documents/Public Value WEB.pdf</u>).

Nesta (2019b). Our futures: by the people, for the people – how mass involvement in shaping the future can solve complex problems. November 2019, Nesta Innovation Foundation, UK. (<u>https://www.nesta.org.uk/report/our-futures-people-people</u>).

New Zealand Government (2018). Better Rules for Government Discovery Report, March 2018. (<u>https://www.digitalgovt.nz/dmsdocument/95-better-rules-for-government-discovery-report/html</u>).

Niedlich, S., Baue, M., Doneliene, M., Jaeger, L., Rieckmann, M. and Bormann, I. (2020). Assessment of Sustainability Governance in Higher Education Institutions—A Systemic Tool Using a Governance Equalizer. *Sustainability* 2020, 12, 1816; doi:10.3390/su12051816.

Niehaves, B. (2007). Innovation Processes in the Public Sector – New Vistas for an Interdisciplinary Perspective on E-Government Research?, Electronic Government, vol. LCNS 4656, pp. 23-34, Springer.

Noring, L. (2016). Teaching material from CBS Course "*From Supply Chains to Circular Economy*". November 2016.

Noring, L. (2018). Financing the Inclusive City, ICLEI. (<u>http://iclei-europe.org/publications-tools/?c=search&uid=qDCrxJk2</u>).

Noring, L., Millard, J. and Sorivelle, M.N. (2019). Urban governance and financing, in book Albert, S (ed) "Innovative solutions for creatiung sustainable cities", Cambridge Scholars Publishing, Newcastle upon Tyne, UK

Nuffield Trust, (2020). A new era for A&E targets: what will be the impact of the new basket of measures? (<u>https://www.nuffieldtrust.org.uk/news-item/a-new-era-for-a-e-targets-what-will-be-the-impact-of-the-new-basket-of-measures</u>).

OECD (2005). eGovernment for better government. OECD, Paris.

OECD (2011). The relationship between local content, internet development and access prices, OECD, Internet Society & UNESCO. (<u>https://www.oecd.org/digital/ieconomy/50305352.pdf</u>).

OECD (2014a). SIGMA Principles of Public Administration. (<u>https://www.sigmaweb.org/publications/principles-public-administration.htm</u>).

OECD (2014b). Recommendation of the Council on Digital Government Strategies. Public Governance and Territorial Development Directorate. (<u>https://www.oecd.org/gov/digital-government/Recommendation-digital-government-strategies.pdf</u>).

OECD (2020a). What does "inclusive governance" mean?: Clarifying theory and practice. *OECD Development Policy Papers*, No. 27, OECD Publishing, Paris, <u>https://doi.org/10.1787/960f5a97-en</u>.

OECD (2020b). Behavioural insights. (https://www.oecd.org/gov/regulatory-policy/behavioural-insights.htm).

OECD (2020c). Building a resilient recovery: Emerging stronger from the COVID-19 pandemic. (<u>https://www.oecd.org/coronavirus/en</u>).

OECD (2020d). A Systemic Approach to Dealing with Covid-19 and Future Shocks, 28 April 2020 (<u>https://www.oecd.org/naec/projects/resilience/NAEC Resilience and Covid19.pdf</u>).

OECD (2020e). Coronavirus: The world economy at risk, (<u>https://www.oecd.org/berlin/publikationen/Interim-</u> Economic-Assessment-2-March-2020.pdf).

OECD (2020f). Responding to COVID-19: the rules of good governance apply now more than ever. (<u>https://www.oecd.org/governance/public-governance-responses-to-covid19</u>).

OECD (2022). Tax policy reforms 2022. (<u>https://www.oecd.org/tax/tax-policy-reforms-26173433.htm</u>).

O'Flynn, J. (2007). From New Public Management to Public Value: Paradigmatic Change and Managerial Implications. *The Australian Journal of Public Administration*, vol. 66, no. 3, pp. 353–366.

O'Reilly, T. (2010). Government as a platform, in the book "Open government", Lathrop, D., and Ruma, L. (eds). (<u>https://www.oreilly.com/library/view/open-government/9781449381936</u>).

Ortiz-Ospina, E., and Roser, M. (2016). Taxation, Our World in Data. (https://ourworldindata.org/taxation).

Osborne, S.P. (2006). The new public governance? *Public Management Review*, 8: 377–87.

Osička, J. and Černochac, F. (2022). European energy politics after Ukraine: The road ahead, *Energy Research & Social Science*, Volume 91, September 2022, 102757, Elsevier ScienceDirect (https://www.sciencedirect.com/science/article/abs/pii/S2214629622002602).

Oxfam (2016). An economy for the 1% -- how privilege and power in the economy drive extreme inequality and how this can be stopped, 210 Oxfam Briefing Paper, Oxfam International 18 January 2016.

Palomino, J.C., Rodríguez, J.G. and Sebastian, R. (2020). Inequality and poverty effects of the lockdown in Europe. CEPR, 16 June 2020. (<u>https://voxeu.org/article/inequality-and-poverty-effects-lockdown-europe</u>).

Pariser E (2011). The filter bubble: how the new personalized web is changing what we read and how we think, the Penguin Press.

Parsons, K., Lang, T., and Barling, D. (2021). London's food policy: Leveraging the policy sub-system, programme and plan. *Food Policy*, 103, 102037, <u>https://doi.org/10.1016/j.foodpol.2021.102037</u>.

Passani A, Bellini F and Vanobberghen W (Eds.) (2016): Exploring Impacts of Collective Awareness Platforms for Sustainability and Social Innovation, Rome: Eurokleis Press (Funded by EC, IA4SI Project).

Perez, C. (2010). Technological revolutions and techno-economic paradigms. *Cambridge Journal of Economics*, 34(1), 185-202.

Periscope Project (2022). Best Practice in Multi-Level Governance During Pandemics: A Case Study Report, Deliverable 9.1 of the Periscope Project (Pan-European Response to the ImpactS of COVID-19 and future Pandemics and Epidemics), 27-06-22. (<u>https://www.feam.eu/wp-content/uploads/Bear best practice in multi level published.pdf</u>).

Peters, P., Ligthart, P. E., Bardoel, A., and Poutsma, E. (2016). Fit'for telework? Cross-cultural variance and taskcontrol explanations in organizations' formal telework practices. *The International Journal of human resource management*, 27(21), 2582-2603.

Piketty, T. (2014). *Capital in the Twenty-First Century.* Translated from the French edition 2013. Cambridge: Harvard University Press.

Plunkett, J. (2022). The platform as the prototype -- Should the state copy big tech — or should we invent something new? Medium, 2 June 2022. (<u>https://medium.com/predict/the-platform-as-the-prototype-1a820d3f49ab</u>).

Politico (2022). Germany and France join forces against Biden in subsidy battle, 22 November 2022. (<u>https://www.politico.eu/article/germany-france-biden-green-subsidy-inflation-reduction-act-robert-habeck-bruno-le-maire</u>).

Pollitt, C. (1990). Managerialism and the Public Service: The Anglo-American Experience. Cambridge: Basil Blackwell.

Pollitt, C., and Bouckaert, G. (2011). Public Management Reform: A Comparative Analysis-New Public Management, Governance, and the Neo-Weberian State, 3rd ed. Oxford: Oxford University Press.

Porter, M., and Kramer, M. (2011). Creating shared value. *Harvard Business Review*, 89(1/2), Jan/Feb 2011, 62-77.

Portuguese Presidency (2021). Lisbon Declaration – Digital Democracy with a Purpose, to reinforce the "European way of doing business" as a world-wide recognised value proposition and a unique competitiveness advantage, 1 June 2021. (<u>https://www.lisbondeclaration.eu/#:~:text=The%20%E2%80%9CLisbon%20</u> Declaration%20%E2%80%93%20Digital%20Democracy.new%20paradigm%20of%20digital%20transition).

Presidence Francaise (2022). Strasbourg Declaration, outlining the coordination strategy for public administrations, 25 February 2022. (<u>https://presidence-francaise.consilium.europa.eu/media/n3xpjy4x/25-02-2022 declaration strasbourg en.pdf</u>).

Preston City Council (2022). What is Community Wealth Building and the Preston Model? (<u>https://www.preston.gov.uk/article/1335/What-is-Community-Wealth-Building</u>).

PwC Consulting (2019). The three trends defining the governments of the future. 13 November 2019. (<u>https://www.pwc.com.au/digitalpulse/government-5-0-whole-of-life-service.html</u>).

Raworh, K. (2017). Doughnut economics: seven ways to think like a 21st Century economist. Penguin, Random House, UK.

Reform Thinktank (2017). Work in progress. Towards a leaner, smarter public-sector workforce, 6 February 2017. (<u>https://reform.uk/publications/work-progress-towards-leaner-smarter-public-sector-workforce</u>).

Regenesys Business School (2020). The Fifth Industrial Revolution (5IR) and how it will change the business landscape, September 8, 2020. (<u>https://www.regenesys.net/reginsights/the-fifth-industrial-revolution-5ir</u>).

Repette, P., Sabatini-Marques, J., Yigitcanlar, T., Sell, D. and Costa, E. (2021). The Evolution of City-as-a-Platform: Smart Urban Development Governance with Collective Knowledge-Based Platform Urbanism. *Land* 2021, 10, 33, <u>https://doi.org/10.3390/land10010033</u>.

Rhodes, J. A. W. (2021). Beyond new public governance. In book: Politics, Policy and Public Administration in Theory and Practice: Essays in honour of Professor John Wanna (pp.235-261), Australian National University Press. (<u>https://www.researchgate.net/publication/352411205_Beyond_new_public_governance</u>).

Rizzi, D. (2021). How Nature Based Solutions can contribute to Resilience, ICLEI (Local Governments for Sustainability, Europe), MCR (Making Cities Resilient) 2030 Webinar, 9 December 2021. (https://mcr2030.undrr.org/sites/default/files/inline-files/21-12-08 ICLEI DanielaRizzi MCR2030 final.pdf).

Rutgers (2019). Building the Assets of Low and Moderate Income Workers and their Families - The Role of Employee Ownership, Institute for the Study of Employee Ownership and Profit Sharing, March 2019. (https://smlr.rutgers.edu/sites/smlr/files/Documents/News/rutgerskelloggreport_april2019.pdf).

Schade, S., Tsinaraki, C., Manzoni, M., Berti Suman, A., Spinelli, F., Mitton, I., Kotsev, A., Delipetrev, B. and Fullerton, K. (2020). Activity Report on Citizen Science – discoveries from a five year journey, EUR 30551 EN, Publications Office of the European Union, Luxembourg, 2020, ISBN 978-92-76-28369-0, doi:10.2760/172609, JRC123500.

Scharfbillig, M., Smillie, L., Mair, D., Sienkiewicz, M., Keimer, J., Pinho Dos Santos, R., Vinagreiro Alves, H, Vecchione, E. and Scheunemann L., (2021). Values and Identities - a policymaker's guide — Executive summary, EUR 30800 EN, Publications Office of the European Union, Luxembourg, 2021, ISBN 978-92-76-42444-4, doi:10.2760/128665, JRC126150.

Schmid, B., Meister, T., Klagge, B, and Seidl, I. (2020). Energy Cooperatives and Municipalities in Local Energy Governance Arrangements in Switzerland and Germany. *Journal of Environment & Development*, 2020, Vol 29(1) 123–146, doi:10.1177/1070496519886013.

Scholl, J. (2005). Digital government, the challenges of integration and interoperability, Proceedings of dg.o2005, the 6th National Conference on Digital Government Research, Atlanta, Georgia, USA, 15-18 May, 2005.

Schraad-Tischler, D., Seelkopf, L. (2016). Sustainable Governance Indicators 2016, concepts and methodology. BertelsmannStiftung. (<u>https://www.bertelsmann-stiftung.de/fileadmin/files/BSt/Publikationen/GrauePublikationen/GrauePublikationen/GrauePublikationen/Studie NW Policy Performance and Governance Capacities in the OECD and EU 2016.pdf).</u>

Schwalb, L., and Walk, H. (2007). Blackbox governance—Lokales engagement im Aufwind [Blackbox governance–Local engagement on the rise]. In L. Schwalb and H. Walk (Eds.), Local Governance — mehr Transparenz und Bürgernähe?? (pp. 7–22).Wiesbaden, Germany: VS Verlag für Sozialwissenschaften.

Schwab, K. (2017). The Fourth Industrial Revolution, World Economic Forum.

Schwendinger, F., Topp, L. and Kovacs, V. (2022) Competences for Policymaking — Competence Frameworks for Policymakers and Researchers working on Public Policy (2022), EUR 31115 EN, Publications Office of the European Union, Luxembourg, 2022, ISBN 978-92-76-53454-9, doi:10.2760/642121, JRC129623.

Scott. R. and Witter, L. (2017). Bridging Governments' Borders, *Stanford Social Innovation Review*, Aug. 16, 2017. (<u>https://ssir.org/articles/entry/bridging_governments_borders</u>).

Seetharaman, P. (2020). Business models shifts: Impact of Covid-19, *International Journal of Information Management*, vol. 54.

Shanley, L., Parker, A., Schade, S., and Bonn, A. (2019). Policy Perspectives on Citizen Science and Crowdsourcing. Citizen Science: Theory and Practice, 4(1), 30. DOI: <u>http://doi.org/10.5334/cstp.293</u>.

Sharifi, A., and Khavarian-Garmsir, A. R. (2020). The COVID-19 pandemic: Impacts on cities and major lessons for urban planning, design, and management. *Science of The Total Environment*, 749, 142391. https://doi.org/10.1016/j.scitotenv.2020.142391.

Sharpe, S. and Lenton, T.M. (2021). Upward-scaling tipping cascades to meet climate goals: plausible grounds for hope, *Climate Policy*, Volume 21, 2021 – Issue 4. Taylor and Francis. (<u>https://www.tandfonline.com/doi/abs/10.1080/14693062.2020.1870097</u>).

Sintomer, Y., Gerzberg, C. and Röcke, A. (2008). Participatory Budgeting in Europe: Potentials and Challenges, *International Journal of Urban and Regional Research*, 13 March 2008, <u>https://doi.org/10.1111/j.1468-2427.2008.00777.x</u>.

Sintomer, Y., Röcke, A. and Herzberg, C. (2016). Participatory Budgeting in Europe: Democracy and public governance., Routledge, London: <u>https://doi:10.4324/9781315599472</u>.

SITRA (2018). Phenomenon-based public administration: discussion paper on reforming the government's operating practices. SITRA Working Paper 31.8.2018. (<u>https://www.sitra.fi/en/publications/phenomenonbased-public-administration</u>).

Smart Cities World (2019). Half public sector say they do "nowhere near enough" to involve citizens. (<u>https://www.smartcitiesworld.net/news/half-public-sector-say-they-do-nowhere-near-enough-to-involve-citizens-4295</u>).

Smith, G. (2009). Democratic Innovations: Designing Institutions for Citizen Participation. Theories of Institutional Design. Cambridge: Cambridge University Press. ISBN 978-0-521-51477-4.

Smith, R. (2003). Focusing on public value: Something old and something new. Victoria, Australia: Monash University.

Social Care Institute for Excellence (2022). Wigan Community Link Worker Scheme, Lead service provider/commissioner. (<u>https://www.scie.org.uk/prevention/research-practice/getdetailedresultbyid?id=a11G00</u> 0000MGeqGIAT).

Sørensen, E., Torfling, J. (2015). Enhancing Public Innovation through Collaboration, Leadership and New Public Governance, chapter in "New Frontiers in Social Innovation Research", edited by Nicholls, A., Simon, J., Gabriel, M. Palgrave & Macmillan, Basingstoke, UK. (<u>https://library.oapen.org/bitstream/handle/20.500.12657/</u>27885/1002117.pdf?sequence=1#page=171).

Steffen, W., Richardson, K., Rockström, J., Cornell, S., Fetzer, I., Bennett, E., Biggs, R., Carpenter, S., De Vries, W., De Wit, C. and Folke, C. (2015). Planetary boundaries: Guiding human development on a changing planet. science, 347(6223), doi:10.1126/science.1259855, 1259855.

Stephenson, P. (2013). Twenty years of multi-level governance: Where Does It Come From? What Is It? Where Is It Going?, *Journal of European Public Policy*, 20:6, 817-837, doi:10.1080/13501763.2013.781818.

Stern, N. (2007). The Stern Review: The Economics of Climate Change. Cambridge University Press, Cambridge.

Stoker, G. (2006). Public Value Management: A New Narrative for Networked Governance?, The American Review of Public Administration, 3(1), 41-57.

Stone, C. (2013). False economies, Part 1: decoding efficiency, landmark report for Effective Government, *The Centre for Policy Development*, Sydney, NSW, Australia.

Strassheim, H. (2021). Who are behavioural public policy experts and how are they organised globally? Policy & Politics, 49(1), 69–86: <u>https://doi.org/10.1332/030557320x15956825120821</u>.

Stupak, I., Mansoor, M., Tattersall Smith, C. (2021). Conceptual framework for increasing legitimacy and trust of sustainability governance. *Energy, Sustainability and Society,* 11, 5. (<u>https://energsustainsoc.biomedcentral.com/articles/10.1186/s13705-021-00280-x</u>).

Šucha, V. & Sienkiewicz, M. (2020). Science for Policy Handbook, Elsevier, ISBN 978-0-12-822596-7, doi:<u>https://doi.org/10.1016/C2018-0-03963-8</u>.

Tang, C. (2019). Exploring the Mondragon cooperative system. *Beneficial State Foundation Perspectives*, 12 July 2019. (<u>https://beneficialstate.org/perspectives/exploring-the-mondragon-cooperative-system</u>).

Taylor Aiken, G. (2016). Polysemic, polyvalent and phatic: A rough evolution of community with reference to low carbon transitions. People, Place and Policy, 10(2), 126-145.

Telecom Review (2022). Government 4.0: Transformation to Digital Government, 4 May 2022. (<u>https://www.telecomreview.com/articles/reports-and-coverage/6108-government-4-0-transformation-to-digital-government</u>).

Thaler, R.H. and Sunstein, C.R. (2008). Nudge: Improving Decisions about Health, Wealth, and Happiness, Yale University Press.

Thaler, R.H, and Sunstein, C.R. (2021). Nudge: the final edition, Penguin Books.

Thisse, J.F., Kichko, S., Matheson, J., Gokan, T. (2022). Teleworking will reshape labour markets and cities, Centre for Economic Policy Research (CEPR), 6 December 2022. (<u>https://cepr.org/voxeu/columns/teleworking-will-reshape-labour-markets-and-cities?utm_source=sfmc&utm_medium=email&utm_campaign=2790721</u> <u>Agenda_weekly-9December2022&utm_term=&emailType=Agenda%20Weekly</u>).</u>

Torfing, J., Bøgh Andersen, L., Greve, C. and Klausen, K.K. (2020). Competing and Co-Existing Policy. Administrative and Institutional Change series, Edward Elgar Publishing, Cheltenham, UK.

Torfing, J., Ansell, C. (2021). Co-creation: the new kid on the block in public governance, Department of Social Sciences and Business, Policy and Politics, RUC, Denmark, doi:10.1332/030557321x16115951196045.

Triple Sustainability (2020). What is sustainability governance. (<u>https://sustainability-governance.com/what-is-sq</u>).

Tummers, L. (2020). Behavioral Public Administration. Published online by Oxford University Press, 30 June 2020. (<u>https://oxfordre.com/politics/view/10.1093/acrefore/9780190228637.001.0001/acrefore-9780190228637-e-1443</u>.

UNEP, United Nations Environmental Programme (2011). Towards a Life Cycle Sustainability Assessment making informed choices on products. Life Cycle Initiative, Sustainable Consumption and Production Branch, UNEP, Paris.

UNESCO (2016). Knowledge Societies Policy Handbook. (<u>https://en.unesco.org/sites/default/files/knowledge socities policy handbookpdf</u>).

United Nations (2009). What is good governance, 10 July 2009, UNESCAP. (<u>https://www.unescap.org/resources/what-good-governance</u>).

United Nations (2015a). Global and National Leadership in Good Governance. UN Chronical, April 2015, No. 4 Vol. LII, Implementing the 2030 Agenda: The Challenge of Conflict:. (<u>https://www.un.org/en/chronicle/article/global-and-national-leadership-good-governance#:~:text=The%20most%20cited%20definition%20 has,follows%20the%20rule%20of%20law</u>).

United Nations (2015b). Transforming our world: the 2030 Agenda for Sustainable Development, New York. (<u>https://www.un.org/en/development/desa/population/migration/generalassembly/docs/globalcompact/A_RES_7_0_1_E.pdf</u>).

United Nations (2015c). 'The Millennium Development Goals Report', New York, 2015.

UNDP, United Nations Development Programme. (2020). COVID-19 and Human Development: Assessing the Crisis, Envisioning the Recovery, 10.11.2020. (<u>http://hdr.undp.org/sites/default/files/covid-19 and human development 0.pdf</u>).

UNCTAD, United Nations Conference on Trade and Development (2020). Digital platforms and value creation in developing countries: Implications for national and international policies, Geneva, 29 April–1 May 2020. (https://unctad.org/system/files/official-document/tdb_ede4d2_en.pdf).

United Nations (2015d). The Paris Agreement. (<u>https://unfccc.int/process-and-meetings/the-paris-agreement/the-paris-agreement</u>).

United Nations Water (2020): Water, food and energy. (<u>https://www.unwater.org/water-facts/water-food-and-energy</u>).

University of Hamburg (2022). Co-creative governance expands scope for action. (<u>https://www.uni-hamburg.de/en/transfer/ko-kreative-forschung/ko-kreative-governance.html</u>).

University of Bristol (2020). Informal governance and achieving sustainability – the next steps for Bristol Pound. (<u>https://www.bristol.ac.uk/research/impact/impact-story-bristol-pound</u>).

Uptech Solution (2022). A Journey from Web 0.0 to Web 5.0. (<u>https://medium.com/@uptechsolution/a-journey-from-web-0-0-to-web-5-0-57a58c9b2a09</u>).

Valayer, C., Schade, S., Hernandez Quiros, L., Tsinaraki, C., Pignatelli, F. and Boguslawski, R. (2022). Benchmarking the role of the Public Sector and Location Intelligence in Smart Spaces, EUR 31109 EN, Publications Office of the European Union, Luxembourg, ISBN 978-92-76-53305-4, doi:10.2760/505730, JRC128862

Upwork (2002). How work should work. (https://www.upwork.com).

Van den Abeele, E. (2021). Towards a New Paradigm in Open Strategic Autonomy? ETUI Research Paper - Working Paper 2021.03. (<u>https://papers.ssrn.com/sol3/papers.cfm?abstract_id=3873798</u>).

Van Veenstra, A.F. and Janssen, M. (2012). Investigating Out-comes of T-Government using a Public Value Management Approach, in Scholl HJ et al (eds.), IFIP EGOV 2012, Springer LNCS 7443, Kristiansand, Norway, pp. 187-197.

Vesnic Alujevic, L. and Scapolo, F. (2019). The Future of Government 2030+: Policy implications and recommendations, EUR 29853 EN, Publications Office of the European Union, Luxembourg, 2019, ISBN 978-92-76-11206-8, doi:10.2760/191958, JRC117971.

Vesnic-Alujevic, L., Stoermer, E., Rudkin, J., Scapolo, F. and Kimbell, L. (2019). The Future of Government 2030+: A Citizen-Centric Perspective on New Government Models. EUR 29664 EN. Publications Office of the European Union, Luxembourg. ISBN 978-92-76-00165-2 doi:10.2760/145751, JRC 115008.

Vesnic-Alujevic, L., Muench, S., Stoermer, E., et al. (forthcoming). Reference foresight scenarios Scenarios on the global standing of the EU in 2040, European Commission, JRC.

Vibert, F. (1001). Europe simple corope siron: the future of corope significant of the property of the future of the property of the property

Vilhelmson, B., and Thulin, E. (2001). Is regular work at fixed places fading away? The development of ICT-based and travel-based modes of work in Sweden. *Environment planning A*, 33(6), 1015-1029.

Vittersø, G., Torjusen, H., Laitala, K., Tocco, B., Biasini, B., Csillag, P., de Labarre, M. D., Lecoeur, J. L., Maj, A., Majewski, E., Malak-Rawlikowska, A., Menozzi, D., Török, Á., and Wavresky, P. (2019). Short food supply chains

and their contributions to sustainability: Participants' views and perceptions from 12 European cases. *Sustainability*, 11(17), 4800, <u>https://doi.org/10.3390/su11174800</u>.

Vittuari, M., Bazzocchi, G., Blasioli, S., Cirone, F., Maggio, A., Orsini, F., Penca, J., Petruzzelli, M., Specht, K., Amghar, S., Atanasov, A.-M., Bastia, T., Bertocchi, I., Coudard, A., Crepaldi, A., Curtis, A., Fox-Kämper, R., Gheorghica, A. E, Lelièvre, A., ... De Menna, F. (2021). Envisioning the Future of European Food Systems: Approaches and Research Priorities After COVID-19. *Frontiers in Sustainable Food Systems*, *5*,58, <u>https://doi.org/10.3389/fsufs.2021.642787</u>.

Voorberg, W., Bekkersa, V., Timeusb, K., Tonuristc, P. and Tummers, L. (2017). Changing public service delivery: learning in co-creation. *Policy and Society*, 2017, Vol. 36, No. 2, 178–194, <u>https://doi.org/10.1080/14494035.2017.1323711</u>

Wade, J., Eyre, N., Hamilton, J. and Parag, Y. (2008). Local energy governance: communities and energy efficiency policy. European Council for an Energy Efficient Economy, Summer Study Proceedings. (<u>https://www.eceee.org/library/conference proceedings/eceee Summer Studies/2013/3-local-action-and-national-examples/local-energy-governance-communities-and-energy-efficiency-policy</u>).

Watts, G. (2020). COVID-19 and the digital divide in the UK. The Lancet Digital Health, August 2020, doi:<u>https://doi.org/10.1016/S2589-7500(20)30169-2</u>.

Weber, M. (1946). Essays in Sociology. Oxford: Oxford University Press.

Weerakkody, V. and Dhillon, G. (2008). Moving from E-government to T-government: A study of process reengineering challenges in a UK local authority perspective. International Journal of Electronic Government Research, 4(4), 1–16.

WEF, World Economic Forum (2016). The fourth industrial revolution: what it means and how to respond, World Economic Forum. (<u>http://www.weforum.org/agenda/2016/01/the-fourth-industrial-revolution-what-it-means-and-how-to-respond</u>).

WEF, World Economic Forum (2020). The great reset. (<u>https://www.weforum.org/great-reset</u>).

WEF, World Economic Forum (2022a). Global Risks Report 2022. (<u>https://www.weforum.org/reports/global-risks-report-2022</u>).

WEF, World Economic Forum (2022b). 4 ways leaders can strengthen Europe's economic resilience, 21 May 2022. (<u>https://www.weforum.org/agenda/2022/05/4-ways-leaders-can-strengthen-europe-economic-resilience)</u>.

WEF, World Economic Forum (2022c). 5 strategies to navigate the shifting frontiers of the energy transition, 26 May 2022. (<u>https://www.weforum.org/agenda/2022/05/5-strategies-actions-navigate-energy-transition</u>).

WEF, World Economic Forum (2022d). Jobs of Tomorrow: The Triple Returns of Social Jobs in the Economic Recovery, May 2022. (<u>https://www3.weforum.org/docs/WEF_Jobs_of_Tomorrow_2022.pdf</u>).

WEF, World Economic Forum (2022e). How can hybrid work models prioritize diversity, equity, and inclusion?, 3 May 2022. (<u>https://www.weforum.org/agenda/2022/05/great-renegotiation-hybrid-work-inclusive</u>).

WEF, World Economic Forum (2022f). How can hybrid working drive diversity and productivity?, 26 My 2022. (<u>https://www.weforum.org/agenda/2022/05/how-hybrid-work-can-drive-diversity-and-productivity</u>).

WEF, World Economic Forum (2022g). What are the risks — and rewards — of new work models?, 24 May 2022. (<u>https://www.weforum.org/agenda/2022/05/what-are-the-risks-and-rewards-of-new-work-models</u>).

Wenger, E (1998). Communities of Practice: Learning, Meaning, and Identity. Cambridge: Cambridge University Press.

Wenger, E. (2012). Digital Habitats: Stewarding technology for communities. Kindle ebook.

Wenger-Trayner, E., and Wenger-Trayner, B. (2015). Communities of practice. A brief introduction. (<u>http://wenger-trayner.com/wp-content/uploads/2015/04/07-Brief-introduction-tocommunities-of-practice.pdf</u>).

Weston, G. (2022). What Is Web 5.0 - Explained. 101 Blockchains. (https://101blockchains.com/web-5-0).

Wigan Council (2019). Happy healthy people, Wigan Borough's Locality Plan (2020-25) for radically improving residents' health and wellbeing.

Wilkinson, P. and Pickett, K. (2009). The spirit level: why more equal societies almost always do better. Allen Lane Press, Penguin Group, London.

Wilkelmann, R., Donges, J.F., Smith, E.K, Milkoreit, M., Eder, C., Heitzig, J., Katsanidou, A., Wiedermann, M, Wunderling, N. and Lenton, T.M. (2022). Social tipping processes towards climate action: A conceptual framework, *Ecological Economics*, Volume 192, February 2022, 107242, Elsevier Science Direct (<u>https://www.sciencedirect.com/science/article/abs/pii/S0921800921003013</u>).

World Bank (2017). Risk-Based Regulation, World Bank Group, Open Learning Campus. (<u>https://olc.worldbank.org/content/risk-based-regulation#:~:text=Risk%2DBased%20Regulation%20(RBR),a%20 more%20level%20playing%20field</u>).

Zuboff, S. (2019). The Age of Surveillance Capitalism: The Fight for a Human Future at the New Frontier of Power, New York, Public Affairs.

List of key abbreviations

- 1IR First Industrial Revolution
- 2IR Second Industrial Revolution
- 3IR Third Industrial Revolution
- 4IR Fourth Industrial Revolution
- 5IR Fifth Industrial Revolution
- AI Artificial Intelligence
- CWB Community Wealth Building
- EC European Commission
- IoT Internet of Things
- JRC The European Commission's Joint Research Centre
- OECD Organisation for Economic Co-operation and Development
- SDGs Sustainable Development Goals
- UN United Nations
- WEF World Economic Forum

List of figures

Figure 1. Evolution of European public governance paradigms	15
Figure 2. Balance between efficiency and resilience	26
Figure 3. Process of innovation towards the societal-ecological transition	39
Figure 4: Re-scaling the food system	48
Figure 5: Changing models of delivering public services and public value: from siloed to an 'ecosystem' approach	64
Figure 6. Generative and emergent governance as a 'fishnet organisation'	79

List of tables

Table 1 . Indicative cumulative developments in government and web generations	13
Table 2 : Tracing the cumulative relationship between digital technology and public services since the ea 1990s	rly 17
Table 3. Cumulative chronological overview and summary of public governance paradigm characteristics	<u>19</u>

GETTING IN TOUCH WITH THE EU

In person

All over the European Union there are hundreds of Europe Direct centres. You can find the address of the centre nearest you online (<u>european-union.europa.eu/contact-eu/meet-us_en</u>).

On the phone or in writing

Europe Direct is a service that answers your questions about the European Union. You can contact this service:

- by freephone: 00 800 6 7 8 9 10 11 (certain operators may charge for these calls),
- at the following standard number: +32 22999696,
- via the following form: european-union.europa.eu/contact-eu/write-us_en.

FINDING INFORMATION ABOUT THE EU

Online

Information about the European Union in all the official languages of the EU is available on the Europa website (<u>european-union.europa.eu</u>).

EU publications

You can view or order EU publications at <u>op.europa.eu/en/publications</u>. Multiple copies of free publications can be obtained by contacting Europe Direct or your local documentation centre (<u>european-union.europa.eu/contact-eu/meet-us en</u>).

EU law and related documents

For access to legal information from the EU, including all EU law since 1951 in all the official language versions, go to EUR-Lex (<u>eur-lex.europa.eu</u>).

Open data from the EU

The portal <u>data.europa.eu</u> provides access to open datasets from the EU institutions, bodies and agencies. These can be downloaded and reused for free, for both commercial and non-commercial purposes. The portal also provides access to a wealth of datasets from European countries.

Science for policy

The Joint Research Centre (JRC) provides independent, evidence-based knowledge and science, supporting EU policies to positively impact society



EU Science Hub joint-research-centre.ec.europa.eu

- () @EU_ScienceHub
- (f) EU Science Hub Joint Research Centre
- (in) EU Science, Research and Innovation
- EU Science Hub
- (@eu_science

